Algebra & Number Theory

Volume 12 2018

No. 4

Parabolic induction and extensions

Julien Hauseux



Parabolic induction and extensions

Julien Hauseux

Let G be a p-adic reductive group. We determine the extensions between admissible smooth mod p representations of G parabolically induced from supersingular representations of Levi subgroups of G, in terms of extensions between representations of Levi subgroups of G and parabolic induction. This proves for the most part a conjecture formulated by the author in a previous article and gives some strong evidence for the remaining part. In order to do so, we use the derived functors of the left and right adjoints of the parabolic induction functor, both related to Emerton's δ -functor of derived ordinary parts. We compute the latter on parabolically induced representations of G by pushing to their limits the methods initiated and expanded by the author in previous articles.

1. Introduction	780
2. Generalised Bruhat filtrations	787
2.1. Double cosets	787
2.2. Definition of filtrations	789
2.3. Computation of the associated graded representations	791
3. Derived ordinary parts	794
3.1. Cohomology, Hecke action and dévissage	794
3.2. Computations on the associated graded representations	799
3.3. Computations on parabolically induced representations	804
4. Derived Jacquet functors	810
4.1. Pro-categories	810
4.2. A second exact sequence	810
4.3. Adaptation of the computations	812
5. Application to extensions	816
5.1. Preliminaries	816
5.2. Extensions between parabolically induced representations	821
5.3. Results for unitary continuous p-adic representations	825
Acknowledgements	830
References	830

This research was partly supported by EPSRC grant EP/L025302/1.

MSC2010: 22E50.

Keywords: p-adic reductive groups, mod p representations, parabolic induction, extensions, derived ordinary parts, Bruhat filtration.

1. Introduction

The study of representations of a p-adic reductive group G over a field of characteristic p has a strong motivation in the search for a possible mod p Langlands correspondence for G. Recently, Abe, Henniart, Herzig and Vignéras [Abe et al. 2017a] gave a complete classification of the irreducible admissible smooth representations of G over an algebraically closed field of characteristic p in terms of supersingular representations of the Levi subgroups of G and parabolic induction, generalising the results of Barthel and Livné [1994] for GL_2 , Herzig [2011] for GL_n and Abe [2013] for a split G.

Two major difficulties come into play when trying to extend the mod p Langlands correspondence beyond $GL_2(\mathbb{Q}_p)$. First, the supersingular representations of G remain completely unknown, except for some reductive groups of relative semisimple rank 1 over \mathbb{Q}_p [Abdellatif 2014; Cheng 2013; Kozioł 2016] using the classification of Breuil [2003] for $GL_2(\mathbb{Q}_p)$. Second, it is expected that such a correspondence would involve representations of G with many irreducible constituents; see, e.g., [Breuil and Herzig 2015]. This phenomenon already appears for $GL_2(\mathbb{Q}_p)$ when the Galois representation is an extension between two characters, in which case the corresponding representation of $GL_2(\mathbb{Q}_p)$ is an extension between two principal series [Colmez 2010]. This raises the question of the extensions between representations of G.

In this article, we intend to compute the extensions between admissible smooth mod p representations of G parabolically induced from supersingular representations of Levi subgroups of G, in terms of extensions between representations of Levi subgroups of G and parabolic induction.

In order to do so, we use the derived functors of the left and right adjoints of the parabolic induction functor, namely the Jacquet functor and the ordinary parts functor [Emerton 2010a], both related to Emerton's δ -functor of derived ordinary parts [Emerton 2010b]. We compute the latter on parabolically induced representations of G by pushing to their limits the methods initiated in [Hauseux 2016a] and expanded in [Hauseux 2016b].

These computations have also been used to study the deformations of parabolically induced admissible smooth mod p representations of G in a joint work with T. Schmidt and C. Sorensen [Hauseux et al. 2016].

Presentation of the main results. We let F/\mathbb{Q}_p and k/\mathbb{F}_p be finite extensions. We fix a connected reductive algebraic F-group G, a minimal parabolic subgroup $B \subseteq G$ and a maximal split torus $S \subseteq B$. We write the corresponding groups of F-points G, B, S, etc. We let Δ denote the set of simple roots of S in B. To each $\alpha \in \Delta$ there corresponds a simple reflection s_α and a root subgroup $U_\alpha \subset B$. We put $\Delta^1 := \{\alpha \in \Delta \mid \dim_F U_\alpha = 1\}$.

Let P = LN be a standard parabolic subgroup. We write $\Delta_L \subseteq \Delta$ for the corresponding subset and we put $\Delta_L^{\perp} := \{\alpha \in \Delta \mid \langle \alpha, \beta^{\vee} \rangle = 0 \ \forall \beta \in \Delta_L \}$ and $\Delta_L^{\perp, 1} := \Delta_L^{\perp} \cap \Delta^1$. For $\alpha \in \Delta_L^{\perp, 1}$, conjugation by (any representative of) s_{α} stabilises L and α extends to an algebraic character of L; see the proof of Lemma 5.1.4.

We let P^- denote the opposite parabolic subgroup. Recall the parabolic induction functor $\operatorname{Ind}_{P^-}^G$ from the category of admissible smooth representations of L over k to the category of admissible smooth

representations of G over k, which is k-linear, fully faithful and exact [Emerton 2010a]. In particular, it induces a k-linear injection $\operatorname{Ext}^1_L \hookrightarrow \operatorname{Ext}^1_G$.

Let σ be an admissible smooth representation of L over k. For $\alpha \in \Delta_L^{\perp,1}$, we consider the admissible smooth representation $\sigma^\alpha \otimes (\omega^{-1} \circ \alpha)$ of L over k where σ^α is the s_α -conjugate of σ and $\omega : F^\times \to \mathbb{F}_p^\times \subseteq k^\times$ is the mod p cyclotomic character. We say that σ is *supersingular* if it is absolutely irreducible and $\overline{\mathbb{F}}_p \otimes_k \sigma$ is supersingular [Abe et al. 2017a].

In cases (iii) and (iv) of the following conjecture, "otherwise" means that the conditions of case (ii) are not all satisfied.

Conjecture 1.1 [Hauseux 2016b, Conjecture 3.17]. Assume G split with connected centre and simply connected derived subgroup. Let P = LN, P' = L'N' be standard parabolic subgroups and σ , σ' be supersingular representations of L, L' over k, respectively. Assume $\operatorname{Ind}_{P^-}^G \sigma$ and $\operatorname{Ind}_{P'^-}^G \sigma'$ irreducible or $p \neq 2$.

- (i) If $\mathbf{P}' \not\subseteq \mathbf{P}$ and $\mathbf{P} \not\subseteq \mathbf{P}'$, then $\operatorname{Ext}_G^1(\operatorname{Ind}_{\mathbf{P}'^-}^G \sigma', \operatorname{Ind}_{\mathbf{P}^-}^G \sigma) = 0$.
- (ii) If $F = \mathbb{Q}_p$, P' = P and $\sigma' \cong \sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha) \ncong \sigma$ for some $\alpha \in \Delta_I^{\perp}$, then

$$\dim_k \operatorname{Ext}_G^1(\operatorname{Ind}_{P^-}^G \sigma', \operatorname{Ind}_{P^-}^G \sigma) = 1.$$

(iii) Otherwise if $P' \subseteq P$, then the functor $\operatorname{Ind}_{P^-}^G$ induces a k-linear isomorphism

$$\operatorname{Ext}^1_L\big(\operatorname{Ind}^L_{L\cap P'^-}\sigma',\,\sigma\big)\xrightarrow{\sim}\operatorname{Ext}^1_G\big(\operatorname{Ind}^G_{P'^-}\sigma',\,\operatorname{Ind}^G_{P^-}\sigma\big).$$

(iv) Otherwise if $P \subseteq P'$, then the functor $\operatorname{Ind}_{P'^-}^G$ induces a k-linear isomorphism

$$\operatorname{Ext}^1_{L'}(\sigma',\operatorname{Ind}^{L'}_{L'\cap P^-}\sigma) \xrightarrow{\sim} \operatorname{Ext}^1_G(\operatorname{Ind}^G_{P'^-}\sigma',\operatorname{Ind}^G_{P^-}\sigma).$$

We prove cases (ii), (iii), and (iv) of this conjecture and give some strong evidence for case (i). We actually work without any assumption on G and our results hold true for broader classes of representations; see Section 5.2 for more precise statements. We also prove similar results for unitary continuous p-adic representations; see Section 5.3.

We treat the cases $F = \mathbb{Q}_p$ and $F \neq \mathbb{Q}_p$ separately. They are in fact the degree 1 case of a more general (but conditional to a conjecture of Emerton) result on the k-linear morphism $\operatorname{Ext}_L^n \to \operatorname{Ext}_G^n$ induced by $\operatorname{Ind}_{P^-}^G$ in all degrees $n \leq [F:\mathbb{Q}_p]$; see Remark 5.2.6.

Theorem 1.2 (Theorem 5.2.2). Assume $F = \mathbb{Q}_p$. Let P = LN, P' = L'N' be standard parabolic subgroups and σ , σ' be supersingular representations of L, L' over k, respectively.

(i) If $\mathbf{P}' = \mathbf{P}$ and $\sigma' \ncong \sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha)$ for all $\alpha \in \Delta_L^{\perp,1}$, then the functor $\operatorname{Ind}_{P^-}^G$ induces a k-linear isomorphism

$$\operatorname{Ext}^1_L\bigl(\sigma',\,\sigma\bigr) \xrightarrow{\sim} \operatorname{Ext}^1_G\bigl(\operatorname{Ind}_{P^-}^G\sigma',\,\operatorname{Ind}_{P^-}^G\sigma\bigr).$$

(ii) If $P' \subsetneq P$, then the functor $\operatorname{Ind}_{P^-}^G$ induces a k-linear isomorphism

$$\operatorname{Ext}_L^1(\operatorname{Ind}_{L\cap P'^-}^L\sigma',\sigma) \xrightarrow{\sim} \operatorname{Ext}_G^1(\operatorname{Ind}_{P'^-}^G\sigma',\operatorname{Ind}_{P^-}^G\sigma).$$

(iii) If $P \subsetneq P'$, then the functor $\operatorname{Ind}_{P'^-}^G$ induces a k-linear isomorphism

$$\operatorname{Ext}^1_{L'}(\sigma',\operatorname{Ind}^{L'}_{L'\cap P^-}\sigma) \xrightarrow{\sim} \operatorname{Ext}^1_G(\operatorname{Ind}^G_{P'^-}\sigma',\operatorname{Ind}^G_{P^-}\sigma).$$

If P' = P, we do not know the exact dimension of the cokernel of the k-linear injection $\operatorname{Ext}^1_L(\sigma', \sigma) \hookrightarrow \operatorname{Ext}^1_G(\operatorname{Ind}_{P^-}^G \sigma', \operatorname{Ind}_{P^-}^G \sigma)$ induced by $\operatorname{Ind}_{P^-}^G$ in general, but we prove that it is at most $\operatorname{card}\{\alpha \in \Delta_L^{\perp,1} \mid \sigma' \cong \sigma^\alpha \otimes (\omega^{-1} \circ \alpha)\}$; see Remark 5.2.3 for more details. Further, we compute it when G is split with connected centre; see Theorem 1.4 below. Note that in cases (ii) and (iii), the source of the isomorphism can be nonzero [Hu 2017].

Theorem 1.3 (Theorem 5.2.4). Assume $F \neq \mathbb{Q}_p$. Let P = LN be a standard parabolic subgroup. The functor $\operatorname{Ind}_{P^-}^G$ induces a k-linear isomorphism

$$\operatorname{Ext}^1_L(\sigma', \sigma) \xrightarrow{\sim} \operatorname{Ext}^1_G(\operatorname{Ind}_{P^-}^G \sigma', \operatorname{Ind}_{P^-}^G \sigma)$$

for all admissible smooth representations σ , σ' of L over k.

In particular, Theorem 1.2(ii) and (iii) hold true for any admissible smooth representations σ , σ' of L, L' respectively over k when $F \neq \mathbb{Q}_p$; see Corollary 5.2.5.

We complete Theorem 1.2(i) when G is split with connected centre (see also Remark 5.2.8 for a more general, but conditional to a conjecture of Emerton, result on the k-linear morphism $\operatorname{Ext}_L^{[F:\mathbb{Q}_p]} \to \operatorname{Ext}_G^{[F:\mathbb{Q}_p]}$ induced by $\operatorname{Ind}_{P^-}^G$).

Theorem 1.4 (Theorem 5.2.7). Assume $F = \mathbb{Q}_p$ and G split with connected centre. Let P = LN be a standard parabolic subgroup and σ , σ' be supersingular representations of L over k.

(i) If $\sigma' \cong \sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha) \ncong \sigma$ for some $\alpha \in \Delta_L^{\perp}$, then $\operatorname{Ext}_L^1(\sigma', \sigma) = 0$ and

$$\dim_k \operatorname{Ext}_G^1(\operatorname{Ind}_{P^-}^G \sigma', \operatorname{Ind}_{P^-}^G \sigma) = 1.$$

(ii) If either $\sigma' \cong \sigma$ and $p \neq 2$, or $\sigma' \ncong \sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha)$ for any $\alpha \in \Delta_{L}^{\perp}$, then the functor $\operatorname{Ind}_{P^{-}}^{G}$ induces a k-linear isomorphism

$$\operatorname{Ext}_L^1(\sigma',\sigma) \xrightarrow{\sim} \operatorname{Ext}_G^1(\operatorname{Ind}_{P^-}^G \sigma', \operatorname{Ind}_{P^-}^G \sigma).$$

(iii) If p = 2, then the functor $\operatorname{Ind}_{P^-}^G$ induces a k-linear injection

$$\operatorname{Ext}^1_L \left(\sigma',\sigma\right) \hookrightarrow \operatorname{Ext}^1_G \left(\operatorname{Ind}_{P^-}^G \sigma',\,\operatorname{Ind}_{P^-}^G \sigma\right)$$

whose cokernel is of dimension $\operatorname{card}\{\alpha \in \Delta_L^{\perp} \mid \sigma' \cong \sigma^{\alpha}\}.$

Finally, we treat the case where there is no inclusion between the two parabolic subgroups, assuming a special case of Conjecture 1.7 below; see also Remark 3.3.6.

Proposition 1.5 (Proposition 5.2.1). Let P = LN, P' = L'N' be standard parabolic subgroups and σ , σ' be supersingular representations of L, L' over k, respectively. Assume Conjecture 1.7 is true for A = k, n = 1 and ${}^{I}w^{J} = 1$. If $P' \nsubseteq P$ and $P \nsubseteq P'$, then

$$\operatorname{Ext}_G^1\left(\operatorname{Ind}_{P'^-}^G\sigma',\,\operatorname{Ind}_{P^-}^G\sigma\right)=0.$$

As a consequence, Conjecture 1.1 is true under the same assumption when G is split with connected centre (without assuming the derived subgroup of G simply connected).

Corollary 1.6 (Corollary 5.2.9). Assume G split with connected centre. If Conjecture 1.7 is true for A = k, n = 1 and ${}^{I}w^{J} = 1$, then Conjecture 1.1 is true.

Strategy of proof and methods used. Let E/\mathbb{Q}_p be a finite extension with ring of integers \mathcal{O} and residue field k. We work more generally with smooth representations over an Artinian local \mathcal{O} -algebra A with residue field k.

The main tools to compute extensions between parabolically induced representations are two exact sequences related to Emerton's δ -functor of derived ordinary parts (see below (1) which is due to Emerton and (2) which is a new feature of this article).

Using these, most of the previous results reduce to computing the derived ordinary parts of parabolically induced representations. We formulate a conjecture on these computations (see Conjecture 1.7 below). We prove it in low degree (see Theorem 1.8 below) and give some strong evidence for it in general.

We proceed in two steps: first we construct filtrations of parabolically induced representations related to the Bruhat decomposition; second we partially compute the derived ordinary parts of the associated graded representations using some dévissages.

Derived ordinary parts and extensions. Let $P \subseteq G$ be a parabolic subgroup and $L \subseteq P$ be a Levi factor. We let $P^- \subseteq G$ denote the parabolic subgroup opposed to P with respect to L. Emerton [2010a; 2010b] constructed a cohomological δ -functor $H^\bullet Ord_P$ from the category of admissible smooth representations of G over G to the category of admissible smooth representations of G over G over G over G of G over G or G over G

$$0 \to \operatorname{Ext}_{L}^{1}(\sigma, \operatorname{Ord}_{P} \pi) \to \operatorname{Ext}_{G}^{1}\left(\operatorname{Ind}_{P^{-}}^{G} \sigma, \pi\right) \to \operatorname{Hom}_{L}\left(\sigma, \operatorname{H}^{1}\operatorname{Ord}_{P} \pi\right)$$
(1)

for all admissible smooth representations σ and π of L and G respectively over A.

In Section 4.2, we construct a second exact sequence in which parabolic induction is on the right. The construction is somewhat dual to that of (1) but not exactly (see Remark 4.2.1(ii)). We let d denote the integer $\dim_F N$ and δ denote the algebraic character of the adjoint representation of L on $\det_F(\operatorname{Lie} N)$. The key fact is that the A-linear functors

$$H_{\bullet}(N, -) := H^{[F:\mathbb{Q}_p]d-\bullet} Ord_P \otimes (\omega \circ \delta).$$

form a homological δ -functor from the category of admissible smooth representations of G over A to the category of admissible smooth representations of L over A, which is isomorphic to the left adjoint functor $(-)_N$ of Ind_P^G in degree 0 (hence the notation). From this and using a result of Oort [1964] to compute extensions using pro-categories (see Section 4.1), we derive a natural exact sequence of A-modules

$$0 \to \operatorname{Ext}_{L}^{1}(\pi_{N}, \sigma) \to \operatorname{Ext}_{G}^{1}(\pi, \operatorname{Ind}_{P}^{G} \sigma) \to \operatorname{Hom}_{L}(\operatorname{H}_{1}(N, \pi), \sigma)$$
 (2)

for all admissible smooth representations π and σ of G and L respectively over A.

Computation of derived ordinary parts. We let W be the Weyl group of (G, S). For $I \subseteq \Delta$, we write $P_I = L_I N_I$ for the corresponding standard parabolic subgroup, $B_I \subseteq L_I$ for the minimal parabolic subgroup $B \cap L_I$ and $W_I \subseteq W$ for the subgroup generated by $(s_\alpha)_{\alpha \in I}$.

Let $I, J \subseteq \Delta$, σ be a locally admissible smooth representation of L_I over A and $n \in \mathbb{N}$. We intend to compute the smooth representation of L_J over A

$$\operatorname{H}^{n}\operatorname{Ord}_{P_{J}}\left(\operatorname{Ind}_{P_{I}^{-}}^{G}\sigma\right).$$

In Section 2.2, we use the generalised Bruhat decomposition

$$G = \bigsqcup_{I_{IW}J \in IWJ} P_I^{-I} w^J P_J$$

where ${}^IW^J$ is the system of representatives of minimal length of the double cosets $W_I \setminus W/W_J$ (see Section 2.1) to define a natural filtration $\operatorname{Fil}_{P_J}^{\bullet}(\operatorname{Ind}_{P_I^-}^G\sigma)$ of $\operatorname{Ind}_{P_I^-}^G\sigma$ by $A[P_J]$ -submodules indexed by ${}^IW^J$ (with the Bruhat order). We also adapt the notion of graded representation associated with such a filtration (in particular, the grading has values in ${}^IW^J$) and we prove that for all ${}^Iw^J \in {}^IW^J$, there is a natural $A[P_J]$ -linear isomorphism

$$\operatorname{Gr}_{P_J}^{{}^Iw^J}(\operatorname{Ind}_{P_I^-}^G\sigma)\cong\operatorname{c-ind}_{P_I^-}^{P_I^{-I}w^JP_J}\sigma.$$

We prove that $\operatorname{Fil}_{P_J}^{\bullet}(\operatorname{Ind}_{P_I^-}^G\sigma)$ induces a filtration of $\operatorname{H}^n\operatorname{Ord}_{P_J}(\operatorname{Ind}_{P_I^-}^G\sigma)$ by $A[L_J]$ -submodules indexed by ${}^IW^J$ (see Proposition 3.3.1).

Finally, we attach to each ${}^Iw^J \in {}^IW^J$ an integer $d_{{}^Iw^J}$ and an algebraic character $\delta_{{}^Iw^J}$ of $L_{J\cap {}^Iw^{J-1}(I)}$ (see Notation 2.3.3 and Remark 2.3.4), and we formulate the following conjecture.

Conjecture 1.7 (Conjecture 3.3.4). Let σ be a locally admissible smooth representation of L_I over A, ${}^Iw^J \in {}^IW^J$ and $n \in \mathbb{N}$. There is a natural $A[L_J]$ -linear isomorphism

$$\mathrm{H}^{n}\mathrm{Ord}_{P_{J}}\left(\mathrm{c\text{-}ind}_{P_{J}^{-}}^{P_{J}^{-}l_{w}^{J}P_{J}}\sigma\right)\cong\mathrm{Ind}_{L_{J}\cap P_{I\cap I_{w}^{J-1}(I)}}^{L_{J}}\left((\mathrm{H}^{n-[F:\mathbb{Q}_{p}]d_{I_{w}^{J}}}\mathrm{Ord}_{L_{I}\cap P_{I\cap I_{w}^{J}(J)}}\sigma\right)^{l_{w}^{J}}\otimes\left(\omega^{-1}\circ\delta_{l_{w}^{J}}\right)\right).$$

We give some strong evidence for this conjecture (see Theorem 3.3.3): we prove that these two representations have natural filtrations by $A[B_J]$ -submodules indexed by ${}^{J\cap^I w^{J-1}(I)}W_J$ (the system of representatives of minimal length of the right cosets $W_{J\cap^I w^{J-1}(I)}\backslash W_J$) such that the associated graded representations are naturally isomorphic; see the subsection below.

We prove this conjecture in several cases (see Proposition 3.3.5): whenever the right-hand side is either zero or a trivially induced representation, in which cases the aforementioned filtrations of both sides are trivial; when n=0, in which case we deduce the result from the computation of $\operatorname{Ord}_{P_J}(\operatorname{Ind}_{P_I^-}^G\sigma)$ in [Abe et al. 2017b]. This allows us to compute $\operatorname{H}^\bullet\operatorname{Ord}_{P_J}(\operatorname{Ind}_{P_I^-}^G\sigma)$ in low degree when there is an inclusion between I and J; see Proposition 3.3.7. In particular, we obtain the following result in the case I=J.

Theorem 1.8 (Corollary 3.3.8). Let P = LN be a standard parabolic subgroup and σ be a locally admissible smooth representation of L over A.

(i) For all $n \in \mathbb{N}$ such that $0 < n < [F : \mathbb{Q}_p]$, we have $H^n \operatorname{Ord}_P(\operatorname{Ind}_{P^-}^G \sigma) = 0$.

(ii) If $\operatorname{Ord}_{L\cap s_{\alpha}Ps_{\alpha}^{-1}}\sigma=0$ for all $\alpha\in\Delta^1\setminus(\Delta_L\cup\Delta_L^\perp)$, then there is a natural A[L]-linear isomorphism

Note that for all $\alpha \in \Delta \setminus \Delta_L$, $L \cap s_\alpha P s_\alpha^{-1}$ is the standard parabolic subgroup of L corresponding to $\Delta_L \cap s_\alpha(\Delta_L)$ and it is proper if and only if $\alpha \notin \Delta_L^{\perp}$. In particular, the condition in (ii) is satisfied when σ is supersingular.

In Section 4.3, we adapt the previous results in order to partially compute $H_n(N_J, \operatorname{Ind}_{P_I}^G \sigma)$. We obtain an analogue of Theorem 1.8; see Corollary 4.3.4.

Filtrations and dévissages. Let ${}^Iw^J \in {}^IW^J$. We explain the partial computation of the smooth representation of L_J over A

$$H^n \operatorname{Ord}_{P_J} \left(\operatorname{c-ind}_{P_I^-}^{P_I^{-l} w^J P_J} \sigma \right).$$

In Section 2.2, we use again the Bruhat decomposition to construct a natural filtration $\operatorname{Fil}_{B}^{\bullet}(\operatorname{c-ind}_{P_{l}^{-}}^{P_{l}^{-}lw^{J}P_{J}}\sigma)$ by A[B]-submodules indexed by ${}^{J\cap lw^{J-1}(I)}W_{J}$, and we prove that for all $w_{J}\in {}^{J\cap lw^{J-1}(I)}W_{J}$ there is a natural A[B]-linear isomorphism

$$\operatorname{Gr}_B^{w_J}(\operatorname{c-ind}_{P_I^{-}}^{P_I^{-}l_W^JP_J}\sigma) \cong \operatorname{c-ind}_{P_I^{-}}^{P_I^{-}l_W^Jw_JB}\sigma.$$

We prove that $\operatorname{Fil}_{P_l}^{\bullet}(\operatorname{c-ind}_{P_l^-}^{P_l^-lw^JP_J}\sigma)$ induces a filtration of $\operatorname{H}^n\operatorname{Ord}_{P_J}(\operatorname{c-ind}_{P_l^-}^{P_l^-lw^JP_J}\sigma)$ by $A[B_J]$ -submodules indexed by ${}^{J\cap^lw^{J-1}(I)}W_J$ (see Proposition 3.3.2). Likewise, we construct a natural filtration

$$\operatorname{Fil}_{B_J}^{ullet} \left(\operatorname{Ind}_{L_J \cap P_{I \cap I_w J - 1_{(I)}}}^{L_J} \tilde{\sigma} \right)$$

by $A[B_J]$ -submodules indexed by ${}^{J\cap^I w^{J-1}(I)}W_J$ for any smooth representation $\tilde{\sigma}$ of $L_{J\cap^I w^{J-1}(I)}$ over A. Let $w_J \in {}^{J\cap^I w^{J-1}(I)}W_J$ and set ${}^I w := {}^I w^J w_J$ and $\pi_{I_w} := \operatorname{c-ind}_{P_I^-}^{P_I^- IwB} \sigma$. We want to compute the A-module $H^n(N_{J,0},\pi_{I_w})$ endowed with the Hecke action of B_J^+ (see Section 3.1), where $N_{J,0} \subseteq N_J$ is a compact open subgroup and $B_J^+ \subseteq B_J$ is the open submonoid stabilising N_0 by conjugation (we use similar notation for subgroups of N_J and B_J by taking intersections with $N_{J,0}$ and B_J^+ respectively).

In Section 2.3, we define closed subgroups $N_{J,l_w} \subseteq N_J$ and $B_{J,w_J} \subseteq B_J$ such that there is a semidirect product $B_{J,w_J} \ltimes N_{J,l_w}$ and we give an explicit description of the actions of N_{J,l_w} and B_{J,w_J} on π_{l_w} for all $w_J \in {}^{J \cap {}^l w^{J-1}(I)} W_J$. Then, we compute the A-module $H^n(N_{J,l_w,0}, \pi_{l_w})$ with the Hecke action of B_{J,w_J}^+ (see Proposition 3.2.6).

The idea is to use a semidirect product $N_{J,^lw}=N''_{J,^lw}\ltimes N'_{J,^lw}$ (also defined in Section 2.3) where $N'_{J,^lw}\subseteq N_{J,^lw}$ is a closed subgroup stable under conjugation by B_{J,w_J} such that π_{lw} is $N'_{J,^lw,0}$ -acyclic and there is an $A[B^+_{J,w_J}]$ -linear surjection with a locally nilpotent kernel from $\pi^{N'_{J,^lw,0}}_{l_m}$ onto

$$\mathrm{Gr}_{B_J}^{w_J} \big(\mathrm{Ind}_{L_J \cap P_{J \cap I_w J - 1}_{(I)}}^{L_J} (\sigma_{|L_{I \cap I_w J}_{(J)}})^{l_w J} \big).$$

Then, taking the $N_{J,^Iw,0}''$ -cohomology changes $\sigma_{|L_{I\cap^Iw^J(J)}}$ into $H^n\mathrm{Ord}_{L_I\cap P_{I\cap^Iw^J(J)}}\sigma$ in the target and the inflation map is an $A[B_{J,w_I}^+]$ -linear isomorphism between the source and $H^n(N_{J,^Iw,0},\pi_{^Iw})$.

Finally, by a technical result on dévissages (see Proposition 3.1.2) and a finiteness property of the A-modules $H^{\bullet}(N_{J,^{I}w,0}, \pi_{I_{w}})$, we can compute the A-module $H^{n}(N_{J,0}, \pi_{I_{w}})$ with the Hecke action of

 B_{J,w_J}^+ from $H^n(N_{J,l_{w,0}}, \pi_{l_w})$. It is this dévissage that introduces the degree shift and the twist (i.e., $d_{l_w J}$ and $\delta_{l_w J}$) in the formulas.

Notation and terminology. Let F/\mathbb{Q}_p be a finite extension. A linear algebraic F-group will be denoted by a boldface letter like H and the group of its F-points H(F) will be denoted by the corresponding ordinary letter H. We will also write H^{der} for its derived subgroup and H° for its identity component. The group of algebraic characters of H will be denoted by $X^*(H)$, the group of algebraic cocharacters of H will be denoted by $X_*(H)$, and we will write $\langle -, - \rangle : X^*(H) \times X_*(H) \to \mathbb{Z}$ for the natural pairing. We now turn to reductive groups. The main reference for these is [Borel and Tits 1965].

Let G be a connected reductive algebraic F-group. We write Z for the centre of G. Let $S \subseteq G$ be a maximal split torus. We write Z (resp. N) for the centraliser (resp. normaliser) of S in G and W for the Weyl group N/Z = N/Z. We write $\Phi \subseteq X^*(S)$ for the set of roots of S in G and $\Phi_0 \subseteq \Phi$ for the subset of reduced roots. To each $\alpha \in \Phi$ correspond a coroot $\alpha^{\vee} \in X_*(S)$, a reflection $s_{\alpha} \in W$ and a root subgroup $U_{\alpha} \subset G$ (which is denoted by $U_{(\alpha)}$ in [loc. cit.]). For $\alpha, \beta \in \Phi$, we write $\alpha \perp \beta$ if and only if $\langle \alpha, \beta^{\vee} \rangle = 0$. For $I \subseteq \Delta$, we put

$$I^{\perp} := \{ \alpha \in \Delta \mid \alpha \perp \beta \ \forall \beta \in I \}.$$

Let $B \subseteq G$ be a minimal parabolic subgroup containing S. We write U for the unipotent radical of B (so that $B = \mathcal{Z}U$), $\Phi^+ \subseteq \Phi$ for the subset of roots of S in U and $\Delta \subseteq \Phi^+$ for the subset of simple roots. We set $\Phi_0^+ := \Phi_0 \cap \Phi^+$. A simple reflection is a reflection $s_\alpha \in W$ with $\alpha \in \Delta$. A reduced decomposition of $w \in W$ is any decomposition into simple reflections $w = s_1 \dots s_n$ with $n \in \mathbb{N}$ minimal, which is called the length of w and denoted by $\ell(w)$. We write w_0 for the element of maximal length in W.

We say that P = LN is a standard parabolic subgroup if $P \subseteq G$ is a parabolic subgroup containing B with unipotent radical N and $L \subseteq P$ is the Levi factor containing S (we say that L is a standard Levi subgroup). In this case, we write P^- for the parabolic subgroup of G opposed to P with respect to L (i.e., $P \cap P^- = L$) and N^- for the unipotent radical of P^- . We write Z_L for the centre of L, $B_L \subseteq L$ for the minimal parabolic subgroup $B \cap L$, $U_L \subseteq B_L$ for the unipotent radical $U \cap L$ (so that $B_L = \mathcal{Z}U_L$ and $U = U_L \ltimes N$) and $\Delta_L \subseteq \Delta$ for the subset of simple roots of S in U_L .

Each parabolic subgroup of G is conjugate to exactly one standard parabolic subgroup and the map $P = LN \mapsto \Delta_L$ yields a bijection between standard parabolic subgroups of G and subsets of Δ . For $I \subseteq \Delta$, we write $P_I = L_I N_I$ for the corresponding standard parabolic subgroup (i.e., $\Delta_{L_I} = I$), Z_I , B_I , U_I instead of Z_{L_I} , B_{L_I} , U_{L_I} respectively, $W_I \subseteq W$ for the subgroup generated by $(s_\alpha)_{\alpha \in I}$ (so that $P_I = BW_IB$), $w_{I,0}$ for the element of maximal length in W_I , $\Phi_I \subseteq \Phi$ for the subset of roots of S in L_I and $\Phi_I^+ \subseteq \Phi^+$ for the subset of roots of S in U_I .

Let E/\mathbb{Q}_p be a finite extension with ring of integers \mathcal{O} and residue field k. We let A be an Artinian local \mathcal{O} -algebra with residue field k. We write $\varepsilon: F^\times \to \mathbb{Z}_p^\times \subseteq \mathcal{O}^\times$ for the p-adic cyclotomic character (defined by

$$\varepsilon(x) = \operatorname{Nrm}_{F/\mathbb{Q}_p}(x) |\operatorname{Nrm}_{F/\mathbb{Q}_p}(x)|_p$$

for all $x \in F^{\times}$) and $\omega : F^{\times} \to A^{\times}$ for its image in A^{\times} .

We use the terminology and notation of [Emerton 2010a, §2] for representations of a p-adic Lie group H over A. An H-representation is a smooth representation of H over A and a morphism between H-representations is A-linear. We write $\operatorname{Mod}_H^{\mathrm{sm}}(A)$ for the category of H-representations and H-equivariant morphisms, and $\operatorname{Mod}_H^{\mathrm{adm}}(A)$ (resp. $\operatorname{Mod}_H^{\mathrm{l.adm}}(A)$, $\operatorname{Mod}_H^{\mathrm{sm}}(A)^{Z_H-\mathrm{l.fin}}$) for the full subcategory of admissible (resp. locally admissible, locally Z_H -finite) H-representations (here Z_H denotes the centre of H).

Assume $H \subseteq G$ is closed and π is an H-representation. For $g \in G$, we write π^g for the $g^{-1}Hg$ -representation with the same underlying A-module as π on which $g^{-1}hg$ acts as h for all $h \in H$. If $g \in H$, then $g^{-1}Hg = H$ and the action of g on π induces a natural H-equivariant isomorphism $\pi \xrightarrow{\sim} \pi^g$.

Assume furthermore $\mathcal{Z} \subseteq H$. For $w \in W$, we write π^w for the $n^{-1}Hn$ -representation π^n where $n \in \mathcal{N}$ is any representative of w (neither $n^{-1}Hn$ nor π^n depend on the choice of n up to isomorphism). For $\alpha \in \Delta$, we simply write π^{α} instead of $\pi^{s_{\alpha}}$.

For a topological space X and an A-module V, we write $C^{\text{sm}}(X, V)$ for the A-module of locally constant functions $f: X \to V$ and $C^{\text{sm}}_{c}(X, V)$ for the A-submodule consisting of those functions with compact support (the support of f is the open and closed subset supp $f := f^{-1}(V \setminus \{0\}) \subseteq X$).

2. Generalised Bruhat filtrations

The aim of this section is to define filtrations of parabolically induced representations and describe the associated graded representations. In Section 2.1, we review some properties of the representatives of minimal length of certain double cosets in W and some variants of the Bruhat decomposition. In Section 2.2, we define the notion of filtration indexed by a poset and we construct filtrations of induced representations indexed by subsets of W with the Bruhat order using the previous decompositions. In Section 2.3, we define several subgroups of U that we use to describe the graded representations associated with the previous filtrations as spaces of locally constant functions with compact support.

2.1. Double cosets. We recall some facts about certain right cosets in W; see [Borel and Tits 1972, Proposition 3.9]. For any $I \subseteq \Delta$, we define a system of representatives of the right cosets $W_I \setminus W$ by setting

$${}^{I}W := \{ w \in W \mid w \text{ is of minimal length in } W_{I}w \}.$$

For all $w \in W$, there exists a unique decomposition $w = w_I{}^I w$ with $w_I \in W_I$ and ${}^I w \in {}^I W$. This decomposition is characterised by the equality

$$\Phi_I^+ \cap w(\Phi^+) = \Phi_I^+ \cap w_I(\Phi_I^+).$$

In particular, we have ${}^Iw^{-1}(\Phi_I^+)\subseteq\Phi^+$. Furthermore, we have $\ell(w)=\ell(w_I)+\ell({}^Iw)$.

We now recall some properties of certain double cosets in W; see, for example, [Digne and Michel 1991, Lemma 5.4]. For any $I, J \subseteq \Delta$, we define a system of representatives of the double cosets $W_I \setminus W/W_J$ by setting

$${}^{I}W^{J} := {}^{I}W \cap ({}^{J}W)^{-1}.$$

For all ${}^Iw \in {}^IW$, there exists a unique decomposition ${}^Iw = {}^Iw^Jw_J$ with ${}^Iw^J \in {}^IW^J$ and $w_J \in W_J$. In fact $w_J \in {}^{J \cap {}^Iw^{J-1}(I)}W_J$. This decomposition is characterised by the equality

$$\Phi_I^+ \cap {}^I w^{-1}(\Phi^+) = \Phi_I^+ \cap w_I^{-1}(\Phi_I^+). \tag{3}$$

In particular, we have ${}^Iw^J(\Phi_J^+) \subseteq \Phi^+$. Furthermore, we have $\ell({}^Iw) = \ell({}^Iw^J) + \ell(w_J)$. Conversely, for all ${}^Iw^J \in {}^IW^J$ and $w_J \in W_J$, we have ${}^Iw^Jw_J \in {}^IW$ if and only if $w_J \in {}^{J \cap {}^Iw^{J-1}(I)}W_J$. Note that the projections $W \to {}^IW$ and ${}^IW \to {}^IW^J$ respect the Bruhat order; see [Björner and Brenti 2005, Proposition 2.5.1].

Lemma 2.1.1. We have the following equalities in G.

- (i) $L_I \cap {}^I w^J U_J {}^I w^{J-1} = U_{I \cap {}^I w^J (J)}$
- (ii) $\boldsymbol{L}_I \cap {}^{I}w^{J}\boldsymbol{L}_J{}^{I}w^{J-1} = \boldsymbol{L}_{I \cap {}^{I}w^{J}(J)}$
- (iii) $L_I \cap {}^I w^J N_J {}^I w^{J-1} = L_I \cap N_{I \cap {}^I w^J (J)}$
- (iv) $\boldsymbol{L}_I \cap {}^{I}w^{J}\boldsymbol{P}_J{}^{I}w^{J-1} = \boldsymbol{L}_I \cap \boldsymbol{P}_{I \cap {}^{I}w^{J}(J)}$

Proof. First, we prove the following equalities in Φ :

$$\Phi_I \cap {}^I w^J(J) = I \cap {}^I w^J(J), \tag{4}$$

$$\Phi_I \cap {}^I w^J \left(\Phi_J^+ \right) = \Phi_{I \cap {}^I w^J (I)}^+. \tag{5}$$

We prove the nontrivial inclusion of (4). Assume $\Phi_I \cap {}^I w^J(J) \neq \emptyset$ and let $\alpha \in \Phi_I \cap {}^I w^J(J)$. Since ${}^I w^J(J) \subseteq \Phi^+$, $\alpha \in \Phi_I^+$ so that there exists $(r_\beta)_{\beta \in I} \in \mathbb{N}^I$ such that $\alpha = \sum_{\beta \in I} r_\beta \beta$. Then ${}^I w^{J-1}(\alpha) = \sum_{\beta \in I} r_\beta {}^I w^{J-1}(\beta) \in \Delta$. Since ${}^I w^{J-1}(\beta) \in \Phi^+$ for all $\beta \in I$, $r_\beta = 0$ for all $\beta \in I \setminus \{\alpha\}$ and $r_\alpha = 1$. Thus $\alpha \in I$. We prove the nontrivial inclusion of (5). Assume $\Phi_I \cap {}^I w^J(\Phi_J^+) \neq \emptyset$ and let $\alpha \in \Phi_I \cap {}^I w^J(\Phi_J^+)$. There exists $(r_\beta)_{\beta \in J} \in \mathbb{N}^J$ such that $\alpha = \sum_{\beta \in J} r_\beta {}^I w^J(\beta)$. Since ${}^I w^J(\beta) \in \Phi^+$ for all $\beta \in J$, ${}^I w^J(\beta) \in \Phi_I^+$ so that ${}^I w^J(\beta) \in I$ by (4) for all $\beta \in J$ such that $r_\beta > 0$. Thus $\alpha \in \Phi_{I \cap I}^+ w^J(J)$.

Now, by considering the Lie algebras, (5) yields (i), (5) and its opposite yield (ii), the equality $\Phi_I \cap {}^I w^J (\Phi^+ \backslash \Phi_J^+) = \Phi_I^+ \backslash \Phi_{I \cap {}^I w^J (J)}^+$ (which follows from (5) and the fact that $\Phi_I \cap {}^I w^J (\Phi^+) = \Phi_I^+$ since ${}^I w^J \in {}^I W$) yields (iii), and we deduce (iv) from (ii) and (iii).

Finally, we give certain decompositions in double cosets (for the notion of "lower set", see foonote 2 on p. 789).

Lemma 2.1.2. (i) We have $G = \bigsqcup_{I_w^J \in I_W^J} P_I^{-I} w^J P_J$ and for any lower set $IW_1^J \subseteq IW^J$, the subset $P_I^{-I} W_1^J P_J \subseteq G$ is open.

- (ii) We have $P_I^{-I}w^JP_J = \bigsqcup_{w_J \in J \cap I_w J^{-1}(I)W_J} P_I^{-I}w^Jw_JB$ and for any lower set $W_J' \subseteq J \cap I_w J^{-1}(I)W_J$, the subset $P_I^{-I}w^JW_J'B \subseteq P_I^{-I}w^JP_J$ is open.
- (iii) We have $L_J = \bigsqcup_{w_J \in J \cap I_w J^{-1}(I)} W_J L_J \cap P_{J \cap I_w J^{-1}(I)}^- w_J B_J$ and for any lower set $W_J' \subseteq J \cap I_w J^{-1}(I) W_J$, the subset $L_J \cap P_{J \cap I_w J^{-1}(I)}^- W_J' B_J \subseteq L_J$ is open.

¹The Bruhat order on W is defined by $w \le w'$ if and only if there exist a reduced decomposition $w' = s_1 \dots s_{\ell(w')}$ and integers $1 \le i_1 < \dots < i_{\ell(w)} \le \ell(w')$ such that $w = s_{i_1} \dots s_{i_{\ell(w)}}$.

Proof. We have $G = \bigsqcup_{l_w \in {}^I W} P_I^{-l} w B$ and for any ${}^I w \in {}^I W$, the closure of $P_I^{-l} w B$ in G is $\bigsqcup_{l_{w'} \geq l_w} P_I^{-l} w' B$. (This can be deduced from the Bruhat decomposition, see, e.g., [Hauseux 2016a, §2.3].) Furthermore, for any ${}^I w^J \in {}^I W^J$, we have

$$P_I^{-I}w^J P_J = \bigcup_{w_J \in W_J} P_I^{-w_0} B w_0^I w^J w_{J,0} B w_J B = \bigcup_{w_J \in W_J} P_I^{-w_0} B w_0^I w^J w_{J,0} w_J B = \bigcup_{w_J \in W_J} P_I^{-I} w^J w_J B.$$

(The first and third equalities follow from the inclusion $w_0Bw_0 = B^- \subseteq P_I^-$ and the decomposition $P_J = BW_JB$, and the second equality follows from [Borel and Tits 1972, Lemme 3.4(iv); Björner and Brenti 2005, Proposition 2.5.4].) From this we deduce (ii), and also (i) using the fact that the projection ${}^IW \to {}^IW^J$ is order-preserving. Finally, (iii) is (i) for the double cosets $L_J \cap P_{J \cap I_W^{J-1}(I)}^- \setminus L_J/B_J$ instead of $P_I^- \setminus G/P_J$.

Remark 2.1.3 (case $w_J = 1$). Note that $P_I^{-I}w^JB$ is $P_{J \cap I_w J^{-1}(I)}$ -invariant by right translation. In general, the stabiliser of $P_I^{-I}wB$ in G for the action by right translation is the (nonstandard) parabolic subgroup $B^Iw^{-1}W_I^IwB$. Likewise, $L_J \cap P_{J \cap I_w J^{-1}(I)}B_J$ is $L_J \cap P_{J \cap I_w J^{-1}(I)}$ -invariant by right translation.

2.2. Definition of filtrations.

Filtration indexed by a poset. Let H be a p-adic Lie group, π be an H-representation and (\widetilde{W}, \leq) be a poset. A filtration of π indexed by \widetilde{W} is a morphism of complete lattices $\mathrm{Fil}_H^\bullet \pi$ from the complete lattice of lower sets of \widetilde{W} to the complete lattice of H-subrepresentations of π , i.e., an H-subrepresentation $\mathrm{Fil}_H^{\widetilde{W}'} \pi \subseteq \pi$ for each lower set $\widetilde{W}' \subseteq \widetilde{W}$ such that for any family $(\widetilde{W}_i)_{i \in \mathcal{I}}$ of lower sets of \widetilde{W} , we have the following equalities in π :

$$\operatorname{Fil}_H^{\bigcap_{i\in\mathcal{I}}\widetilde{W}_i}\pi=\bigcap_{i\in\mathcal{I}}\operatorname{Fil}_H^{\widetilde{W}_i}\pi,\quad \operatorname{Fil}_H^{\bigcup_{i\in\mathcal{I}}\widetilde{W}_i}\pi=\sum_{i\in\mathcal{I}}\operatorname{Fil}_H^{\widetilde{W}_i}\pi.$$

When \widetilde{W} is finite, these two equalities are equivalent (by induction) to the following conditions: Fil $_H^{\bullet}$ π is inclusion-preserving with Fil $_H^{\varnothing}$ $\pi=0$ and Fil $_H^{\widetilde{W}}$ $\pi=\pi$ (i.e., the empty family case), and for any lower sets \widetilde{W}_1 , $\widetilde{W}_2\subseteq\widetilde{W}$ the short sequence of H-representations

$$0 \to \operatorname{Fil}_H^{\widetilde{W}_1 \cap \widetilde{W}_2} \pi \to \operatorname{Fil}_H^{\widetilde{W}_1} \pi \oplus \operatorname{Fil}_H^{\widetilde{W}_2} \pi \to \operatorname{Fil}_H^{\widetilde{W}_1 \cup \widetilde{W}_2} \pi \to 0,$$

defined by $v \mapsto (v, -v)$ and $(v_1, v_2) \mapsto v_1 + v_2$, is exact.

Each $\widetilde{w} \in \widetilde{W}$ defines a principal lower set $\{\widetilde{w}' \in \widetilde{W} \mid \widetilde{w}' \leq \widetilde{w}\}$ and we write $\mathrm{Fil}_H^{\widetilde{w}} \pi$ for the corresponding H-subrepresentation of π . Note that for any lower set $\widetilde{W}' \subseteq \widetilde{W}$, we have the following equality in π :

$$\operatorname{Fil}_H^{\widetilde{W}'}\pi = \sum_{\widetilde{w}' \in \widetilde{W}'} \operatorname{Fil}_H^{\widetilde{w}'}\pi.$$

Thus, we can recover the whole filtration from the H-subrepresentations of π corresponding to the elements of \widetilde{W} ; hence the terminology. We define the graded representation $Gr_H^{\bullet} \pi$ associated with the filtration $Fil_H^{\bullet} \pi$ by setting

$$\operatorname{Gr}_H^{\tilde{w}}\pi := \operatorname{Fil}_H^{\tilde{w}}\pi \big/ \underset{\tilde{w}' < \tilde{w}}{\sum} \operatorname{Fil}_H^{\tilde{w}'}\pi \quad \text{ for each } \tilde{w} \in \widetilde{W}.$$

A lower set of \widetilde{W} is a subset \widetilde{W}' such that $\widetilde{w} < \widetilde{w}' \Rightarrow \widetilde{w} \in \widetilde{W}'$ for any $\widetilde{w} \in \widetilde{W}$ and $\widetilde{w}' \in \widetilde{W}'$.

Let $\tilde{\ell}: \widetilde{W} \to \mathbb{Z}$ be a monotonic map (i.e., $\tilde{w} \leq \tilde{w}' \Rightarrow \tilde{\ell}(\tilde{w}) \leq \tilde{\ell}(\tilde{w}')$ for any $\tilde{w}, \tilde{w}' \in \widetilde{W}$). For each $n \in \mathbb{Z}$, we set

$$\operatorname{Fil}_{H}^{\tilde{\ell},n}\pi := \sum_{\tilde{\ell}(\tilde{w}) \leq n} \operatorname{Fil}_{H}^{\tilde{w}}\pi.$$

We obtain a filtration of π indexed by \mathbb{Z} (in the usual sense).

Lemma 2.2.1. Assume $\tilde{\ell}: \widetilde{W} \to \mathbb{Z}$ is strictly monotonic (i.e., $\tilde{w} < \tilde{w}' \Rightarrow \tilde{\ell}(\tilde{w}) < \tilde{\ell}(\tilde{w}')$ for any $\tilde{w}, \tilde{w}' \in \widetilde{W}$). For all $n \in \mathbb{Z}$, there is a natural H-equivariant isomorphism

$$\operatorname{Gr}_H^{\tilde{\ell},n}\pi\cong\bigoplus_{\tilde{\ell}(\tilde{w})=n}\operatorname{Gr}_H^{\tilde{w}}\pi.$$

Proof. Let $n \in \mathbb{Z}$. By definition of $\operatorname{Fil}_H^{\tilde{\ell},n} \pi$ and $\operatorname{Gr}_H^{\tilde{\ell},n} \pi$, there are natural H-equivariant surjections

$$\bigoplus_{\tilde{\ell}(\tilde{w}) \le n} \operatorname{Fil}_{H}^{\tilde{w}} \pi \twoheadrightarrow \operatorname{Fil}_{H}^{\tilde{\ell},n} \pi \twoheadrightarrow \operatorname{Gr}_{H}^{\tilde{\ell},n} \pi. \tag{6}$$

The kernel of (6) contains $\bigoplus_{\tilde{\ell}(\tilde{w}) \leq n} \operatorname{Fil}_H^{\tilde{w}} \pi \cap \operatorname{Fil}_H^{\tilde{\ell},n-1} \pi$, and $\operatorname{Fil}_H^{\tilde{w}} \pi \cap \operatorname{Fil}_H^{\tilde{\ell},n-1} \pi = \operatorname{Fil}_H^{\tilde{w}} \pi$ for all $\tilde{w} \in \widetilde{W}$ such that $\tilde{\ell}(\tilde{w}) < n$. Now, for any $\tilde{w}_0 \in \widetilde{W}$ such that $\tilde{\ell}(\tilde{w}_0) = n$, we have the following equality in π :

$$\operatorname{Fil}_{H}^{\tilde{w}_{0}} \pi \cap \sum_{\substack{\tilde{\ell}(\tilde{w}) \leq n \\ \tilde{w} \neq \tilde{w}_{0}}} \operatorname{Fil}_{H}^{\tilde{w}} \pi = \sum_{\tilde{w} < \tilde{w}_{0}} \operatorname{Fil}_{H}^{\tilde{w}} \pi,$$

which results from the following equality in \widetilde{W} :

$$\left\{\tilde{w}' \in \widetilde{W} \;\middle|\; \tilde{w}' \leq \tilde{w}_0\right\} \cap \bigcup_{\substack{\tilde{\ell}(\tilde{w}) \leq n \\ \tilde{w} \neq \tilde{w}_0}} \left\{\tilde{w}' \in \widetilde{W} \;\middle|\; \tilde{w}' \leq \tilde{w}\right\} = \bigcup_{\tilde{w} < \tilde{w}_0} \left\{\tilde{w}' \in \widetilde{W} \;\middle|\; \tilde{w}' \leq \tilde{w}\right\},$$

which in turn follows from the fact that $\tilde{w}_0 \not\leq \tilde{w}$ for all $\tilde{w} \in \widetilde{W} \setminus \{\tilde{w}_0\}$ such that $\tilde{\ell}(\tilde{w}) \leq n$ by strict monotonicity of $\tilde{\ell}$. We deduce that the kernel of (6) is $\bigoplus_{\tilde{\ell}(\tilde{w}) \leq n} \operatorname{Fil}_H^{\tilde{w}} \pi \cap \operatorname{Fil}_H^{\tilde{\ell},n-1} \pi$, and that $\operatorname{Fil}_H^{\tilde{w}} \pi \cap \operatorname{Fil}_H^{\tilde{\ell},n-1} \pi = \sum_{\tilde{w}' < \tilde{w}} \operatorname{Fil}_H^{\tilde{w}'} \pi$ for all $\tilde{w} \in \widetilde{W}$ such that $\tilde{\ell}(\tilde{w}) = n$. We conclude that (6) induces an isomorphism as in the statement.

Filtrations of induced representations. Let $I, J \subseteq \Delta$ and σ be an L_I -representation. Recall that for any locally closed subset $X \subseteq G$ and for any open subset $Y \subseteq X$, both P_I^- -invariant by left translation, there is a natural short exact sequence of A-modules

$$0 \to \operatorname{c-ind}_{P_l^-}^Y \sigma \to \operatorname{c-ind}_{P_l^-}^X \sigma \to \operatorname{c-ind}_{P_l^-}^{X \setminus Y} \sigma \to 0.$$

(See [Bernšteĭn and Zelevinskiĭ 1976, Proposition 1.8]; see also the proof of [Hauseux 2016a, Proposition 2.1.3].) Note that there is a natural A-linear isomorphism $\operatorname{c-ind}_{P_l^-}^G \sigma \xrightarrow{\sim} \operatorname{Ind}_{P_l^-}^G \sigma$ since $P_l^- \setminus G$ is compact.

For each lower set ${}^IW_1^J \subseteq {}^IW^J$, we define a P_J -subrepresentation of $\operatorname{Ind}_{P_r}^G \sigma$ by setting

$$\operatorname{Fil}_{P_J}^{{}^IW_1^J} \left(\operatorname{Ind}_{P_I^-}^G \sigma \right) := \operatorname{c-ind}_{P_I^-}^{P_I^{-I}W_1^J P_J} \sigma.$$

Using Lemma 2.1.2(i), we obtain a filtration of $\operatorname{Ind}_{P_I^-}^G \sigma$ indexed by ${}^IW^J$ such that for all ${}^Iw^J \in {}^IW^J$, there is a natural P_J -equivariant isomorphism

$$\operatorname{Gr}_{P_J}^{l_w J} \left(\operatorname{Ind}_{P_I^-}^G \sigma \right) \cong \operatorname{c-ind}_{P_I^-}^{P_I^- l_w J} P_J \sigma. \tag{7}$$

Let ${}^I w^J \in {}^I W^J$. For each lower set $W_J' \subseteq {}^{J \cap {}^I w^{J-1}(I)} W_J$, we define a B-subrepresentation of c-ind ${}^{P_I^{-1} w^J P_J}_{P_I^-} \sigma$ by setting

$$\operatorname{Fil}_B^{W'_J} \left(\operatorname{c-ind}_{P_I^-}^{P_I^- l_W ^J P_J} \sigma \right) := \operatorname{c-ind}_{P_I^-}^{P_I^- l_W ^J W'_J ^B} \sigma.$$

Using Lemma 2.1.2(ii), we obtain a filtration of c-ind $P_I^{-l}w^J P_J \sigma$ indexed by $J \cap lw^{J-1}(I) W_J$ such that for all $W_J \in J \cap lw^{J-1}(I) W_J$, there is a natural B-equivariant isomorphism

$$\operatorname{Gr}_{B}^{w_{J}}\left(\operatorname{c-ind}_{P_{I}^{-}}^{P_{I}^{-}l_{w^{J}}P_{J}}\sigma\right) \cong \operatorname{c-ind}_{P_{I}^{-}}^{P_{I}^{-}l_{w^{J}}w_{J}B}\sigma. \tag{8}$$

Likewise, for any $L_{J\cap I_w^{J-1}(I)}$ -representation $\tilde{\sigma}$ and using Lemma 2.1.2(iii), we define for each lower set $W_J'\subseteq {}^{J\cap I_w^{J-1}(I)}W_J$ a B_J -subrepresentation of $\mathrm{Ind}_{L_J\cap P_{J\cap I_w^{J-1}(I)}}^{L_J}\tilde{\sigma}$ by setting

$$\operatorname{Fil}_{B_J}^{W_J'} \left(\operatorname{Ind}_{L_J \cap P_{J \cap I_w J - 1(I)}}^{L_J} \tilde{\sigma} \right) := \operatorname{c-ind}_{L_J \cap P_{J \cap I_w J - 1(I)}}^{L_J \cap P_{J \cap I_w J - 1(I)}^-} W_J'^{B_J} \tilde{\sigma}$$

and we obtain a filtration of $\operatorname{Ind}_{L_J \cap P_{J \cap I_w J^{-1}(I)}}^{L_J} \tilde{\sigma}$ indexed by ${}^{J \cap I_w J^{-1}(I)} W_J$ such that for all $w_J \in {}^{J \cap I_w J^{-1}(I)} W_J$, there is a natural B_J -equivariant isomorphism

$$\operatorname{Gr}_{B_{J}}^{w_{J}}\left(\operatorname{Ind}_{L_{J}\cap P_{J\cap I_{w}J^{-1}(I)}}^{L_{J}}\tilde{\sigma}\right) \cong \operatorname{c-ind}_{L_{J}\cap P_{J\cap I_{w}J^{-1}(I)}}^{L_{J}\cap P_{J\cap I_{w}J^{-1}(I)}}\tilde{\sigma}. \tag{9}$$

Remark 2.2.2 (case $w_J = 1$). Note that

$$\operatorname{Gr}_{B}^{1}(\operatorname{c-ind}_{P_{I}^{-}}^{P_{I}^{-}I_{w}^{J}P_{J}}\sigma) \cong \operatorname{c-ind}_{P_{I}^{-}}^{P_{I}^{-}I_{w}^{J}B}\sigma$$

is a $P_{J\cap^l w^{J-1}(I)}$ -subrepresentation of c-ind $P_{P_I}^{I-l}w^{J}P_{J}$ σ and likewise

$$\operatorname{Gr}^1_{B_J}(\operatorname{Ind}_{L_J\cap P_{J\cap I_wJ^{-1}(I)}}^{L_J}\tilde{\sigma})\cong\operatorname{c-ind}_{L_J\cap P_{J\cap I_wJ^{-1}(I)}}^{L_J\cap P_{J\cap I_wJ^{-1}(I)}^{-1}B_J}\tilde{\sigma}$$

is an $L_J \cap P_{J \cap I_W^{J-1}(I)}$ -subrepresentation of $\operatorname{Ind}_{L_J \cap P_{J \cap I_W^{J-1}(I)}}^{L_J} \tilde{\sigma}$ (see Remark 2.1.3).

2.3. Computation of the associated graded representations. For each $w \in W$, we define a closed subgroup of U stable under conjugation by Z by setting

$$\boldsymbol{U}_w := \boldsymbol{U} \cap w^{-1} \boldsymbol{U} w$$

and we let $B_w \subseteq B$ be the closed subgroup $\mathcal{Z}U_w$. For any order on $\Phi^+ \cap w^{-1}(\Phi_0^+)$, the product induces an isomorphism of F-varieties

$$\prod_{\alpha \in \Phi^+ \cap w^{-1}(\Phi_0^+)} \boldsymbol{U}_{\alpha} \xrightarrow{\sim} \boldsymbol{U}_w. \tag{10}$$

Let $I \subseteq \Delta$ and $Iw \in IW$. We define closed subgroups of U_{Iw} stable under conjugation by \mathcal{Z} by setting

$$U'_{I_{w}} := U \cap {}^{I}w^{-1}N_{I}{}^{I}w, \quad U''_{I_{w}} := U \cap {}^{I}w^{-1}U_{I}{}^{I}w,$$

and we let $B''_{l_w} \subseteq B_{l_w}$ be the closed subgroup ZU''_{l_w} . We have semidirect products $U_{l_w} = U''_{l_w} \ltimes U'_{l_w}$ and $B_{l_w} = B''_{l_w} \ltimes U'_{l_w}$.

Let σ be an L_I -representation. The product induces an isomorphism of F-varieties

$$P_I^- \times \{^I w\} \times U_{I_w}' \xrightarrow{\sim} P_I^{-I} w B,$$

hence an A-linear isomorphism

$$\operatorname{c-ind}_{P_{r}^{-}}^{P_{l}^{-}l_{W}B} \sigma \cong \mathcal{C}_{\operatorname{c}}^{\operatorname{sm}}\left(U_{l_{w}}', \sigma^{l_{w}}\right) \tag{11}$$

via which U'_{l_w} acts on $C_c^{sm}(U'_{l_w}, \sigma^{l_w})$ by right translation and the action of $b'' \in B''_{l_w}$ on $f \in C_c^{sm}(U'_{l_w}, \sigma^{l_w})$ is given by

$$(b'' \cdot f)(u') = b'' \cdot f(b''^{-1}u'b'')$$

for all $u' \in U'_{I_m}$.

Let $J \subseteq \Delta$. We write ${}^Iw = {}^Iw^Jw_J$ with ${}^Iw^J \in {}^IW^J$ and $w_J \in W_J$. We define closed subgroups of N_J and U_J stable under conjugation by $\mathcal Z$ by setting

$$N_{J,^Iw} := N_J \cap U_{^Iw} = N_J \cap {}^Iw^{-1}U^Iw, \quad U_{J,w_I} := U_J \cap U_{^Iw} = U_J \cap {}^Iw^{-1}U^Iw = U_J \cap w_J^{-1}U_Jw_J,$$

the last equality resulting from (3), and we let $B_{J,w_J} \subseteq B_J$ be the closed subgroup $\mathbb{Z}U_{J,w_J}$. We have semidirect products $U_{l_w} = U_{J,w_J} \ltimes N_{J,l_w}$ and $B_{l_w} = B_{J,w_J} \ltimes N_{J,l_w}$. We define closed subgroups of N_{J,l_w} and U_{J,w_J} stable under conjugation by \mathbb{Z} by setting

$$\begin{split} N'_{J,^{I}w} &:= N_J \cap U'_{I_w} = N_J \cap {}^I w^{-1} N_I{}^I w, \\ N''_{J,^{I}w} &:= N_J \cap U''_{I_w} = N_J \cap {}^I w^{-1} U_I{}^I w, \\ U'_{J,w_J} &:= U_J \cap U'_{I_w} = U_J \cap {}^I w^{-1} N_I{}^I w, \\ U''_{J,w_J} &:= U_J \cap U''_{I_w} = U_J \cap {}^I w^{-1} U_I{}^I w, \end{split}$$

and we let $B''_{J,w_J} \subseteq B_J$ be the closed subgroup $\mathcal{Z}U''_{J,w_J}$. We have semidirect products $N_{J,l_w} = N''_{J,l_w} \ltimes N'_{J,l_w}$, $U_{J,w_J} = U''_{J,w_J} \ltimes U'_{J,w_J}$ and $B_{J,w_J} = B''_{J,w_J} \ltimes U'_{J,w_J}$. Note that U'_{J,w_J} and U''_{J,w_J} actually depend on l_w (not only on w_J).

Likewise, for any $L_{J\cap I_w^{J-1}(I)}$ -representation $\tilde{\sigma}$ and using Lemma 2.1.1 with I and J swapped and I^W^J inverted, the product induces an isomorphism of F-varieties

$$L_J \cap P_{J \cap lw^{J-1}(I)}^- \times \{w_J\} \times U'_{J,w_J} \xrightarrow{\sim} L_J \cap P_{J \cap lw^{J-1}(I)}^- w_J B_J,$$

hence an A-linear isomorphism

$$\operatorname{c-ind}_{L_{J} \cap P_{J_{\cap}}^{-} I_{w^{J-1}(I)}}^{L_{J} \cap P_{J_{\cap}}^{-} I_{w^{J-1}(I)}} \tilde{\sigma} \cong \mathcal{C}_{\operatorname{c}}^{\operatorname{sm}} \left(U'_{J,w_{J}}, \tilde{\sigma}^{w_{J}} \right)$$

$$\tag{12}$$

via which U'_{J,w_J} acts on $\mathcal{C}^{\mathrm{sm}}_{\mathrm{c}}(U'_{J,w_J},\tilde{\sigma}^{w_J})$ by right translation and the action of $b''\in B''_{J,w_J}$ on $f\in\mathcal{C}^{\mathrm{sm}}_{\mathrm{c}}(U'_{J,w_J},\tilde{\sigma}^{w_J})$ is given by

$$(b'' \cdot f)(u') = b'' \cdot f(b''^{-1}u'b'')$$

for all $u' \in U'_{J,w_J}$. In particular with $\tilde{\sigma} = \sigma^{lw^J}$, we have defined a natural smooth A-linear action of B_{J,w_J} on $C_c^{\rm sm}(U'_{J,w_J}, \sigma^{lw})$.

We have a semidirect product $U'_{l_w} = U'_{J,w_J} \ltimes N'_{J,l_w}$, so that (11) composed with the *A*-linear morphism defined by $f \mapsto (n' \mapsto (u' \mapsto f(u'n')))$ is an *A*-linear isomorphism

$$\operatorname{c-ind}_{P_{l}^{-}}^{P_{l}^{-}l_{W}B} \sigma \cong \mathcal{C}_{c}^{\operatorname{sm}}(N'_{J,l_{W}}, \mathcal{C}_{c}^{\operatorname{sm}}(U'_{J,w_{J}}, \sigma^{l_{W}}))$$

$$(13)$$

via which $N'_{J,lw}$ acts on $\mathcal{C}^{\mathrm{sm}}_{\mathrm{c}}(N'_{J,lw},\mathcal{C}^{\mathrm{sm}}_{\mathrm{c}}(U'_{J,w_J},\sigma^{lw}))$ by right translation, the action of $b\in B_{J,w_J}$ on $f\in\mathcal{C}^{\mathrm{sm}}_{\mathrm{c}}(N'_{J,lw},\mathcal{C}^{\mathrm{sm}}_{\mathrm{c}}(U'_{J,w_J},\sigma^{lw}))$ is given by

$$(b \cdot f)(n') = b \cdot f(b^{-1}n'b)$$

for all $n' \in N'_{I_{lw}}$ and the action of $N''_{I_{lw}}$ on $C_c^{sm}(N'_{I_{lw}}, C_c^{sm}(U'_{I,w_l}, \sigma^{lw}))$ is given by the following result.

Lemma 2.3.1. Let $f \in C_c^{sm}(N'_{J,l_w}, C_c^{sm}(U'_{J,w_J}, \sigma^{l_w}))$ and $n'' \in N''_{J,l_w}$. Via (13), the action of n'' on f is given by

$$(n'' \cdot f)(n')(u') = n'' \cdot f(u'^{-1}n''^{-1}u'n'n'')(u')$$

for all $n' \in N'_{I,I_w}$ and $u' \in U'_{I,w_I}$.

Proof. Let $n' \in N'_{J,l_w}$ and $u' \in U'_{J,w_J}$. We have

$${}^{I}wu'n'n'' = ({}^{I}wn''{}^{I}w^{-1}){}^{I}wu'(u'^{-1}n''^{-1}u'n'n'').$$

Thus, it is enough to check that $u'^{-1}n''^{-1}u'n'n'' \in N_J'$. Since $u' \in U_J$ and $n', n'' \in N_J$, we have $(u'^{-1}n''^{-1}u')n'n'' \in N_J$. Since $n'' \in {}^Iw^{-1}U_I{}^Iw$ and $n', u' \in {}^Iw^{-1}N_I{}^Iw$, we have $u'^{-1}(n''^{-1}(u'n')n'') \in {}^Iw^{-1}N_I{}^Iw$. Hence the result.

Remark 2.3.2 (case $w_J=1$). We can also give the action of $L_{J\cap^I w^{J-1}(I)}$ (which normalises $U'_{J,1}, N'_{J,^I w^J}, N''_{J,^I w^J}$, and thus $U'_{I_w J}, N_{J,^I w^J}$) on c-ind $P_I^{-I_w J} \sigma$ and c-ind $P_I^{-I_w J} \sigma$ and c-ind $P_I^{-I_w J} \sigma$ (see Remark 2.2.2) via (13) and (12) respectively, by replacing $P_{J, w_J} = P'_{J, w_J} \ltimes U'_{J, w_J}$ by $P_{J, w_J} = P_{J, w_J J} = P_{J, w_J} = P_{J, w_$

We end this subsection with some more notation.

Notation 2.3.3. For each $w \in W$, we let d_w be the integer $\dim_F(U/U_w)$ and $\delta_w \in X^*(S)$ be the algebraic character of the adjoint representation of S on $\det_F((\text{Lie }U)/(\text{Lie }U_w))$. Note that $d_w \geq \ell(w)$ and δ_w extends to an algebraic character of Z. For $\alpha \in \Delta$, we have $d_{s_\alpha} = \dim_F U_\alpha$ and $\delta_{s_\alpha} = d_{s_\alpha}\alpha$. We define a subset of Δ by setting

$$\Delta^1 := \left\{ \alpha \in \Delta \mid \dim_F \mathbf{U}_{\alpha} = 1 \right\}.$$

For $I \subseteq \Delta$, we put $I^1 := I \cap \Delta^1$.

Remark 2.3.4. For ${}^Iw^J \in {}^IW^J$, we have $U_J \subseteq U_{{}^Iw^J}$ and $L_{J \cap {}^Iw^{J-1}(I)}$ normalises $N_{J,{}^Iw^J}$. Thus, the inclusion $N_J \hookrightarrow U$ induces an isomorphism of F-varieties

$$N_J/N_{I I_w J} \xrightarrow{\sim} U/U_{I_w J}$$

and there is an adjoint action of $L_{J\cap l_W J^{-1}(I)}$ on $(\text{Lie } N_J)/(\text{Lie } N_{J,l_W J})$. Therefore, we have $d_{l_W J} = \dim_F (N_J/N_{J,l_W J})$ and $\delta_{l_W J}$ extends to an algebraic character of $L_{J\cap l_W J^{-1}(I)}$.

3. Derived ordinary parts

The aim of this section is to compute the derived ordinary parts of a parabolically induced representation. In Section 3.1, we show how to compute the cohomology of certain groups with a Hecke action from the cohomology of certain subgroups with the induced Hecke action, provided the latter satisfy some finiteness condition. In Section 3.2, we make a computation of cohomology and Hecke action on a compactly induced representation. In Section 3.3, we use the previous results to partially compute the derived ordinary parts of the graded representations associated with the Bruhat filtrations, we formulate a conjecture on the complete result and we prove it in many cases in low degree.

3.1. Cohomology, Hecke action and dévissage. Let \widetilde{L} be a linear algebraic F-group and \widetilde{N} be a unipotent algebraic F-group endowed with an action of \widetilde{L} that we identify with the conjugation in $\widetilde{L} \ltimes \widetilde{N}$. We let \widetilde{d} denote the integer $\dim_F \widetilde{N}$ and $\widetilde{\delta} \in X^*(\widetilde{L})$ denote the algebraic character of the adjoint representation of \widetilde{L} on $\det_F(\operatorname{Lie} \widetilde{N})$.

Let $\widetilde{L}^+ \subseteq \widetilde{L}$ be an open submonoid and $\widetilde{N}_0 \subseteq \widetilde{N}$ be a standard³ compact open subgroup stable under conjugation by \widetilde{L}^+ . If π is an $\widetilde{L}^+ \ltimes \widetilde{N}_0$ -representation,⁴ then the A-modules of \widetilde{N}_0 -cohomology $H^\bullet(\widetilde{N}_0, \pi)$ computed using locally constant cochains (or equivalently an N_0 -injective resolution of π ; see [Emerton 2010b, Proposition 2.2.6]) are naturally endowed with the Hecke action of \widetilde{L}^+ (denoted $\overset{\mathrm{H}}{\cdot}$), defined for every $\widetilde{l} \in \widetilde{L}^+$ as the composite

$$H^{\bullet}(\widetilde{N}_{0}, \pi) \to H^{\bullet}(\widetilde{l}\widetilde{N}_{0}\widetilde{l}^{-1}, \pi) \to H^{\bullet}(\widetilde{N}_{0}, \pi)$$

where the first morphism is induced by the action of \tilde{l} on π and the second morphism is the corestriction from $\tilde{l}\widetilde{N}_0\tilde{l}^{-1}$ to \widetilde{N}_0 (this defines a natural smooth A-linear action of \widetilde{L}^+ in degree 0 [Emerton 2010a, Lemma 3.1.4] that extends in higher degrees by universality of $H^{\bullet}(\widetilde{N}_0, -)$). We obtain a universal δ -functor

$$\mathrm{H}^{\bullet}(\widetilde{N}_{0},-):\mathrm{Mod}^{\mathrm{sm}}_{\widetilde{L}^{+}\ltimes\widetilde{N}_{0}}(A)\to\mathrm{Mod}^{\mathrm{sm}}_{\widetilde{L}^{+}}(A),$$

since an injective $\widetilde{L}^+ \ltimes \widetilde{N}_0$ -representation is \widetilde{N}_0 -acyclic [Emerton 2010b, Proposition 2.1.11; Hauseux 2016a, Lemme 3.1.1].

³The exponential map $\exp: \operatorname{Lie} \widetilde{N} \to \widetilde{N}$ is an isomorphism of *F*-varieties [Demazure and Gabriel 1970, Chapitre IV, §2, Proposition 4.1] and we say that \widetilde{N}_0 is *standard* if $\operatorname{Lie} \widetilde{N}_0 := \exp^{-1}(\widetilde{N}_0) \subseteq \operatorname{Lie} \widetilde{N}$ is a \mathbb{Z}_p -Lie subalgebra. The identity of \widetilde{N} admits a basis of neighbourhoods consisting of standard compact open subgroups [Emerton 2010b, Lemma 3.5.2]

⁴Given a *p*-adic Lie group *H* and an open submonoid $H^+ \subseteq H$, a representation of H^+ over *A* is *smooth* if its restriction to an open subgroup of *H* contained in H^+ is smooth.

Let $\widetilde{\mathbf{Z}} \subseteq \widetilde{\mathbf{L}}$ be a central split torus and $\widetilde{Z}^+ \subseteq \widetilde{Z}$ be the open submonoid $\widetilde{Z} \cap \widetilde{L}^+$. Since $\widetilde{\mathbf{Z}}$ is split, its adjoint representation on Lie \widetilde{N} is a direct sum of weights. We assume that there exists $\widetilde{\lambda} \in X_*(\widetilde{\mathbf{Z}})$ such that $\langle \widetilde{\mu}, \widetilde{\lambda} \rangle > 0$ for any weight $\widetilde{\mu}$ of $\widetilde{\mathbf{Z}}$ in Lie \widetilde{N} . We fix an element $\widetilde{z} := \widetilde{\lambda}(p^j) \in \widetilde{Z}$ with $j \in \mathbb{N}$ large enough so that \widetilde{z} is *strictly contracting* \widetilde{N}_0 , i.e., $(\widetilde{z}^i \widetilde{N}_0 \widetilde{z}^{-i})_{i \in \mathbb{N}}$ is a basis of neighbourhoods of the identity in \widetilde{N}_0 ; see [Emerton 2010b, Lemma 3.1.3] using the fact that $\operatorname{ord}_p(\widetilde{\mu}(\widetilde{z})) = \langle \widetilde{\mu}, \widetilde{\lambda} \rangle j$ for any weight $\widetilde{\mu}$ of $\widetilde{\mathbf{Z}}$ in Lie \widetilde{N}). In particular $\widetilde{z} \in \widetilde{Z}^+$.

If π is a \widetilde{Z}^+ -representation, we say that π is *locally* \widetilde{z} -finite if for every $v \in \pi$, the A-submodule $A[\widetilde{z}] \cdot v$ is of finite type, and we say that the action of \widetilde{z} on π is *locally nilpotent* if for every $v \in \pi$, there exists $i \in \mathbb{N}$ such that $\widetilde{z}^i \cdot v = 0$.

Lemma 3.1.1. Let π be a locally \tilde{z} -finite $\widetilde{L}^+ \ltimes \widetilde{N}_0$ -representation and $n \in \mathbb{N}$.

- (i) If $n = [F : \mathbb{Q}_p]\tilde{d}$, then the action of \tilde{z} on the kernel of the natural \tilde{L}^+ -equivariant surjection $\pi \otimes (\omega^{-1} \circ \tilde{\delta}) \twoheadrightarrow H^n(\tilde{N}_0, \pi)$ is locally nilpotent.
- (ii) If $n < [F : \mathbb{Q}_p]\tilde{d}$, then the Hecke action of \tilde{z} on $H^n(\widetilde{N}_0, \pi)$ is locally nilpotent.

Proof. We prove (i). The natural \widetilde{L}^+ -equivariant surjection in the statement is the composite

$$\pi \otimes (\omega^{-1} \circ \widetilde{\delta}) \twoheadrightarrow \pi_{\widetilde{N}_0} \otimes (\omega^{-1} \circ \widetilde{\delta}) \cong \mathrm{H}^{[F:\mathbb{Q}_p]\widetilde{d}}(\widetilde{N}_0, \pi)$$

where the first morphism is the natural projection onto the \widetilde{N}_0 -coinvariants of π and the second morphism is the natural isomorphism [Hauseux 2016b, (2.2)] which is due to Emerton (in loc. cit. $\tilde{\alpha} \in X^*(\operatorname{Res}_{F/\mathbb{Q}_p} \widetilde{L})$ is the algebraic character of the adjoint representation of $\operatorname{Res}_{F/\mathbb{Q}_p} \widetilde{L}$ on $\det_{\mathbb{Q}_p}(\operatorname{Lie}(\operatorname{Res}_{F/\mathbb{Q}_p} \widetilde{N}))$ so that $\tilde{\alpha} = \operatorname{Nrm}_{F/\mathbb{Q}_p} \circ \tilde{\delta}$ as \mathbb{Q}_p^{\times} -valued characters of \widetilde{L} , hence $\tilde{\alpha}^{-1} |\tilde{\alpha}|_p^{-1} = \omega^{-1} \circ \tilde{\delta}$ as \mathbb{Q}_p^{\times} -valued characters of \widetilde{L}). For every $v \in \pi$, there exists $i \in \mathbb{N}$ such that $\tilde{z}^i \widetilde{N}_0 \tilde{z}^{-i}$ fixes $A[\tilde{z}] \cdot v$ (since π is locally \tilde{z} -finite and \tilde{z} is strictly contracting \widetilde{N}_0), so that for all $\tilde{n} \in \widetilde{N}_0$ we have

$$\tilde{z}^i \cdot (\tilde{n} \cdot v - v) = (\tilde{z}^i \tilde{n} \tilde{z}^{-i}) \cdot (\tilde{z}^i \cdot v) - (\tilde{z}^i \cdot v) = 0.$$

Thus the action of \tilde{z} on the kernel of the above surjection is locally nilpotent.

We prove (ii). Let $(\tilde{\mu}_r)_{r\in \llbracket 0,m-1\rrbracket}$ be an enumeration of the weights of $\widetilde{\mathbf{Z}}$ in Lie \widetilde{N} such that the sequence $(\langle \tilde{\mu}_r, \tilde{\lambda} \rangle)_{r\in \llbracket 0,m-1\rrbracket}$ is increasing. If $\tilde{\mu}_i + \tilde{\mu}_j = \tilde{\mu}_r$ with $i,j,r\in \llbracket 0,m-1\rrbracket$, then $r>\max\{i,j\}$ (since $\langle \tilde{\mu}_r, \tilde{\lambda} \rangle > \max\{\langle \tilde{\mu}_i, \tilde{\lambda} \rangle, \langle \tilde{\mu}_j, \tilde{\lambda} \rangle\}$). Thus for all $r\in \llbracket 0,m\rrbracket$, the direct sum of the weight spaces corresponding to $\tilde{\mu}_r,\ldots,\tilde{\mu}_{m-1}$ is an ideal of Lie \tilde{N} stable under the adjoint action of $\tilde{\mathbf{Z}}$ and we let $\tilde{N}^{(r)}\subseteq \tilde{N}$ be the corresponding closed normal subgroup stable under conjugation by $\tilde{\mathbf{Z}}$, \tilde{d}_r denote the integer $\dim_F \tilde{N}^{(r)}$, $\tilde{\delta}_r \in X^*(\tilde{\mathbf{L}})$ denote the algebraic character of the adjoint representation of $\tilde{\mathbf{L}}$ on $\det_F(\operatorname{Lie} \tilde{N}^{(r)})$ and $\tilde{N}_0^{(r)}\subseteq \tilde{N}^{(r)}$ be the standard compact open subgroup $\tilde{N}^{(r)}\cap \tilde{N}_0$ stable under conjugation by $\tilde{\mathbf{Z}}^+$.

Let $r \in [0, m]$. We assume that $n < [F : \mathbb{Q}_p]\tilde{d}_r$ and we prove that the Hecke action of \tilde{z} on $H^n(\widetilde{N}_0^{(r)}, \pi)$ is locally nilpotent by induction on r. The result is trivial for r = m. We assume r < m and the result true for r + 1. We have a short exact sequence of topological groups

$$1 \to \widetilde{N}_0^{(r+1)} \to \widetilde{N}_0^{(r)} \to \widetilde{N}_0^{(r)}/\widetilde{N}_0^{(r+1)} \to 1.$$

The Lyndon–Hochschild–Serre spectral sequence corresponding to this dévissage is naturally a spectral sequence of \widetilde{L}^+ -representations (see [Hauseux 2016b, (2.3)])

$$H^{i}\left(\widetilde{N}_{0}^{(r)}/\widetilde{N}_{0}^{(r+1)}, H^{j}\left(\widetilde{N}_{0}^{(r+1)}, \pi\right)\right) \Rightarrow H^{i+j}\left(\widetilde{N}_{0}^{(r)}, \pi\right). \tag{14}$$

Let $i, j \in \mathbb{N}$ be such that i+j=n. If $j < [F:\mathbb{Q}_p] \tilde{d}_{r+1}$, the Hecke action of \tilde{z} on $\mathrm{H}^j(\widetilde{N}_0^{(r+1)},\pi)$ is locally nilpotent by the induction hypothesis; thus the Hecke action of \tilde{z} on $\mathrm{H}^i(\widetilde{N}_0^{(r)}/\widetilde{N}_0^{(r+1)},\mathrm{H}^j(\widetilde{N}_0^{(r+1)},\pi))$ is also locally nilpotent (since the image of a locally constant cochain is finite by compactness). If $j=[F:\mathbb{Q}_p]\tilde{d}_{r+1}$, then $i<[F:\mathbb{Q}_p](\tilde{d}_r-\tilde{d}_{r+1})$ and we deduce from (i) with $\widetilde{N}^{(r+1)}$ and j instead of \widetilde{N} and n respectively that $\mathrm{H}^j(\widetilde{N}_0^{(r+1)},\pi)$ is locally \tilde{z} -finite; thus the Hecke action of \tilde{z} on $\mathrm{H}^i(\widetilde{N}_0^{(r)}/\widetilde{N}_0^{(r+1)},\mathrm{H}^j(\widetilde{N}_0^{(r+1)},\pi))$ is locally nilpotent by the sublemma below with $\widetilde{\mu}=\widetilde{\mu}_r$ and $\widetilde{N}^{(r)}/\widetilde{N}^{(r+1)}$, $\mathrm{H}^j(\widetilde{N}_0^{(r+1)},\pi)$, i instead of \widetilde{N} , π , n respectively. If $j>[F:\mathbb{Q}_p]\widetilde{d}_{r+1}$, then $\mathrm{H}^j(\widetilde{N}_0^{(r+1)},\pi)=0$ by [Emerton 2010b, Lemma 3.5.4]; thus $\mathrm{H}^i(\widetilde{N}_0^{(r)}/\widetilde{N}_0^{(r+1)},\mathrm{H}^j(\widetilde{N}_0^{(r+1)},\pi))=0$. Using (14), we conclude that the action of \tilde{z} on $\mathrm{H}^n(\widetilde{N}_0^{(r)},\pi)$ is locally nilpotent.

Sublemma. Let π be a locally \tilde{z} -finite $\widetilde{Z}^+ \ltimes \widetilde{N}_0$ -representation, $\tilde{\mu} \in X^*(\widetilde{\mathbf{Z}})$ and $n \in \mathbb{N}$. Assume that the adjoint action of $\widetilde{\mathbf{Z}}$ on Lie \widetilde{N} factors through $\tilde{\mu}$. If $n < [F : \mathbb{Q}_p]\tilde{d}$, then the Hecke action of \tilde{z} on $H^n(\widetilde{N}_0, \pi)$ is locally nilpotent.

Proof. Let $\widetilde{S} \subseteq \operatorname{Res}_{F/\mathbb{Q}_p} \widetilde{Z}$ be the maximal split subtorus, $\widetilde{S} \subseteq \widetilde{Z}$ be the closed subgroup $\widetilde{S}(\mathbb{Q}_p)$ and $\widetilde{S}^+ \subseteq \widetilde{S}$ be the open submonoid $\widetilde{S} \cap \widetilde{Z}^+$. Every algebraic (co)character of \widetilde{Z} induces by restriction of scalars a (co)character of \widetilde{S} (since the image of a split torus by a morphism of algebraic groups is a split torus [Borel and Tits 1965, §1.4]). In particular, the restriction of $\widetilde{\lambda} : F^\times \to \widetilde{Z}$ to \mathbb{Q}_p^\times takes values in \widetilde{S} and the restriction of $\widetilde{\mu} : \widetilde{Z} \to F^\times$ to \widetilde{S} takes values in \mathbb{Q}_p^\times .

We deduce on the one hand that $\tilde{z} \in \widetilde{S}^+$, and on the other hand that the adjoint action of \widetilde{S} on $\operatorname{Lie}(\operatorname{Res}_{F/\mathbb{Q}_p} \widetilde{N})$ factors through an algebraic character so that any closed subgroup of $\operatorname{Res}_{F/\mathbb{Q}_p} \widetilde{N}$ is stable under conjugation by \widetilde{S} . Since $\operatorname{Res}_{F/\mathbb{Q}_p} \widetilde{N}$ is unipotent, there exists a composition series

$$\operatorname{Res}_{F/\mathbb{Q}_p} \widetilde{N} = \widetilde{N}^{(0)} \rhd \widetilde{N}^{(1)} \rhd \cdots \rhd \widetilde{N}^{([F:\mathbb{Q}_p]\widetilde{d})} = 1$$

whose successive quotients are isomorphic to the additive group over \mathbb{Q}_p and for all $r \in [0, [F : \mathbb{Q}_p]\tilde{d}]$, we let $\widetilde{N}^{(r)} \subseteq \widetilde{N}$ be the closed subgroup $\widetilde{N}^{(r)}(\mathbb{Q}_p)$ and $\widetilde{N}_0^{(r)} \subseteq \widetilde{N}^{(r)}$ be the standard compact open subgroup $\widetilde{N}^{(r)} \cap \widetilde{N}_0$ stable under conjugation by \widetilde{S}^+ .

Let $r \in [\![0, [F:\mathbb{Q}_p]\tilde{d}]\!]$. We assume that $n < [F:\mathbb{Q}_p]\tilde{d} - r$ and we prove by induction on r that the Hecke action of \tilde{z} on $H^n(\widetilde{N}_0^{(r)}, \pi)$ is locally nilpotent. The result is trivial for $r = [F:\mathbb{Q}_p]\tilde{d}$. We assume $r < [F:\mathbb{Q}_p]\tilde{d}$ and the result true for r+1. Since $\dim_{\mathbb{Q}_p}(\widetilde{N}^{(r)}/\widetilde{N}^{(r+1)}) = 1$, we have a short exact sequence of \widetilde{S}^+ -representations (see [Hauseux 2016b, (2.4)])

$$0 \to H^{1}(\widetilde{N}_{0}^{(r)}/\widetilde{N}_{0}^{(r+1)}, H^{n-1}(\widetilde{N}_{0}^{(r+1)}, \pi)) \to H^{n}(\widetilde{N}_{0}^{(r)}, \pi) \to H^{n}(\widetilde{N}_{0}^{(r+1)}, \pi)^{\widetilde{N}_{0}^{(r)}/\widetilde{N}_{0}^{(r+1)}} \to 0.$$
 (15)

The Hecke action of \tilde{z} on $\mathrm{H}^{n-1}(\widetilde{N}_0^{(r+1)},\pi)$ is locally nilpotent by the induction hypothesis; thus the Hecke action of \tilde{z} on $\mathrm{H}^1(\widetilde{N}_0^{(r)}/\widetilde{N}_0^{(r+1)},\mathrm{H}^{n-1}(\widetilde{N}_0^{(r+1)},\pi))$ is also locally nilpotent. If $n<[F:\mathbb{Q}_p]\tilde{d}-(r+1)$, then the Hecke action of \tilde{z} on $\mathrm{H}^n(\widetilde{N}_0^{(r+1)},\pi)$ is locally nilpotent by induction; thus the Hecke action of \tilde{z} on $\mathrm{H}^n(\widetilde{N}_0^{(r+1)},\pi)^{\widetilde{N}_0^{(r)}/\widetilde{N}_0^{(r+1)}}$ is also locally nilpotent. If $n=[F:\mathbb{Q}_p]\tilde{d}-(r+1)$, then we have a

natural \widetilde{S}^+ -equivariant surjection $\pi \otimes \widetilde{\mu}^{-n} | \widetilde{\mu}|_p^{-n} \to H^n(\widetilde{N}_0^{(r+1)}, \pi)$ [op. cit., (2.2)] and we deduce that $H^n(\widetilde{N}_0^{(r+1)}, \pi)$ is locally \widetilde{z} -finite. In this case, we put $\widetilde{N}_0'' := \widetilde{N}_0^{(r)} / \widetilde{N}_0^{(r+1)}$. For every $v \in H^n(\widetilde{N}_0^{(r+1)}, \pi)$, there exists $i \in \mathbb{N}$ such that $\widetilde{z}^i \widetilde{N}_0'' \widetilde{z}^{-i}$ fixes $A[\widetilde{z}] \cdot v$, so that for all $j \in \mathbb{N}$ we have

$$\begin{split} \tilde{z}^{i+j} & \stackrel{\mathrm{H}}{\cdot} v = \sum_{\tilde{n}'' \in \widetilde{N}_0''/\tilde{z}^{i+j} \widetilde{N}_0'' \tilde{z}^{-(i+j)}} \tilde{n}'' \cdot (\tilde{z}^{i+j} \cdot v) \\ &= (\tilde{z}^i \widetilde{N}_0'' \tilde{z}^{-i} : \tilde{z}^{i+j} \widetilde{N}_0'' \tilde{z}^{-(i+j)}) \sum_{\tilde{n}'' \in \widetilde{N}_0''/\tilde{z}^i \widetilde{N}_0'' \tilde{z}^{-i}} \tilde{n}'' \cdot (\tilde{z}^{i+j} \cdot v) \\ &= (\widetilde{N}_0'' : \tilde{z}^j \widetilde{N}_0'' \tilde{z}^{-j}) \sum_{\tilde{n}'' \in \widetilde{N}_0''/\tilde{z}^i \widetilde{N}_0'' \tilde{z}^{-i}} \tilde{n}'' \cdot (\tilde{z}^{i+j} \cdot v). \end{split}$$

Now \widetilde{N}_0'' is an infinite pro-p group, \widetilde{z} is strictly contracting \widetilde{N}_0'' and A is Artinian. Thus $(\widetilde{N}_0'':\widetilde{z}^j\widetilde{N}_0''\widetilde{z}^{-j})$ is zero in A for $j \in \mathbb{N}$ large enough. Therefore, the Hecke action of \widetilde{z} on $H^n(\widetilde{N}_0^{(r+1)},\pi)^{\widetilde{N}_0^{(r)}/\widetilde{N}_0^{(r+1)}}$ is locally nilpotent. Using (15), we conclude that the Hecke action of \widetilde{z} on $H^n(\widetilde{N}_0^{(r)},\pi)$ is locally nilpotent. \square

Let $\widetilde{N}'\subseteq\widetilde{N}$ be a closed subgroup such that $\mathrm{Lie}\ \widetilde{N}'\subseteq\mathrm{Lie}\ \widetilde{N}$ is a direct sum of weight spaces of $\widetilde{\mathbf{Z}}$. We stress that \widetilde{N}' need not be normal. Since $\widetilde{\mathbf{Z}}$ is central in $\widetilde{\mathbf{L}}$, $\mathrm{Lie}\ \widetilde{N}'$ is stable under the adjoint action of $\widetilde{\mathbf{L}}$; thus \widetilde{N}' is stable under conjugation by $\widetilde{\mathbf{L}}$. We let \widetilde{d}' denote the integer $\dim_F \widetilde{N}'$ and $\widetilde{\delta}' \in \mathrm{X}^*(\widetilde{\mathbf{L}})$ denote the algebraic character of the adjoint representation of $\widetilde{\mathbf{L}}$ on $\det_F(\mathrm{Lie}\ \widetilde{N}')$. We let $\widetilde{N}_0'\subseteq\widetilde{N}'$ be the standard compact open subgroup $\widetilde{N}'\cap\widetilde{N}_0$ stable under conjugation by $\widetilde{\mathbf{L}}^+$.

Proposition 3.1.2. Let π be an $\widetilde{L}^+ \ltimes \widetilde{N}_0$ -representation. For all $n \in \mathbb{N}$, there is a natural \widetilde{L}^+ -equivariant morphism

$$H^{n-[F:\mathbb{Q}_p](\tilde{d}-\tilde{d}')}(\widetilde{N}'_0,\pi)\otimes(\omega^{-1}\circ(\tilde{\delta}-\tilde{\delta}'))\to H^n(\widetilde{N}_0,\pi).$$

Furthermore, the Hecke action of \tilde{z} on its kernel and cokernel is locally nilpotent if the \widetilde{L}^+ -representations $H^{\bullet}(\widetilde{N}'_0, \pi)$ are locally \tilde{z} -finite.

Proof. Let $(\tilde{\mu}_r)_{r\in \llbracket 0,m-m'-1\rrbracket}$ be an enumeration of the weights of $\widetilde{\mathbf{Z}}$ in $(\text{Lie }\widetilde{N})/(\text{Lie }\widetilde{N}')$ such that the sequence $(\langle \tilde{\mu}_r, \tilde{\lambda} \rangle)_{r\in \llbracket 0,m-m'-1\rrbracket}$ is increasing and $(\tilde{\mu}_r)_{r\in \llbracket m-m',m-1\rrbracket}$ be an enumeration of the weights of $\widetilde{\mathbf{Z}}$ in Lie \widetilde{N}' such that the sequence $(\langle \tilde{\mu}_r, \tilde{\lambda} \rangle)_{r\in \llbracket m-m',m-1\rrbracket}$ is increasing. If $\tilde{\mu}_i + \tilde{\mu}_j = \tilde{\mu}_r$ with $i,j,r\in \llbracket 0,m-1\rrbracket$, then $r>\min\{i,j\}$ (since $\langle \tilde{\mu}_r, \tilde{\lambda} \rangle > \max\{\langle \tilde{\mu}_i, \tilde{\lambda} \rangle, \langle \tilde{\mu}_j, \tilde{\lambda} \rangle\}$). Thus for all $r\in \llbracket 0,m-m'\rrbracket$, the direct sum of the weight spaces corresponding to $\tilde{\mu}_r,\ldots,\tilde{\mu}_{m-1}$ is a Lie subalgebra of Lie \tilde{N} stable under the adjoint action of $\tilde{\mathbf{Z}}$ and we use the notations $\tilde{N}^{(r)}, \tilde{d}_r, \tilde{\delta}_r$ and $\tilde{N}_0^{(r)}$ as in the proof of Lemma 3.1.1(ii). Moreover for all $r\in \llbracket 0,m-m'-1\rrbracket$, Lie $\tilde{N}^{(r+1)}$ is an ideal of Lie $\tilde{N}^{(r)}$ so that $\tilde{N}^{(r+1)}$ is a normal subgroup of $\tilde{N}^{(r)}$.

Let $r \in [0, m-m']$. We prove by induction on r that for all $n \in \mathbb{N}$, there is a natural \widetilde{L}^+ -equivariant morphism

$$H^{n-[F:\mathbb{Q}_p](\tilde{d}_r-\tilde{d}')}(\tilde{N}'_0,\pi)\otimes(\omega^{-1}\circ(\tilde{\delta}_r-\tilde{\delta}'))\to H^n(\tilde{N}_0^{(r)},\pi). \tag{16}$$

The result is trivial for r = m - m'. We assume r < m - m' and the result true for r + 1. Let $n \in \mathbb{N}$. Since $\dim_F(\widetilde{N}^{(r)}/\widetilde{N}^{(r+1)}) = \widetilde{d}_r - \widetilde{d}_{r+1}$, we deduce from [Emerton 2010b, Lemma 3.5.4] that (14) yields

a natural \widetilde{L}^+ -equivariant morphism

$$\mathbf{H}^{[F:\mathbb{Q}_p](\tilde{d}_r - \tilde{d}_{r+1})} (\widetilde{N}_0^{(r)} / \widetilde{N}_0^{(r+1)}, \mathbf{H}^{n-[F:\mathbb{Q}_p](\tilde{d}_r - \tilde{d}_{r+1})} (\widetilde{N}_0^{(r+1)}, \pi)) \to \mathbf{H}^n (\widetilde{N}_0^{(r)}, \pi)$$
(17)

whose kernel and cokernel are built out of subquotients of $\mathrm{H}^i \big(\widetilde{N}_0^{(r)} / \widetilde{N}_0^{(r+1)}, \mathrm{H}^j (\widetilde{N}_0^{(r+1)}, \pi) \big)$ with $i, j \in \mathbb{N}$ such that $i < [F:\mathbb{Q}_p](\tilde{d}_r - \tilde{d}_{r+1})$. Furthermore, Lemma 3.1.1(i) with \widetilde{N} , π , n and $\widetilde{\delta}$ replaced by $\widetilde{N}^{(r)} / \widetilde{N}^{(r+1)}$, $\mathrm{H}^{n-[F:\mathbb{Q}_p](\tilde{d}_r - \tilde{d}_{r+1})}(\widetilde{N}_0^{(r+1)}, \pi)$, $[F:\mathbb{Q}_p](\tilde{d}_r - \tilde{d}_{r+1})$ and $\widetilde{\delta}_r - \widetilde{\delta}_{r+1}$ respectively yields a natural \widetilde{L}^+ -equivariant surjection

$$H^{n-[F:\mathbb{Q}_{p}](\tilde{d}_{r}-\tilde{d}_{r+1})}(\widetilde{N}_{0}^{(r+1)},\pi) \otimes (\omega^{-1} \circ (\tilde{\delta}_{r}-\tilde{\delta}_{r+1}))
\to H^{[F:\mathbb{Q}_{p}](\tilde{d}_{r}-\tilde{d}_{r+1})}(\widetilde{N}_{0}^{(r)}/\widetilde{N}_{0}^{(r+1)},H^{n-[F:\mathbb{Q}_{p}](\tilde{d}_{r}-\tilde{d}_{r+1})}(\widetilde{N}_{0}^{(r+1)},\pi)).$$
(18)

Finally, by the induction hypothesis with $n - [F : \mathbb{Q}_p](\tilde{d}_r - \tilde{d}_{r+1})$ instead of n, there is a natural \widetilde{L}^+ equivariant morphism

$$H^{n-[F:\mathbb{Q}_p](\tilde{d}_r-\tilde{d}')}(\widetilde{N}_0',\pi)\otimes(\omega^{-1}\circ(\widetilde{\delta}_r-\widetilde{\delta}'))\to H^{n-[F:\mathbb{Q}_p](\tilde{d}_r-\tilde{d}_{r+1})}(\widetilde{N}_0^{(r+1)},\pi)\otimes(\omega^{-1}\circ(\widetilde{\delta}_r-\widetilde{\delta}_{r+1})). \tag{19}$$

The composition of (17), (18) and (19) yields the natural \widetilde{L}^+ -equivariant morphism (16).

Now, we assume that the \widetilde{L}^+ -representations $\operatorname{H}^\bullet(\widetilde{N}_0',\pi)$ are locally \widetilde{z} -finite and we prove by induction on r that for all $n\in\mathbb{N}$, the Hecke action of \widetilde{z} on the kernel and cokernel of (16) is locally nilpotent, or equivalently that the localisation of (16) with respect to $\widetilde{z}^\mathbb{N}$ is an isomorphism. The result is trivial for r=m-m'. We assume r< m-m' and the result true for r+1. Let $n\in\mathbb{N}$. By composition, it is enough to prove that the Hecke action of \widetilde{z} on the kernels and cokernels of (17), (18) and (19) is locally nilpotent. By the induction hypothesis with j instead of n, the Hecke action of \widetilde{z} on the kernel and cokernel of the natural \widetilde{L}^+ -equivariant morphism

$$\mathrm{H}^{j-[F:\mathbb{Q}_p](\tilde{d}_{r+1}-\tilde{d}')}(\widetilde{N}'_0,\pi)\otimes(\omega^{-1}\circ(\widetilde{\delta}_{r+1}-\widetilde{\delta}'))\to\mathrm{H}^{j}(\widetilde{N}_0^{(r+1)},\pi)$$

is locally nilpotent for all $j \in \mathbb{N}$. With $j = n - [F : \mathbb{Q}_p](\tilde{d}_r - \tilde{d}_{r+1})$, we deduce that the Hecke action of \tilde{z} on the kernel and cokernel of (19) is locally nilpotent. Furthermore, we deduce that $H^j(\widetilde{N}_0^{(r+1)}, \pi)$ is locally \tilde{z} -finite for all $j \in \mathbb{N}$ and we use Lemma 3.1.1 with $\widetilde{N}^{(r)}/\widetilde{N}^{(r+1)}$, $H^j(\widetilde{N}_0^{(r+1)}, \pi)$ and i instead of \widetilde{N} , π and n respectively: we deduce from (i) with $i = [F : \mathbb{Q}_p](\tilde{d}_r - \tilde{d}_{r+1})$ that the Hecke action of \tilde{z} on the kernel of (18) is locally nilpotent, and we deduce from (ii) that the Hecke action of \tilde{z} on the kernel and cokernel of (17) is locally nilpotent (since the Hecke action of \tilde{z} on $H^i(\widetilde{N}_0^{(r)}/\widetilde{N}_0^{(r+1)}, H^j(\widetilde{N}_0^{(r+1)}, \pi))$ is locally nilpotent for all $i, j \in \mathbb{N}$ such that $i < [F : \mathbb{Q}_p](\tilde{d}_r - \tilde{d}_{r+1})$).

We end this subsection by reviewing and generalising the construction of Emerton's δ -functor of derived ordinary parts [Emerton 2010b, §3.3]. Let $\widetilde{Z}_{\widetilde{L}}$ denote the centre of \widetilde{L} . Assume that $\widetilde{Z}_{\widetilde{L}}^{\circ}$ is a torus, that $\widetilde{Z}_{\widetilde{L}}$ is generated by $\widetilde{Z}_{\widetilde{L}}^{+} := \widetilde{Z}_{\widetilde{L}} \cap \widetilde{L}^{+}$ as a group, and that \widetilde{L} is generated by \widetilde{L}^{+} and $\widetilde{Z}_{\widetilde{L}}^{\circ}$ as a monoid. Then, the product induces a group isomorphism $\widetilde{L}^{+} \times_{\widetilde{Z}_{\widetilde{L}}^{+}} \widetilde{Z}_{\widetilde{L}} \xrightarrow{\sim} \widetilde{L}$ [Emerton 2006, Proposition 3.3.6]. Thus, for any \widetilde{L}^{+} -representation π , the A-module $\operatorname{Hom}_{A[\widetilde{Z}_{\widetilde{L}}^{+}]}(A[\widetilde{Z}_{\widetilde{L}}], \pi)^{\widetilde{Z}_{\widetilde{L}}^{-1.fin}}$ is naturally an \widetilde{L} -representation [Emerton 2010a, Lemma 3.1.7]. Therefore, we obtain an A-linear left-exact functor $\operatorname{Mod}_{\widetilde{L}^{+}}^{\operatorname{sm}}(A) \to \operatorname{Mod}_{\widetilde{L}}^{\operatorname{sm}}(A)^{\widetilde{Z}_{\widetilde{L}}^{-1.fin}}$ which commutes with inductive limits [op. cit., Lemma 3.2.2].

Remark 3.1.3. Let $\tilde{z} \in \widetilde{Z}_{\widetilde{L}}^+$. Assume moreover that $\widetilde{Z}_{\widetilde{L}}$ is generated by $\widetilde{Z}_{\widetilde{L}}^+$ and \tilde{z}^{-1} as a monoid. Then, for any locally finite $\widetilde{Z}_{\widetilde{L}}^+$ -representation π , there is a natural $\widetilde{Z}_{\widetilde{L}}$ -equivariant isomorphism

$$\operatorname{Hom}_{A[\widetilde{Z}_{\widetilde{L}}^{+}]}(A[\widetilde{Z}_{\widetilde{L}}],\pi)^{\widetilde{Z}_{\widetilde{L}}-\operatorname{l.fin}} \xrightarrow{\sim} A[\widetilde{z}^{\pm 1}] \otimes_{A[\widetilde{z}]} \pi$$

[Emerton 2010b, Lemma 3.2.1]. Thus, the functor $\operatorname{Hom}_{A[\widetilde{Z}_{\widetilde{L}}^+]}(A[\widetilde{Z}_{\widetilde{L}}],-)^{\widetilde{Z}_{\widetilde{L}}-1.\operatorname{fin}}$ restricted to the category $\operatorname{Mod}_{\widetilde{L}^+}^{\operatorname{sm}}(A)^{\widetilde{Z}_{\widetilde{L}}^+-1.\operatorname{fin}}$ is isomorphic to the localisation with respect to $\widetilde{z}^{\mathbb{N}}$. In particular, it is exact.

Definition 3.1.4. For a connected linear algebraic F-group \widetilde{P} with unipotent radical \widetilde{N} such that $\widetilde{P} \cong \widetilde{L} \ltimes \widetilde{N}$, we define A-linear functors $\operatorname{Mod}_{\widetilde{P}}^{\operatorname{sm}}(A) \to \operatorname{Mod}_{\widetilde{L}}^{\operatorname{sm}}(A)^{\widetilde{Z}_{\widetilde{L}}-1.\operatorname{fin}}$ which commute with inductive limits by setting

 $\mathrm{H}^{ullet}\mathrm{Ord}_{\widetilde{P}}:=\mathrm{Hom}_{A[\widetilde{Z}_{\widetilde{I}}^+]}\left(A[\widetilde{Z}_{\widetilde{L}}],\mathrm{H}^{ullet}(\widetilde{N}_0,-)\right)^{\widetilde{Z}_{\widetilde{L}}-\mathrm{l.fin}}.$

If $\widetilde{B} \subseteq \widetilde{P}$ is a connected closed subgroup containing \widetilde{N} and $\widetilde{Z}_{\widetilde{L}}$, then $\widetilde{B}_{\widetilde{L}} := \widetilde{B} \cap \widetilde{L}$ is generated by $\widetilde{B}_{\widetilde{L}}^+ := \widetilde{B}_{\widetilde{L}} \cap \widetilde{L}^+$ and $\widetilde{Z}_{\widetilde{L}}$ as a monoid, so that $\operatorname{H}^{\bullet}\operatorname{Ord}_{\widetilde{P}}$ naturally extend to A-linear functors $\operatorname{Mod}_{\widetilde{B}}^{\operatorname{sm}}(A) \to \operatorname{Mod}_{\widetilde{B}_{\widetilde{L}}}^{\operatorname{sm}}(A)^{\widetilde{Z}_{\widetilde{L}}-1.\operatorname{fin}}$ which commute with inductive limits.

3.2. Computations on the associated graded representations. Let $J \subseteq \Delta$. We fix a totally decomposed⁵ standard compact open subgroup $N_{J,0} \subseteq N_J$ and we define an open submonoid of L_J by setting

$$L_J^+ := \{ l \in L_J \mid l N_{J,0} l^{-1} \subseteq N_{J,0} \}.$$

We let $Z_J^+ \subseteq Z_J$ be the open submonoid $Z_J \cap L_J^+$. Note that Z_J is generated by Z_J^+ as a group and L_J is generated by L_J^+ and Z_J as a monoid [Emerton 2006, Proposition 3.3.2]. Moreover, any $\lambda \in X_*(S)$ corresponding to P_J has its image contained in the maximal split subtorus S_J of Z_J° and satisfies $\langle \alpha, \lambda \rangle > 0$ for all $\alpha \in \Phi^+ \setminus \Phi_J^+$; thus the assumption of Section 3.1 with $\widetilde{N} = N_J$ and $\widetilde{Z} = S_J$ is satisfied. We fix $z \in Z_J^+$ strictly contracting $N_{J,0}$ (equivalently Z_J is generated by Z_J^+ and z^{-1} as a monoid).

Let $I \subseteq \Delta$ and ${}^Iw \in {}^IW$. We write ${}^Iw = {}^Iw^Jw_J$ with ${}^Iw^J \in {}^IW^J$ and $w_J \in W_J$. Let σ be an L_I -representation. We set⁶

$$\pi_{I_w} := \operatorname{c-ind}_{P_I^-}^{P_I^{-I_w}B} \sigma.$$

We use the notation of Section 2.3. The subgroup $N_{J,^lw}\subseteq N_J$ is stable under conjugation by \pmb{B}_{J,w_J} , and we have a semidirect product $N_{J,^lw}=N''_{J,^lw}\ltimes N'_{J,^lw}$. The subgroup $N'_{J,^lw}$ is stable under conjugation by \pmb{B}_{J,w_J} , and we endow $N''_{J,^lw}$ (which may not be stable under conjugation by \pmb{B}_{J,w_J}) with the quotient action of \pmb{B}_{J,w_J} via the isomorphism $N''_{J,^lw}\cong N_{J,^lw}/N'_{J,^lw}$. We let $N_{J,^lw,0}\subseteq N_{J,^lw}$ (resp. $N'_{J,^lw,0}\subseteq N'_{J,^lw},N''_{J,^lw,0}\subseteq N''_{J,^lw}$) be the totally decomposed standard compact open subgroup $N_{J,^lw}\cap N_{J,0}$ (resp. $N'_{J,^lw}\cap N_{J,0},N''_{J,^lw}\cap N_{J,0}$) and $B^+_{J,w_J}\subseteq B_{J,w_J}$ be the open submonoid $B_{J,w_J}\cap L^+_J$. Since $N_{J,^lw,0}$ is totally decomposed, we have a short exact sequence of topological groups

$$1 \to N'_{J,l_{w},0} \to N_{J,l_{w},0} \to N''_{J,l_{w},0} \to 1.$$
 (20)

⁵Given a closed subgroup $\widetilde{U} \subseteq U$ stable under conjugation by S, we say that a compact open subgroup $\widetilde{U}_0 \subseteq \widetilde{U}$ is *totally decomposed* if the product induces a homeomorphism $\prod_{\alpha \in \Phi_0^+} (U_\alpha \cap \widetilde{U}_0) \xrightarrow{\sim} \widetilde{U}_0$ for any order on Φ_0^+ (e.g., $\widetilde{U}_0 = \widetilde{U} \cap K$ where $K \subseteq G$ is a maximal compact subgroup which is special with respect to \mathcal{Z} [Henniart and Vignéras 2015, §6.6, Remark 2]).

⁶The naturality of a morphism involving π_{I_w} will mean its functoriality with respect to σ .

In particular, $N''_{J,l_{w,0}}$ is stable under the quotient action of B^+_{J,w_J} on N''_{J,l_w} .

Lemma 3.2.1. For all $n \in \mathbb{N}$, the inflation map is a natural B_{L,w_l}^+ -equivariant isomorphism

$$\mathrm{H}^n\left(N_{J,l_w,0}^{\prime\prime},\pi_{l_w}^{N_{J,l_w,0}^{\prime}}\right) \xrightarrow{\sim} \mathrm{H}^n(N_{J,l_w,0},\pi_{l_w}).$$

Proof. The Lyndon–Hochschild–Serre spectral sequence corresponding to (20) is naturally a spectral sequence of $B_{Im_I}^+$ -representations [Hauseux 2016b, (2.3)]

$$H^{i}(N''_{J,l_{w},0}, H^{j}(N'_{J,l_{w},0}, \pi_{l_{w}})) \Rightarrow H^{i+j}(N_{J,l_{w},0}, \pi_{l_{w}}).$$
 (21)

The inflation maps are the edge maps of (21) for j=0; thus they are B_{J,w_J}^+ -equivariant and in order to prove that they are bijective, it is enough to show that (21) degenerates, i.e., that $H^j(N'_{J,l_w,0}, \pi_{l_w}) = 0$ for all integers j>0.

Since the left cosets $N'_{J,l_w}/N'_{J,l_w,0}$ form an open partition of N'_{J,l_w} , we deduce from (13) a natural $N'_{J,l_w,0}$ -equivariant isomorphism

$$\pi_{l_w} \cong \bigoplus_{n' \in N'_{J,l_w}/N'_{J,l_{w,0}}} \mathcal{C}^{\mathrm{sm}} \big(n' N'_{J,l_{w,0}}, \mathcal{C}^{\mathrm{sm}}_{\mathrm{c}} \big(U'_{J,w_J}, \sigma^{l_w} \big) \big)$$

where $N'_{J,^Iw,0}$ acts by right translation on the terms of the direct sum. The latter are $N'_{J,^Iw,0}$ -acyclic by Shapiro's lemma (since they are induced discrete $A[N'_{J,^Iw,0}]$ -modules) and the $N'_{J,^Iw,0}$ -cohomology commutes with direct sums (since the image of a locally constant cochain is finite by compactness); thus π_{I_w} is $N'_{I^Iw,0}$ -acyclic.

There is a natural smooth A-linear action of $B''_{J,w_J} \ltimes (U'_{J,w_J} \times N''_{J,l_w})$ on $C_c^{sm}(U'_{J,w_J}, \sigma^{l_w})$: we already defined the action of $B_{J,w_J} = B''_{J,w_J} \ltimes U'_{J,w_J}$ in Section 2.3 and we define the action of $n'' \in N''_{J,l_w}$ on $f \in C_c^{sm}(U'_{J,w_J}, \sigma^{l_w})$ by setting

$$(n'' \cdot f)(u') := n'' \cdot f(u')$$

for all $u' \in U'_{J,w_J}$.

Lemma 3.2.2. For all $n \in \mathbb{N}$, there is a natural B_{J,w_J}^+ -equivariant morphism

$$\mathrm{H}^{n}\left(N_{J,^{l}w,0}^{"},\pi_{l_{w}}^{N_{J,^{l}w,0}^{'}}\right)\rightarrow\mathrm{H}^{n}\left(N_{J,^{l}w,0}^{"},\mathcal{C}_{\mathrm{c}}^{\mathrm{sm}}\left(U_{J,w_{J}}^{'},\sigma_{u}^{lw}\right)\right)$$

such that the Hecke action of z on its kernel and cokernel is locally nilpotent.

Proof. We will implicitly make use of the isomorphism (13). For each $n' \in N'_{J,l_w}/N'_{J,l_w,0}$, evaluation at n' induces a natural A-linear surjection

$$\operatorname{ev}_{n'}: \pi_{I_w}^{N'_{J,I_{w,0}}} \mathcal{C}_{\operatorname{c}}^{\operatorname{sm}} (U'_{J,w_J}, \sigma^{I_w}).$$

We define a natural A-linear surjection

$$\operatorname{Ev} := \sum_{n' \in N'_{J,I_{w}}/N'_{J,I_{w},0}} \operatorname{ev}_{n'} : \pi_{I_{w}}^{N'_{J,I_{w},0}} \twoheadrightarrow \mathcal{C}^{\operatorname{sm}}_{\operatorname{c}} \big(U'_{J,w_{J}}, \sigma^{I_{w}} \big).$$

We prove that Ev is B_{J,w_J}^+ -equivariant: for any $f \in \pi_{I_w}^{N_{J,I_w,0}'}$ and $b \in B_{J,w_J}^+$, we have

$$\begin{split} \operatorname{Ev}(b \overset{\operatorname{H}}{\cdot} f) &= \sum_{n' \in N'_{J, I_{w}} / N'_{J, I_{w,0}}} \sum_{n'_{0} \in N'_{J, I_{w,0}} / b N'_{J, I_{w,0}} b^{-1}} (n'_{0}b \cdot f)(n') \\ &= \sum_{n' \in N'_{J, I_{w}} / N'_{J, I_{w,0}}} \sum_{n'_{0} \in N'_{J, I_{w,0}} / b N'_{J, I_{w,0}} b^{-1}} b \cdot f(b^{-1}n'n'_{0}b) \\ &= \sum_{n' \in N'_{J, I_{w}} / b N'_{J, I_{w,0}} b^{-1}} b \cdot f(b^{-1}n'b) \\ &= b \cdot \operatorname{Ev}(f) \end{split}$$

where the last equality results from the change of variable $n'\mapsto bn'b^{-1}$. We prove that Ev is also $N''_{J,^Iw,0}$ -equivariant: for any $f\in\pi^{N'_{J,^Iw,0}}_{I_w}$, $n''\in N''_{J,^Iw,0}$ and $u'\in U'_{J,w_J}$, we have

$$\operatorname{Ev}(n'' \cdot f)(u') = \sum_{n' \in N'_{J,I_w}/N'_{J,I_{w,0}}} (n'' \cdot f)(n')(u')$$

$$= \sum_{n' \in N'_{J,I_w}/N'_{J,I_{w,0}}} n'' \cdot f(u'^{-1}n''^{-1}u'n'n'')(u')$$

$$= n'' \cdot \operatorname{Ev}(f)(u'),$$

where the last equality results from the fact that when n' runs among $N'_{J,^Iw}/N'_{J,^Iw,0}$ we have

$$u'^{-1}n''^{-1}u'n'n'' = (u'^{-1}n''^{-1}u'n'')(n''^{-1}n'n'')$$

with on the one hand $n''^{-1}n'n''$ running among $N'_{J,^Iw}/N'_{J,^Iw,0}$ and on the other hand $u'^{-1}n''^{-1}u'n'' \in N'_{J,^Iw}$ being constant. We deduce that Ev induces natural B^+_{J,w_J} -equivariant morphisms in $N''_{J,^Iw,0}$ -cohomology.

We prove that the Hecke action of z on the kernel of Ev is locally nilpotent: for any $f \in \pi_{l_w}^{N_{J,l_w,0}}$ there exists $i \in \mathbb{N}$ such that $\operatorname{supp}(f) \subseteq z^{-i}N'_{J,l_w,0}z^i$ (since z is strictly contracting $N'_{J,l_w,0}$ which is open in N'_{J,l_w}), thus for any $n' \in N'_{J,l_w}/N'_{J,l_w,0}$, we have

$$\begin{split} (z^{i} & \stackrel{\mathrm{H}}{\cdot} f)(n') = \sum_{n'_{0} \in N'_{J,I_{w,0}}/z^{i}N'_{J,I_{w,0}}z^{-i}} (n'_{0}z^{i} \cdot f)(n') \\ &= \sum_{n'_{0} \in N'_{J,I_{w,0}}/z^{i}N'_{J,I_{w,0}}z^{-i}} z^{i} \cdot f\left((z^{-i}n'z^{i})(z^{-i}n'_{0}z^{i})\right) \\ &= \begin{cases} z^{i} \cdot \operatorname{Ev}(f) & \text{if } n' \in N'_{J,I_{w,0}}, \\ 0 & \text{if } n' \notin N'_{J,I_{w,0}}. \end{cases} \end{split}$$

Using the long exact sequence of $N''_{J,^Iw,0}$ -cohomology, we deduce that the Hecke action of z on the kernels and cokernels of the morphisms induced by Ev in $N''_{J,^Iw,0}$ -cohomology is locally nilpotent. \square

The subgroup $B''_{J,w_J} \subseteq B_{J,w_J}$ normalises N''_{J,l_w} and the conjugation action coincides with the action induced by the quotient action of B_{J,w_J} on N''_{J,l_w} . We define an open submonoid $B''_{J,w_J} \subseteq B''_{J,w_J}$ by setting

$$B_{J,w_J}^{\prime\prime+} := \big\{b^{\prime\prime} \in B_{J,w_J}^{\prime\prime} \; \big| \; b^{\prime\prime} N_{J,^Iw,0}^{\prime\prime} b^{\prime\prime-1} \subseteq N_{J,^Iw,0}^{\prime\prime} \big\}.$$

Lemma 3.2.3. We have $B_{J,w_I}^+ \subseteq B_{J,w_I}''^+ \ltimes U_{J,w_I}'$.

Proof. We have a semidirect product $B_{J,w_J} = B''_{J,w_J} \ltimes U'_{J,w_J}$. Let $b \in B^+_{J,w_J}$. We write b = b''u' with $b'' \in B''_{J,w_J}$ and $u' \in U'_{J,w_J}$. We prove that $b'' \in B''_{J,w_J}$. Let $n'' \in N''_{J,l_{w,0}}$. Proceeding as in the proof of Lemma 2.3.1, we see that $u'n''u'^{-1}n''^{-1} \in N'_{J,l_w}$ so that $u'n''u'^{-1} = n'n''$ with $n' \in N'_{J,l_w}$. Thus $bn''b^{-1} = (b''n'b''^{-1})(b''n''b''^{-1}) \in N_{J,l_{w,0}}$, and since $N_{J,l_{w,0}}$ is totally decomposed, we deduce that $b''n''b''^{-1} \in N''_{J,l_{w,0}}$.

Lemma 3.2.4. For all $n \in \mathbb{N}$, there is a natural $B_{J,w_J}^{"+} \ltimes U_{J,w_J}'$ -equivariant isomorphism

$$H^{n}\left(N_{J,^{l}w,0}^{"},\mathcal{C}_{c}^{sm}(U_{J,w_{J}}^{'},\sigma^{^{l}w})\right) \cong \mathcal{C}_{c}^{sm}\left(U_{J,w_{J}}^{'},H^{n}(N_{J,^{l}w,0}^{"},\sigma^{^{l}w})\right).$$

Proof. Let $\tilde{\sigma}$ be a $B_{J,w_J}''^+ \ltimes N_{J,I_{w,0}}''$ -representation. The A-modules $\operatorname{H}^{\bullet}\left(N_{J,I_{w,0}}'',\mathcal{C}_{\operatorname{c}}^{\operatorname{sm}}(U_{J,w_J}',\tilde{\sigma})\right)$ and $\mathcal{C}_{\operatorname{c}}^{\operatorname{sm}}\left(U_{J,w_J}',\operatorname{H}^{\bullet}(N_{J,I_{w,0}}'',\tilde{\sigma})\right)$ are naturally $B_{J,w_J}''^+ \ltimes U_{J,w_J}'$ -representations. The identity of $\mathcal{C}_{\operatorname{c}}^{\operatorname{sm}}(U_{J,w_J}',\tilde{\sigma})$ induces a natural U_{J,w_J}' -equivariant isomorphism

$$\iota: \mathcal{C}_{\operatorname{c}}^{\operatorname{sm}} \big(U_{J,w_{I}}^{\prime}, \tilde{\sigma}\big)^{N_{J,I_{w,0}}^{\prime\prime}} \xrightarrow{\sim} \mathcal{C}_{\operatorname{c}}^{\operatorname{sm}} \big(U_{J,w_{I}}^{\prime}, \tilde{\sigma}^{N_{J,I_{w,0}}^{\prime\prime}}\big).$$

We prove that ι is also $B_{J,w_J}^{\prime\prime+}$ -equivariant: for any $f\in\mathcal{C}_{\operatorname{c}}^{\operatorname{sm}}(U_{J,w_J}^\prime,\tilde{\sigma})^{N_{J,I_{w,0}}^{\prime\prime}},b^{\prime\prime}\in B_{J,w_J}^{\prime\prime+}$ and $u^\prime\in U_{J,w_J}^\prime,$ we have

$$\iota(b'' \stackrel{\mathrm{H}}{\cdot} f)(u') = \sum_{n'' \in N''_{J, I_{w,0}}/b'' N''_{J, I_{w,0}}b''-1} \iota(n''b'' \cdot f)(u')$$

$$= \sum_{n'' \in N''_{J, I_{w,0}}/b'' N''_{J, I_{w,0}}b''-1} n''b'' \cdot \iota(f)(b''^{-1}u'b'')$$

$$= b'' \stackrel{\mathrm{H}}{\cdot} \iota(f)(b''^{-1}u'b'')$$

$$= (b'' \cdot \iota(f))(u').$$

We will prove that deriving ι yields the desired isomorphisms with $\tilde{\sigma} = \sigma^{l_w}$.

The functor $C_c^{sm}(U'_{J,w_J},-)$ is A-linear and exact and the δ -functor $H^{\bullet}(N''_{J,I_{w,0}},-)$ is universal; thus, denoting by R^{\bullet} the right derived functors on the category $\operatorname{Mod}_{B''_{J,w_J} \ltimes N''_{J,I_{w,0}}}^{sm}(A)$, we have morphisms of functors

$$\begin{split} \mathbf{R}^{\bullet} \big(\mathcal{C}^{\mathrm{sm}}_{\mathbf{c}} \big(U'_{J,w_J}, (-)^{N''_{J,I_{w,0}}} \big) \big) & \cong \mathcal{C}^{\mathrm{sm}}_{\mathbf{c}} \big(U'_{J,w_J}, \mathbf{H}^{\bullet} (N''_{J,I_{w,0}}, -) \big), \\ \mathbf{R}^{\bullet} \big(\mathcal{C}^{\mathrm{sm}}_{\mathbf{c}} \big(U'_{J,w_J}, - \big)^{N''_{J,I_{w,0}}} \big) & \to \mathbf{H}^{\bullet} \big(N''_{J,I_{w,0}}, \mathcal{C}^{\mathrm{sm}}_{\mathbf{c}} \big(U'_{J,w_J}, - \big) \big). \end{split}$$

In order to show that the second one is also an isomorphism, it is enough to prove that $C_c^{sm}(U'_{J,w_J},-)$ takes injective objects of $\mathrm{Mod}_{B''_{J,w_J} \ltimes N''_{J,I_{w,0}}}^{sm}(A)$ to $N''_{J,I_{w,0}}$ -acyclic objects. If $\tilde{\sigma}$ is an A-module, then we

have a natural $N_{J,I_{W},0}^{\prime\prime}$ -equivariant isomorphism

$$\mathcal{C}_{\operatorname{c}}^{\operatorname{sm}}\big(U_{J,w_{J}}',\mathcal{C}^{\operatorname{sm}}\big(N_{J,l_{w},0}'',\tilde{\sigma}\big)\big) \cong \mathcal{C}^{\operatorname{sm}}\big(N_{J,l_{w},0}'',\mathcal{C}_{\operatorname{c}}^{\operatorname{sm}}\big(U_{J,w_{J}}',\tilde{\sigma}\big)\big),$$

so $\mathcal{C}_{\mathbf{c}}^{\mathrm{sm}}\left(U'_{J,w_J},\mathcal{C}^{\mathrm{sm}}(N''_{J,I_{w,0}},\tilde{\sigma})\right)$ is $N''_{J,I_{w,0}}$ -acyclic. Now if $\tilde{\sigma}$ is an injective object of $\mathrm{Mod}_{B''_{J,w_J} \ltimes N''_{J,I_{w,0}}}^{\mathrm{sm}}(A)$, it is also an injective object of $\mathrm{Mod}_{N''_{J,I_{w,0}}}^{\mathrm{sm}}(A)$ [Emerton 2010b, Proposition 2.1.11; Hauseux 2016a, Lemme 3.1.1]; thus the natural $N''_{J,I_{w,0}}$ -equivariant injection $\tilde{\sigma} \hookrightarrow \mathcal{C}^{\mathrm{sm}}(N''_{J,I_{w,0}},\tilde{\sigma})$ defined by $v \mapsto (n'' \mapsto n'' \cdot v)$ admits an $N''_{J,I_{w,0}}$ -equivariant retraction, so that $\tilde{\sigma}$ is a direct factor of $\mathcal{C}^{\mathrm{sm}}(N''_{J,I_{w,0}},\tilde{\sigma})$, and therefore $\tilde{\sigma}$ is $N''_{J,I_{w,0}}$ -acyclic.

We now assume that σ is locally admissible.

Lemma 3.2.5. For all $n \in \mathbb{N}$, there is a natural $B_{J,w_J}^{\prime\prime+} \ltimes U_{J,w_J}^{\prime}$ -equivariant morphism

$$\mathcal{C}_{\mathbf{c}}^{\mathrm{sm}}\left(U_{J,w_{J}}^{\prime}, \mathbf{H}^{n}\left(N_{J,l_{w},0}^{\prime\prime}, \sigma^{l_{w}}\right)\right) \rightarrow \mathbf{c} \cdot \mathrm{ind}_{L_{J} \cap P_{J \cap l_{w}J - 1(I)}^{\prime\prime}}^{L_{J} \cap P_{J \cap l_{w}J - 1(I)}^{\prime\prime}} \left(\mathbf{H}^{n} \mathrm{Ord}_{L_{I} \cap P_{I \cap l_{w}J(J)}^{\prime\prime}} \sigma\right)^{l_{w}J}$$

such that the action of z on its kernel and cokernel is locally nilpotent.

Proof. We have natural $B_{J,w_I}^{"+}$ -equivariant isomorphisms

$$H^{\bullet}(N''_{J,l_{w},0}, \sigma^{l_{w}}) \cong H^{\bullet}({}^{l}wN''_{J,l_{w},0}{}^{l}w^{-1}, \sigma)^{l_{w}}.$$
 (22)

Since ${}^{I}wN_{J,^{I}w}^{"}{}^{I}w^{-1} = U_{I} \cap {}^{I}w^{J}N_{J}{}^{I}w^{J-1}$ is the unipotent radical of $L_{I} \cap P_{I \cap {}^{I}w^{J}(J)}$ (see Lemma 2.1.1(iii)), we define an open submonoid of $L_{I \cap {}^{I}w^{J}(J)}$ by setting

$$L^+_{I\cap {}^I\!w^J(J)} := \big\{ l \in L_{I\cap {}^I\!w^J(J)} \; \big| \; l^I\!w N_{J,{}^I\!w,0}'' {}^I\!w^{-1} l^{-1} \subseteq {}^I\!w N_{J,{}^I\!w,0}'' {}^I\!w^{-1} \big\}.$$

We have ${}^IwB_{J,w_J}^{\prime\prime+}{}^Iw^{-1}={}^IwB_{J,w_J}^{\prime\prime}{}^Iw^{-1}\cap L_{I\cap^Iw^J(J)}^+$. We let $Z_{I\cap^Iw^J(J)}^+\subseteq Z_{I\cap^Iw^J(J)}$ be the open submonoid $Z_{I\cap^Iw^J(J)}\cap L_{I\cap^Iw^J(J)}^+$. Since σ is locally admissible, $H^{\bullet}({}^IwN_{J,^Iw,0}^{\prime\prime}{}^Iw^{-1},\sigma)$ is locally $Z_{I\cap^Iw^J(J)}^+$ -finite [Emerton 2010b, Theorem 3.4.7(1)], and thus locally ${}^Iwz^Iw^{-1}$ -finite. Note that ${}^Iwz^Iw^{-1}\in Z_{I\cap^Iw^J(J)}^+$ is strictly contracting ${}^IwN_{J,^Iw,0}^{\prime\prime}{}^Iw^{-1}$. Therefore, localising with respect to $({}^Iwz^Iw^{-1})^{\mathbb{N}}$ gives rise to $L_{I\cap^Iw^J(J)}^+$ -equivariant morphisms

$$\operatorname{H}^{\bullet}({}^{I}wN_{J,{}^{I}w,0}^{"}{}^{I}w^{-1},\sigma) \to \operatorname{H}^{\bullet}\operatorname{Ord}_{L_{I}\cap P_{I\cap I_{w}J(J)}}\sigma$$

such that the action of ${}^{I}wz{}^{I}w^{-1}$ on their kernels and cokernel is locally nilpotent (see Remark 3.1.3). Using (22), we deduce $B_{J,w_{I}}^{\prime\prime+}$ -equivariant morphisms

$$\operatorname{H}^{\bullet}\left(N_{J,^{l}w,0}^{"},\sigma^{^{l}w}\right) \to \left(\operatorname{H}^{\bullet}\operatorname{Ord}_{L_{I}\cap P_{I\cap^{I}w^{J}(J)}}\sigma\right)^{^{l}w}$$

such that the action of z on their kernels and cokernels is locally nilpotent. Applying the functor $C_c^{\rm sm}(U'_{J,w_J},-)$, we obtain $B''^+_{J,w_J} \ltimes U'_{J,w_J}$ -equivariant morphisms

$$\mathcal{C}_{\mathbf{c}}^{\mathrm{sm}}\left(U_{J,w_{J}}^{\prime}, \mathbf{H}^{\bullet}\left(N_{J,^{I}w,0}^{\prime\prime}, \sigma^{^{I}w}\right)\right) \to \mathcal{C}_{\mathbf{c}}^{\mathrm{sm}}\left(U_{J,w_{J}}^{\prime}, \left(\mathbf{H}^{\bullet}\mathrm{Ord}_{L_{I}\cap P_{I\cap I_{w}^{J}(J)}}\sigma\right)^{^{I}w}\right)$$

such that the action of z on their kernel and cokernel is still locally nilpotent (because the functions in their sources and targets have finite images). We conclude using the inverse of the B_{J,w_J} -equivariant isomorphism (12) with $\tilde{\sigma} = (H^{\bullet}Ord_{L_I \cap P_{I \cap I_w J_{(I)}}} \sigma)^{I_w J}$.

We combine the previous results into the following one.

Proposition 3.2.6. Let σ be a locally admissible L_I -representation and ${}^Iw \in {}^IW$. We write ${}^Iw = {}^Iw^Jw_J$ with ${}^Iw^J \in {}^IW^J$ and $w_J \in W_J$. For all $n \in \mathbb{N}$, there is a natural B_{J,w_J}^+ -equivariant morphism

$$\mathrm{H}^{n}\big(N_{J,^{l}w,0},\operatorname{c-ind}_{P_{I}^{-}}^{P_{I}^{-}lwB}\sigma\big) \to \operatorname{c-ind}_{L_{J}\cap P_{J\cap l_{w}J^{-1}(I)}^{-}}^{L_{J}\cap P_{J\cap l_{w}J^{-1}(I)}^{-}} \big(\mathrm{H}^{n}\operatorname{Ord}_{L_{I}\cap P_{I\cap l_{w}J(J)}}\sigma\big)^{lw^{J}}$$

such that the action of z on its kernel and cokernel is locally nilpotent. Furthermore, this morphism is even $L_J^+ \cap P_{J \cap l_W J^{-1}(J)}$ -equivariant when $w_J = 1$ (see Remark 2.2.2).

Proof. Combining Lemmas 3.2.1, 3.2.2, 3.2.4, 3.2.5, and using Lemma 3.2.3, we obtain the desired morphism. If $w_J = 1$, then the previous lemmas and their proofs are true verbatim with $L_J \cap P_{J \cap l_w J^{-1}(I)}$ and $L_{J \cap l_w J^{-1}(I)}$ instead of B_{J,w_J} and B''_{J,w_J} respectively (see Remark 2.3.2); thus the morphism is $L_J^+ \cap P_{J \cap l_w J^{-1}(I)}$ -equivariant.

3.3. Computations on parabolically induced representations. Let $I, J \subseteq \Delta$, σ be a locally admissible L_I -representation and $n \in \mathbb{N}$. For any lower set ${}^IW_1^J \subseteq {}^IW^J$, the natural P_J -equivariant injection $\operatorname{Fil}_{P_J}^{IW_J^I}(\operatorname{Ind}_{P_I^-}^G\sigma) \hookrightarrow \operatorname{Ind}_{P_I^-}^G\sigma$ induces an L_J -equivariant morphism

$$H^n \operatorname{Ord}_{P_J} \left(\operatorname{Fil}_{P_J}^{IW_J^J} (\operatorname{Ind}_{P_I^-}^G \sigma) \right) \to H^n \operatorname{Ord}_{P_J} (\operatorname{Ind}_{P_I^-}^G \sigma),$$
 (23)

and by taking its image we define an L_I -subrepresentation

$$\operatorname{Fil}_{P_J}^{{}^IW_J^I}\left(\operatorname{H}^n\operatorname{Ord}_{P_J}(\operatorname{Ind}_{P_J^-}^G\sigma)\right)\subseteq \operatorname{H}^n\operatorname{Ord}_{P_J}(\operatorname{Ind}_{P_J^-}^G\sigma).$$

Proposition 3.3.1. The L_J -subrepresentations $\operatorname{Fil}_{P_J}^{\bullet}(\operatorname{H}^n\operatorname{Ord}_{P_J}(\operatorname{Ind}_{P_J}^G-\sigma))$ form a natural filtration of $\operatorname{H}^n\operatorname{Ord}_{P_J}(\operatorname{Ind}_{P_J^-}^G\sigma)$ indexed by ${}^IW^J$. Furthermore, for all ${}^Iw^J \in {}^IW^J$ there is a natural L_J -equivariant isomorphism

$$\operatorname{Gr}_{P_J}^{I_W^J}\left(\operatorname{H}^n\operatorname{Ord}_{P_J}(\operatorname{Ind}_{P_I^-}^G\sigma)\right) \cong \operatorname{H}^n\operatorname{Ord}_{P_J}\left(\operatorname{c-ind}_{P_I^-}^{P_I^{-1}w^JP_J}\sigma\right).$$

Proof. First, we prove for any lower sets ${}^IW_2^J \subseteq {}^IW_1^J \subseteq {}^IW^J$, the short exact sequence of P_J -representations

$$0 \to \operatorname{Fil}_{P_J}^{l_{W_2^J}}(\operatorname{Ind}_{P_I^-}^G \sigma) \to \operatorname{Fil}_{P_J}^{l_{W_1^J}}(\operatorname{Ind}_{P_I^-}^G \sigma) \to \operatorname{Fil}_{P_J}^{l_{W_1^J}}(\operatorname{Ind}_{P_I^-}^G \sigma) / \operatorname{Fil}_{P_J}^{l_{W_2^J}}(\operatorname{Ind}_{P_I^-}^G \sigma) \to 0$$
 (24)

induces a short exact sequence of L_J -representations

$$0 \to \operatorname{H}^{n}\operatorname{Ord}_{P_{J}}\left(\operatorname{Fil}_{P_{J}}^{IW_{2}^{J}}(\operatorname{Ind}_{P_{I}^{-}}^{G}\sigma)\right) \to \operatorname{H}^{n}\operatorname{Ord}_{P_{J}}\left(\operatorname{Fil}_{P_{J}}^{IW_{1}^{J}}(\operatorname{Ind}_{P_{I}^{-}}^{G}\sigma)\right)$$

$$\to \operatorname{H}^{n}\operatorname{Ord}_{P_{J}}\left(\operatorname{Fil}_{P_{J}}^{IW_{1}^{J}}(\operatorname{Ind}_{P_{I}^{-}}^{G}\sigma)/\operatorname{Fil}_{P_{J}}^{IW_{2}^{J}}(\operatorname{Ind}_{P_{I}^{-}}^{G}\sigma)\right) \to 0. \quad (25)$$

In particular, (23) is injective and (7) induces the isomorphism in the statement.

Let $N_{J,0} \subseteq N_J$, $L_J^+ \subseteq L_J$, $Z_J^+ \subseteq Z_J$ and $z \in Z_J^+$ be as in Section 3.2. Proceeding as in the proof of [Hauseux 2016a, Proposition 2.2.3], we see that the first nontrivial morphism of (24) induces an injection

in $N_{J,0}$ -cohomology. Using the long exact sequence of $N_{J,0}$ -cohomology, we deduce that (24) induces a short exact sequence of L_J^+ -representations

$$0 \to \operatorname{H}^{n}\left(N_{J,0}, \operatorname{Fil}_{P_{J}}^{IW_{J}^{J}}(\operatorname{Ind}_{P_{l}^{-}}^{G}\sigma)\right) \to \operatorname{H}^{n}\left(N_{J,0}, \operatorname{Fil}_{P_{J}}^{IW_{J}^{J}}(\operatorname{Ind}_{P_{l}^{-}}^{G}\sigma)\right)$$

$$\to \operatorname{H}^{n}\left(N_{J,0}, \operatorname{Fil}_{P_{J}}^{IW_{J}^{J}}(\operatorname{Ind}_{P_{l}^{-}}^{G}\sigma)/\operatorname{Fil}_{P_{J}}^{IW_{J}^{J}}(\operatorname{Ind}_{P_{l}^{-}}^{G}\sigma)\right) \to 0. \quad (26)$$

Since σ is locally admissible, $\operatorname{Ind}_{P_l^-}^G \sigma$ is locally admissible [Emerton 2010a, Proposition 4.1.7]; thus $\operatorname{H}^{\bullet}(N_{J,0},\operatorname{Ind}_{P_l^-}^G \sigma)$ is locally Z_J^+ -finite [Emerton 2010b, Theorem 3.4.7(1)]. We deduce that each term of (26) is locally Z_J^+ -finite (as a subquotient). We conclude that localising (26) with respect to $z^{\mathbb{N}}$ yields (25) (see Remark 3.1.3).

We now prove that $\operatorname{Fil}_{P_J}^{\bullet}(\operatorname{H}^n\operatorname{Ord}_{P_J}(\operatorname{Ind}_{P_l^-}^G\sigma))$ is a filtration of $\operatorname{H}^n\operatorname{Ord}_{P_J}(\operatorname{Ind}_{P_l^-}^G\sigma)$ indexed by ${}^IW^J$. Since ${}^IW^J$ is finite and $\operatorname{Fil}_{P_J}^{\bullet}(\operatorname{H}^n\operatorname{Ord}_{P_J}(\operatorname{Ind}_{P_l^-}^G\sigma))$ is inclusion-preserving with $\operatorname{Fil}_{P_J}^{\emptyset}(\operatorname{H}^n\operatorname{Ord}_{P_J}(\operatorname{Ind}_{P_l^-}^G\sigma))=0$ and $\operatorname{Fil}_{P_J}^{IW^J}(\operatorname{H}^n\operatorname{Ord}_{P_J}(\operatorname{Ind}_{P_l^-}^G\sigma))=\operatorname{H}^n\operatorname{Ord}_{P_J}(\operatorname{Ind}_{P_l^-}^G\sigma)$ by construction, it remains to prove that for any lower sets ${}^IW_1^J, {}^IW_2^J\subseteq {}^IW^J$, the natural short exact sequence of P_J -representations

$$0 \rightarrow \mathrm{Fil}_{P_J}^{{}^I\!W_1^J \cap {}^I\!W_2^J}(\mathrm{Ind}_{P_l^-}^G\sigma) \rightarrow \mathrm{Fil}_{P_J}^{{}^I\!W_1^J}(\mathrm{Ind}_{P_l^-}^G\sigma) \oplus \mathrm{Fil}_{P_J}^{{}^I\!W_2^J}(\mathrm{Ind}_{P_l^-}^G\sigma) \rightarrow \mathrm{Fil}_{P_J}^{{}^I\!W_1^J \cup {}^I\!W_2^J}(\mathrm{Ind}_{P_l^-}^G\sigma) \rightarrow 0$$

induces a short exact sequence of L_J -representations

$$0 \to \operatorname{Fil}_{P_{J}}^{IW_{1}^{J} \cap IW_{2}^{J}} \left(\operatorname{H}^{n} \operatorname{Ord}_{P_{J}} (\operatorname{Ind}_{P_{I}^{-}}^{G} \sigma) \right) \to \operatorname{Fil}_{P_{J}}^{IW_{1}^{J}} \left(\operatorname{H}^{n} \operatorname{Ord}_{P_{J}} (\operatorname{Ind}_{P_{I}^{-}}^{G} \sigma) \right) \oplus \operatorname{Fil}_{P_{J}}^{IW_{2}^{J}} \left(\operatorname{H}^{n} \operatorname{Ord}_{P_{J}} (\operatorname{Ind}_{P_{I}^{-}}^{G} \sigma) \right) \\ \to \operatorname{Fil}_{P_{J}}^{IW_{1}^{J} \cup IW_{2}^{J}} \left(\operatorname{H}^{n} \operatorname{Ord}_{P_{J}} (\operatorname{Ind}_{P_{I}^{-}}^{G} \sigma) \right) \to 0.$$

This follows from the same arguments as above.

Let ${}^Iw^J \in {}^IW^J$. For any lower set $W_J' \subseteq {}^{J\cap {}^Iw^{J-1}(I)}W_J$, the natural B-equivariant (resp. $P_{J\cap {}^Iw^{J-1}(I)}$ -equivariant when $W_J' = \{1\}$; see Remark 2.2.2) injection $\operatorname{Fil}_B^{W_J'}(\operatorname{c-ind}_{P_I^-}^{P_I^-Iw^JP_J}\sigma) \hookrightarrow \operatorname{c-ind}_{P_I^-}^{P_I^-Iw^JP_J}\sigma$ induces a B_J -equivariant (resp. $L_J \cap P_{J\cap {}^Iw^{J-1}(I)}$ -equivariant when $W_J' = \{1\}$) morphism

$$H^{n}\operatorname{Ord}_{P_{J}}\left(\operatorname{Fil}_{B}^{W'_{J}}\left(\operatorname{c-ind}_{P_{I}^{-}}^{P_{I}^{-}l_{w}^{J}P_{J}}\sigma\right)\right) \to H^{n}\operatorname{Ord}_{P_{J}}\left(\operatorname{c-ind}_{P_{I}^{-}}^{P_{I}^{-}l_{w}^{J}P_{J}}\sigma\right),\tag{27}$$

and by taking its image we define a B_J -subrepresentation (resp. $L_J \cap P_{J \cap l_W J^{-1}(I)}$ -subrepresentation when $W'_J = \{1\}$)

$$\operatorname{Fil}_{B}^{W'_{J}}\left(\operatorname{H}^{n}\operatorname{Ord}_{P_{J}}\left(\operatorname{c-ind}_{P_{I}^{-}}^{P_{I}^{-}lw^{J}P_{J}}\sigma\right)\right)\subseteq \operatorname{H}^{n}\operatorname{Ord}_{P_{J}}\left(\operatorname{c-ind}_{P_{I}^{-}}^{P_{I}^{-}lw^{J}P_{J}}\sigma\right).$$

Proceeding as in the proof of Proposition 3.3.1 and using (8), we prove that (27) is injective and the following result.

Proposition 3.3.2. The B_J -subrepresentations $\operatorname{Fil}^\bullet_{\mathcal{B}}(\operatorname{H}^n\operatorname{Ord}_{P_J}(\operatorname{c-ind}_{P_I^{-1}w^JP_J}^{P_I^{-1}w^JP_J}\sigma))$ form a natural filtration of $\operatorname{H}^n\operatorname{Ord}_{P_J}(\operatorname{c-ind}_{P_I^{-1}}^{P_I^{-1}w^JP_J}\sigma)$ indexed by $J\cap^{l_wJ-1}(I)W_J$. Furthermore, for all $w_J\in J\cap^{l_wJ-1}(I)W_J$ there is a natural B_J -equivariant isomorphism

$$\operatorname{Gr}_{B}^{w_{J}}(\operatorname{H}^{n}\operatorname{Ord}_{P_{J}}(\operatorname{c-ind}_{P_{I}^{-}}^{P_{I}^{-}l_{W}^{J}P_{J}}\sigma)) \cong \operatorname{H}^{n}\operatorname{Ord}_{P_{J}}(\operatorname{c-ind}_{P_{I}^{-}}^{P_{I}^{-}l_{W}^{J}w_{J}B}\sigma)$$

which is even $L_J \cap P_{J \cap I_W^{J-1}(I)}$ -equivariant when $w_J = 1$ (see Remark 2.2.2).

We now state the main result of this section using Notation 2.3.3 and Remark 2.3.4.

Theorem 3.3.3. Let σ be a locally admissible L_I -representation, ${}^Iw^J \in {}^IW^J$ and $n \in \mathbb{N}$. For all $w_J \in {}^{J \cap {}^Iw^{J-1}(I)}W_J$, there is a natural B_{J,w_J} -equivariant isomorphism

$$\operatorname{Gr}_{B}^{w_{J}}(\operatorname{H}^{n}\operatorname{Ord}_{P_{J}}(\operatorname{c-ind}_{P_{I}^{-}}^{P_{I}^{-}l_{W}^{J}P_{J}}\sigma))$$

$$\cong \operatorname{Gr}_{B_J}^{w_J} \left(\operatorname{Ind}_{L_J \cap P_{J \cap I_w J - 1}(J)}^{L_J} \left(\left(\operatorname{H}^{n - [F:\mathbb{Q}_p] d_{I_w J}} \operatorname{Ord}_{L_I \cap P_{I \cap I_w J}(J)} \sigma \right)^{I_w J} \otimes (\omega^{-1} \circ \delta_{I_w J}) \right) \right)$$

which is even $L_J \cap P_{J \cap I_W J^{-1}(I)}$ -equivariant when $w_J = 1$ (see Remark 2.2.2).

Proof. We use the notation of Section 2.3. We let $w_J \in {}^{J \cap {}^I w^{J-1}(I)} W_J$ and we put ${}^I w := {}^I w^J w_J$. We let $N_{J,0} \subseteq N_J$, $L_J^+ \subseteq L_J$, $Z_J^+ \subseteq Z_J$, $z \in Z_J^+$ and π_{I_w} be as in Section 3.2. In the course of the proof of Proposition 3.3.2, we see that $H^n(N_{J,0}, \pi_{I_w})$ is locally Z_J^+ -finite (as we saw it for $H^n(N_{J,0}, \operatorname{c-ind}_{P_I^-}^{P_I^- I w^J P_J} \sigma)$ in the course of the proof of Proposition 3.3.1).

Since σ is locally admissible, the $L_{I\cap^I w^J(J)}$ -representations H ullet Ord $_{L_I\cap P_{I\cap^I w^J(J)}}\sigma$ are locally admissible by [Emerton 2010b, Theorem 3.4.7(2)]; thus locally $Z_{I\cap^I w^J(J)}$ -finite by [Emerton 2010a, Lemma 2.3.4]. Therefore, the B_J -representations

$$\operatorname{c-ind}_{L_{J}\cap P^{-}_{I\cap I_{w}J^{-1}(I)}}^{L_{J}\cap P^{-}_{J\cap I_{w}J^{-1}(I)}} (\operatorname{H}^{\bullet}\operatorname{Ord}_{L_{I}\cap P_{I\cap I_{w}J}(J)} \sigma)^{I_{w}J}$$

are locally Z_J -finite; thus locally z-finite. We deduce from Proposition 3.2.6 that the B_{J,w_J}^+ -representations $H^{\bullet}(N_{J,^Iw,0},\pi_{I_w})$ are locally z-finite and that there is a natural B_{J,w_J}^+ -equivariant (resp. $L_J^+ \cap P_{J \cap ^I w^{J^{-1}}(I)}$ -equivariant when $w_J=1$) morphism

$$\mathrm{H}^{n-[F:\mathbb{Q}_p]d_{I_w^J}}(N_{J,{}^I\!w,0},\pi_{{}^I\!w})\otimes(\omega^{-1}\circ\delta_{{}^I\!w^J})^{w_J}$$

$$\rightarrow \left(\operatorname{c-ind}_{L_{J} \cap P_{J \cap I_{w}J^{-1}(I)}^{I_{J} \cap I_{w}J^{-1}(I)}}^{L_{J} \cap P_{J \cap I_{w}J^{-1}(I)}^{-1} w_{J}B_{J}} \left(\operatorname{H}^{n-[F:\mathbb{Q}_{p}]d_{I_{w}J}}\operatorname{Ord}_{L_{I} \cap P_{I \cap I_{w}J(J)}}\sigma\right)^{I_{w}J}\right) \otimes (\omega^{-1} \circ \delta_{I_{w}J})^{w_{J}}$$
(28)

such that the action of z on its kernel and cokernel is locally nilpotent.

Using Proposition 3.1.2 with $\widetilde{L} = B_{J,w_J}$ (resp. $\widetilde{L} = L_J \cap P_{J \cap I_w J^{-1}(I)}$ when $w_J = 1$), $\widetilde{N} = N_J$, $\widetilde{N}' = N_{J,I_w}$ (so that $\widetilde{d} - \widetilde{d}' = d_{I_w J}$ and $\widetilde{\delta} - \widetilde{\delta}' = w_J^{-1}(\delta_{I_w J})$ since conjugation by w_J induces an isomorphism of F-varieties $N_J/N_{J,I_w} \stackrel{\sim}{\to} N_J/N_{J,I_w J}$), $\widetilde{z} = z$ and $\pi = \pi_{I_w}$, we deduce a natural B_{J,w_J}^+ -equivariant (resp. $L_J^+ \cap P_{J \cap I_w J^{-1}(I)}$ -equivariant when $w_J = 1$) morphism

$$H^{n-[F:\mathbb{Q}_p]d_{I_wJ}}(N_{J,I_{w,0}}, \pi_{I_w}) \otimes (\omega^{-1} \circ \delta_{I_wJ})^{w_J} \to H^n(N_{J,0}, \pi_{I_w})$$
(29)

and the Hecke action of z on its kernel and cokernel is locally nilpotent.

Using Proposition 3.3.2, (9) with $\tilde{\sigma} = (H^{n-[F:\mathbb{Q}_p]d_{I_w^J}} \operatorname{Ord}_{L_I \cap P_{I \cap I_w^J(J)}} \sigma)^{I_w^J} \otimes (\omega^{-1} \circ \delta_{I_w^J})$ and the natural B_J -equivariant (resp. $L_J \cap P_{J \cap I_w^{J-1}(I)}$ -equivariant when $w_J = 1$) isomorphism

$$\left(\operatorname{c-ind}_{L_{J}\cap P_{J\cap I_{w}J^{-1}(I)}^{-1}}^{L_{J}\cap P_{J\cap I_{w}J^{-1}(I)}^{-1}} \left(\operatorname{H}^{n-[F:\mathbb{Q}_{p}]d_{I_{w}J}}\operatorname{Ord}_{L_{I}\cap P_{I\cap I_{w}J(J)}}\sigma\right)^{l_{w}J}\right) \otimes (\omega^{-1} \circ \delta_{l_{w}J})^{w_{J}} \\
\overset{\sim}{\longrightarrow} \operatorname{c-ind}_{L_{J}\cap P_{J\cap I_{w}J^{-1}(I)}^{-1}}^{L_{J}\cap P_{I\cap I_{w}J^{-1}(I)}^{-1}} \left(\left(\operatorname{H}^{n-[F:\mathbb{Q}_{p}]d_{I_{w}J}}\operatorname{Ord}_{L_{I}\cap P_{I\cap I_{w}J(J)}}\sigma\right)^{l_{w}J} \otimes (\omega^{-1} \circ \delta_{l_{w}J})\right),$$

the localisation of (28) with respect to $z^{\mathbb{N}}$ and the inverse of the localisation of (29) with respect to $z^{\mathbb{N}}$ yield the desired isomorphism (see Remark 3.1.3).

In particular with $w_J = 1$ and $\tilde{\sigma} := \left(H^{n-[F:\mathbb{Q}_p]d_{I_wJ}} \operatorname{Ord}_{L_I \cap P_{I \cap I_wJ}(J)} \sigma \right)^{I_wJ} \otimes (\omega^{-1} \circ \delta_{I_wJ})$, there is a natural $L_J \cap P_{J \cap I_wJ-1}(I)$ -equivariant injection

$$\mathrm{Gr}^1_{B_J} \left(\mathrm{Ind}_{L_J \cap P_{J \cap I_w J^{-1}(I)}}^{L_J} \tilde{\sigma} \right) \hookrightarrow \mathrm{H}^n \mathrm{Ord}_{P_J} \left(\mathrm{c\text{-}ind}_{P_I^-}^{P_I^{-1} w^J P_J} \sigma \right),$$

hence a natural L_J -equivariant morphism

$$A[L_J] \otimes_{A[L_J \cap P_{J \cap I_w J^{-1}(I)}]} \operatorname{Gr}^1_{B_J} \left(\operatorname{Ind}_{L_J \cap P_{I \cap I_w J^{-1}(I)}}^{L_J} \tilde{\sigma} \right) \to \operatorname{H}^n \operatorname{Ord}_{P_J} \left(\operatorname{c-ind}_{P_I^{-1} w^J}^{P_I^{-1} w^J P_J} \sigma \right).$$

In the proof of [Emerton 2010a, Theorem 4.4.6], it is shown that such a morphism factors uniquely through the natural L_J -equivariant surjection

$$A[L_J] \otimes_{A[L_J \cap P_{J \cap I_w J - 1(I)}]} \operatorname{Gr}^1_{B_J} \left(\operatorname{Ind}_{L_J \cap P_{I \cap I_w J - 1(I)}}^{L_J} \tilde{\sigma} \right) \twoheadrightarrow \operatorname{Ind}_{L_J \cap P_{I \cap I_w J - 1(I)}}^{L_J} \tilde{\sigma}.$$

Thus, the previous injection naturally extends to an L_J -equivariant morphism

$$\operatorname{Ind}_{L_{J}\cap P_{I\cap I_{w}J-1(I)}^{-}}^{L_{J}}\left(\left(\operatorname{H}^{n-[F:\mathbb{Q}_{p}]d_{I_{w}J}}\operatorname{Ord}_{L_{I}\cap P_{I\cap I_{w}J(J)}}\sigma\right)^{l_{w}J}\otimes(\omega^{-1}\circ\delta_{l_{w}J})\right)\to\operatorname{H}^{n}\operatorname{Ord}_{P_{J}}\left(\operatorname{c-ind}_{P_{I}^{-}}^{P_{I}^{-}l_{w}J}P_{J}\sigma\right). \tag{30}$$

Conjecture 3.3.4. The natural morphism (30) is an isomorphism.

We prove Conjecture 3.3.4 in some special cases.

 $\textbf{Proposition 3.3.5.} \quad \text{(i)} \ \, If \, \mathbf{H}^{n-[F:\mathbb{Q}_p]d_{I_wJ}} \, \mathrm{Ord}_{L_I \cap P_{I \cap I_wJ}(J)} \, \sigma = 0, \, then$

$$\operatorname{H}^{n}\operatorname{Ord}_{P_{J}}\left(\operatorname{c-ind}_{P_{I}^{-}}^{P_{I}^{-}l_{w}^{J}P_{J}}\sigma\right)=0.$$

(ii) If ${}^{I}w^{J}(J) \subseteq I$, then (30) is a natural L_{J} -equivariant isomorphism

$$\left(\mathrm{H}^{n-[F:\mathbb{Q}_p]d_{I_w^J}}\mathrm{Ord}_{L_I\cap P_{I\cap I_w^J(J)}}\,\sigma\right)^{I_w^J}\otimes (\omega^{-1}\circ \delta_{I_w^J})\xrightarrow{\sim} \mathrm{H}^n\mathrm{Ord}_{P_J}\big(\mathrm{c\text{-}ind}_{P_I^{-1}w^J}^{P_I^{-1}w^J}P_J\,\sigma\big).$$

(iii) If n = 0 and ${}^{I}w^{J} = 1$, then (30) is a natural L_{J} -equivariant isomorphism

$$\operatorname{Ind}_{L_I \cap P_I^-}^{L_J} \left(\operatorname{Ord}_{L_I \cap P_J} \sigma \right) \xrightarrow{\sim} \operatorname{Ord}_{P_J} \left(\operatorname{c-ind}_{P_I^-}^{P_I^- P_J} \sigma \right).$$

Proof. We first use Theorem 3.3.3.

If $H^{n-[F:\mathbb{Q}_p]d_{I_w^J}}$ $Ord_{L_I\cap P_{I\cap I_w^J(J)}}\sigma=0$, we deduce that $Gr_B^{\bullet}\big(H^nOrd_{P_J}(\operatorname{c-ind}_{P_I^{-1}w^JP_J}^{P_I^{-1}w^JP_J}\sigma)\big)=0$, hence (i). If ${}^Iw^J(J)\subseteq I$, we deduce from [Emerton 2010b, Proposition 3.6.1] that $Gr_B^{\bullet}\big(H^nOrd_{P_J}(\operatorname{c-ind}_{P_I^{-1}w^JP_J}\sigma)\big)$ is concentrated in degree 1; thus (30) is an isomorphism, hence (ii).

We now prove (iii). Since all the functors involved commute with inductive limits, we reduce to the case where σ is admissible. By [Abe et al. 2017b, Corollaries 4.13 and 5.9], there is a natural L_J -equivariant isomorphism

$$\operatorname{Ind}_{L_{I} \cap P_{I}}^{L_{J}} \left(\operatorname{Ord}_{L_{I} \cap P_{J}} \sigma \right) \xrightarrow{\sim} \operatorname{Ord}_{P_{J}} \left(\operatorname{Ind}_{P_{I}^{-}}^{G} \sigma \right). \tag{31}$$

Using (i), we deduce from Proposition 3.3.1 with n = 0 that $Gr_{P_J}^{\bullet}(Ord_{P_J}(Ind_{P_I}^G \sigma))$ is concentrated in degree 1, hence a natural L_J -equivariant isomorphism

$$\operatorname{Ord}_{P_J}\left(\operatorname{c-ind}_{P_I^-}^{P_I^-P_J}\sigma\right) \xrightarrow{\sim} \operatorname{Ord}_{P_J}\left(\operatorname{Ind}_{P_I^-}^G\sigma\right). \tag{32}$$

The composition of (30) with n=0 and ${}^Iw^J=1$, (32) and the inverse of (31) yields an L_J -equivariant endomorphism φ of $\operatorname{Ind}_{L_J\cap P_I^-}^{L_J}(\operatorname{Ord}_{L_I\cap P_J}\sigma)$ which is injective in restriction to $\operatorname{Fil}_{B_J}^1(\operatorname{Ind}_{L_J\cap P_I^-}^{L_J}(\operatorname{Ord}_{L_I\cap P_J}\sigma))$. From [Emerton 2010a, Lemma 4.3.1 and Proposition 4.3.4] and the left-exactness of $\operatorname{Ord}_{L_I\cap P_J}$, we deduce that $\operatorname{Ord}_{L_J\cap P_I}\varphi$ is an injective $L_{I\cap J}$ -equivariant endomorphism of $\operatorname{Ord}_{L_I\cap P_J}\sigma$. Since the latter is admissible by [op. cit., Theorem 3.3.3], it is Artinian (see Section 4.1 below), and thus co-Hopfian so that $\operatorname{Ord}_{L_I\cap P_J}\varphi$ is an isomorphism. We deduce that φ is an isomorphism using [op. cit., Proposition 4.3.4 and Theorem 4.4.6]. We conclude that (30) with n=0 and ${}^Iw^J=1$ is an isomorphism as in the statement. \square

Remark 3.3.6. Let R*Ord_{$L_I \cap P_J$} denote the derived functors of Ord_{$L_I \cap P_J$} on Mod^{l.adm}_{L_I}(A). By universality of derived functors, the isomorphism in (iii) extends uniquely to a morphism of δ -functors

$$\operatorname{Ind}_{L_I \cap P_I^-}^{L_J} \circ \operatorname{R}^{\bullet} \operatorname{Ord}_{L_I \cap P_J} \to \operatorname{H}^{\bullet} \operatorname{Ord}_{P_J} \circ \operatorname{c-ind}_{P_I^-}^{P_I^-}$$
(33)

(the left-hand side is the derived functor of $\operatorname{Ind}_{L_J \cap P_I^-}^{L_J} \circ \operatorname{Ord}_{L_I \cap P_J}$ by exactness of $\operatorname{Ind}_{L_J \cap P_I^-}^{L_J}$, and the right-hand side is a δ -functor by the same arguments as in the proof of Proposition 3.3.1). Now, assume that [Emerton 2010b, Conjecture 3.7.2] is true for $L_I \cap P_J$, i.e., $\operatorname{R}^{\bullet}\operatorname{Ord}_{L_I \cap P_J} \xrightarrow{\sim} \operatorname{H}^{\bullet}\operatorname{Ord}_{L_I \cap P_J}$. Then Conjecture 3.3.4 for ${}^Iw^J = 1$ is equivalent to (33) being an isomorphism. We could prove this if we knew that the isomorphism of Theorem 3.3.3 with ${}^Iw^J = 1$ were B_J -equivariant for all $w_J \in {}^{J \cap I}W_J$.

Finally, we compute the derived ordinary parts of a parabolically induced representation in low degree when there is an inclusion between I and J.

Proposition 3.3.7. (i) If $I \subseteq J$ and $0 < n < [F : \mathbb{Q}_p]$, then $H^n \operatorname{Ord}_{P_J}(\operatorname{Ind}_{P_J^-}^G \sigma) = 0$.

(ii) If $J \subseteq I$ and $n < [F : \mathbb{Q}_p]$, then there is a natural L_J -equivariant isomorphism

$$H^n Ord_{L_I \cap P_J} \sigma \xrightarrow{\sim} H^n Ord_{P_J} (Ind_{P_I^-}^G \sigma).$$

(iii) If $J \subseteq I$ and $\operatorname{Ord}_{L_I \cap P_{I \cap S_{\alpha(J)}}} \sigma = 0$ for all $\alpha \in \Delta^1 \setminus (I \cup J^{\perp})$, then there is a natural short exact sequence of L_J -representations

$$0 \to \mathrm{H}^{[F;\mathbb{Q}_p]}\mathrm{Ord}_{L_I \cap P_J} \, \sigma \to \mathrm{H}^{[F;\mathbb{Q}_p]}\mathrm{Ord}_{P_J}(\mathrm{Ind}_{P_I^-}^G \, \sigma) \to \bigoplus_{\alpha \in J^{\perp,1} \setminus I} (\mathrm{Ord}_{L_I \cap P_J} \, \sigma)^\alpha \otimes (\omega^{-1} \circ \alpha) \to 0.$$

Proof. We use Proposition 3.3.1 and Lemma 2.2.1 with $\ell: {}^{I}W^{J} \to \mathbb{N}$ to obtain a filtration

$$\operatorname{Fil}_{P_J}^{\ell, \bullet} \left(\operatorname{H}^n \operatorname{Ord}_{P_J} (\operatorname{Ind}_{P_I^-}^G \sigma) \right)$$

indexed by \mathbb{N} such that for all $i \in \mathbb{N}$, there is a natural L_J -equivariant isomorphism

$$\operatorname{Gr}_{P_{J}}^{\ell,i}\left(\operatorname{H}^{n}\operatorname{Ord}_{P_{J}}(\operatorname{Ind}_{P_{I}^{-}}^{G}\sigma)\right) \cong \bigoplus_{\ell(Iw^{J})=i} \operatorname{H}^{n}\operatorname{Ord}_{P_{J}}\left(\operatorname{c-ind}_{P_{I}^{-}}^{P_{I}^{-}Iw^{J}P_{J}}\sigma\right). \tag{34}$$

Assume $n < [F: \mathbb{Q}_p]$. If ${}^Iw^J \neq 1$ (i.e., $d_{I_w^J} > 0$), then $H^n\mathrm{Ord}_{P_J} \big(\mathrm{c\text{-}ind}_{P_l^-}^{P_l^-I_w^J} P_J \sigma \big) = 0$ by Proposition 3.3.5(i) since $n - [F: \mathbb{Q}_p] d_{I_w^J} < 0$; thus we deduce from (34) that $\mathrm{Gr}_{P_J}^{\ell, \bullet} \big(H^n\mathrm{Ord}_{P_J} (\mathrm{Ind}_{P_l^-}^G \sigma) \big)$ is concentrated in degree 0, so that assuming Conjecture 3.3.4 for ${}^Iw^J = 1$, we obtain a natural L_J -equivariant isomorphism

$$\operatorname{Ind}_{L_I \cap P_I^-}^{L_J} \left(\operatorname{H}^n \operatorname{Ord}_{L_I \cap P_J} \sigma \right) \xrightarrow{\sim} \operatorname{H}^n \operatorname{Ord}_{P_J} (\operatorname{Ind}_{P_I^-}^G \sigma). \tag{35}$$

Now, Conjecture 3.3.4 is true for ${}^Iw^J=1$ in the following cases: n>0 and $I\subseteq J$ by Proposition 3.3.5(i) since $H^n\mathrm{Ord}_{L_I\cap P_J}=H^n\mathrm{Ord}_{L_I}=0$ [Emerton 2010b, Proposition 3.6.1], in which case the source of (35) is zero, hence (i); $J\subseteq I$ by Proposition 3.3.5(ii), in which case the source of (35) is $H^n\mathrm{Ord}_{L_I\cap P_J}\sigma$, hence (ii).

Likewise, if ${}^Iw^J \neq 1$ and ${}^Iw^J \neq s_\alpha$ for all $\alpha \in \Delta^1 \setminus (I \cup J)$ (i.e., $d_{I_wJ} > 1$), then

$$H^{[F:\mathbb{Q}_p]}\mathrm{Ord}_{P_J}(\mathrm{c\text{-}ind}_{P_I^{-}}^{P_I^{-}lw^JP_J}\sigma)=0$$

by Proposition 3.3.5(i) since $[F:\mathbb{Q}_p]-[F:\mathbb{Q}_p]d_{lw^J}<0$; thus we deduce from (34) that

$$\operatorname{Gr}_{P_J}^{\ell, \bullet} \left(\operatorname{H}^{[F:\mathbb{Q}_p]} \operatorname{Ord}_{P_J} (\operatorname{Ind}_{P_I^-}^G \sigma) \right)$$

is concentrated in degrees 0 and 1, so that assuming Conjecture 3.3.4 for $n = [F : \mathbb{Q}_p]$ and ${}^Iw^J = 1$ or ${}^Iw^J = s_\alpha$ for all $\alpha \in \Delta^1 \setminus (I \cup J)$, we obtain a short exact sequence of L_J -representations

$$0 \to \operatorname{Ind}_{L_{J} \cap P_{I}^{-}}^{L_{J}} \left(\operatorname{H}^{[F:\mathbb{Q}_{p}]} \operatorname{Ord}_{L_{I} \cap P_{J}} \sigma \right) \to \operatorname{H}^{[F:\mathbb{Q}_{p}]} \operatorname{Ord}_{P_{J}} \left(\operatorname{Ind}_{P_{I}^{-}}^{G} \sigma \right)$$

$$\to \bigoplus_{\alpha \in \Lambda^{1} \setminus \{I \cup I\}} \operatorname{Ind}_{L_{J} \cap P_{J \cap s_{\alpha}(I)}}^{L_{J}} \left(\left(\operatorname{Ord}_{L_{I} \cap P_{I \cap s_{\alpha}(J)}} \sigma \right)^{\alpha} \otimes (\omega^{-1} \circ \alpha) \right) \to 0. \quad (36)$$

Assume $J \subseteq I$ and $\operatorname{Ord}_{L_I \cap P_{I \cap s_\alpha(J)}} \sigma = 0$ for all $\alpha \in \Delta^1 \setminus (I \cup J^\perp)$. Then Conjecture 3.3.4 is indeed true for $n = [F : \mathbb{Q}_p]$ in the following cases: ${}^I w^J = 1$ by Proposition 3.3.5(ii), and the first nontrivial term of (36) is $\operatorname{H}^{[F : \mathbb{Q}_p]} \operatorname{Ord}_{L_I \cap P_J} \sigma$; ${}^I w^J = s_\alpha$ with $\alpha \in \Delta^1 \setminus (I \cup J^\perp)$ by Proposition 3.3.5(i) and the hypothesis on σ , and the corresponding summand of the last nontrivial term of (36) is zero; ${}^I w^J = s_\alpha$ with $\alpha \in J^{\perp,1} \setminus I$ by Proposition 3.3.5(ii) since $s_\alpha(J) = J \subseteq I$, and the corresponding summand of the last nontrivial term of (36) is $(\operatorname{Ord}_{L_I \cap P_J} \sigma)^\alpha \otimes (\omega^{-1} \circ \alpha)$. Hence (iii).

We reformulate Proposition 3.3.7 in the case I = J, using the fact that in this case $H^n \operatorname{Ord}_{L_I \cap P_J} = 0$ if n > 0 [Emerton 2010b, Proposition 3.6.1]. Note that if P = LN is a standard parabolic subgroup, then for all $\alpha \in \Delta \setminus \Delta_L$ the standard parabolic subgroup of L corresponding to $\Delta_L \cap s_\alpha(\Delta_L)$ is $L \cap s_\alpha P s_\alpha^{-1}$ and it is proper if and only if $\alpha \notin \Delta_L^{\perp}$.

Corollary 3.3.8. Let P = LN be a standard parabolic subgroup and σ be a locally admissible L-representation.

- (i) For all $n \in \mathbb{N}$ such that $0 < n < [F : \mathbb{Q}_p]$, we have $H^n \operatorname{Ord}_P(\operatorname{Ind}_{P^-}^G \sigma) = 0$.
- (ii) If $\operatorname{Ord}_{L\cap s_{\alpha}Ps_{\alpha}^{-1}}\sigma=0$ for all $\alpha\in\Delta^1\setminus(\Delta_L\cup\Delta_L^\perp)$, then there is a natural L-equivariant isomorphism

$$\mathrm{H}^{[F:\mathbb{Q}_p]}\mathrm{Ord}_P(\mathrm{Ind}_{P^-}^G\sigma)\cong\bigoplus_{\alpha\in\Delta_L^{\perp,1}}\sigma^\alpha\otimes(\omega^{-1}\circ\alpha).$$

4. Derived Jacquet functors

The aim of this section is to study the derived functors of the Jacquet functor. In Section 4.1, we review some results on pro-categories. In Section 4.2, we relate the left derived functors of the Jacquet functor in a pro-category with the derived ordinary parts functors and we construct a new exact sequence to compute extensions by a parabolically induced representation. In Section 4.3, we adapt the results of Section 3.3 in order to partially compute the derived Jacquet functors on a parabolically induced representation.

4.1. *Pro-categories.* Let H be a p-adic Lie group. Let $\mathfrak C$ be the category whose objects are the A[H]-modules such that for some (equivalently any) compact open subgroup $H_0 \subseteq H$, the $A[H_0]$ -action extends to a structure of $A[H_0]$ -module of finite type, and whose morphisms are the A[H]-linear maps. Since the completed group rings are Noetherian [Emerton 2010a, Theorem 2.1.2], the category $\mathfrak C$ is A-abelian and Noetherian, i.e., it is essentially small⁷ and its objects are Noetherian. Let $\mathfrak C^{\wedge}$ be the category of contravariant functors $\mathfrak C \to S$ et and Ind- $\mathfrak C$ be the full subcategory of $\mathfrak C^{\wedge}$ whose objects are the functors isomorphic to a small inductive limit in $\mathfrak C^{\wedge}$ of objects of $\mathfrak C$ (using the Yoneda embedding $\mathfrak C \to \mathfrak C^{\wedge}$). By [Kashiwara and Schapira 2006, Theorem 8.6.5], the category Ind- $\mathfrak C$ is a Grothendieck category (in particular it has enough injectives; see [Kashiwara and Schapira 2006, Theorem 9.6.2]) and the natural A-linear functor $\mathfrak C \to Ind-\mathfrak C$ is fully faithful and exact.

Now, Pontryagin duality induces an equivalence of categories [Emerton 2010a, (2.2.12)]

$$Mod_H^{adm}(A) \cong \mathfrak{C}^{op}$$
.

Thus, the category $\operatorname{Mod}_{H}^{\operatorname{adm}}(A)$ is Artinian, the pro-category

$$\operatorname{Pro-Mod}_{H}^{\operatorname{adm}}(A) := (\operatorname{Ind-\mathfrak{C}})^{\operatorname{op}}$$

has enough projectives, and the natural A-linear functor

$$\operatorname{Mod}_{H}^{\operatorname{adm}}(A) \to \operatorname{Pro-Mod}_{H}^{\operatorname{adm}}(A)$$
 (37)

is fully faithful and exact. We let $\operatorname{Ext}_{Pro-H}^{\bullet}$ denote the bifunctors of Yoneda extensions in the categories $\operatorname{Mod}_{H}^{\operatorname{adm}}(A)$ and $\operatorname{Pro-Mod}_{H}^{\operatorname{adm}}(A)$ respectively. By [Oort 1964, Theorem 3.5], (37) induces A-linear isomorphisms

$$\operatorname{Ext}_{H}^{\bullet}(\pi',\pi) \xrightarrow{\sim} \operatorname{Ext}_{\operatorname{Pro-}H}^{\bullet}(\pi',\pi)$$
 (38)

for all objects π , π' of $Mod_H^{adm}(A)$.

are exact.

4.2. A second exact sequence. Let $P \subseteq G$ be a parabolic subgroup and $L \subseteq P$ be a Levi factor. We let $P^- \subseteq G$ denote the parabolic subgroup opposed to P with respect to L. There is a natural exact sequence of A-modules [Emerton 2010b, (3.7.6)]

$$0 \to \operatorname{Ext}_{L}^{1}(\sigma, \operatorname{Ord}_{P} \pi) \to \operatorname{Ext}_{G}^{1}(\operatorname{Ind}_{P^{-}}^{G} \sigma, \pi) \to \operatorname{Hom}_{L}(\sigma, \operatorname{H}^{1}\operatorname{Ord}_{P} \pi)$$
(39)

⁷A category is essentially small if it is equivalent to a small category, i.e., if the isomorphism classes of its objects form a set. ⁸A Grothendieck category is an abelian category that admits a generator and small direct sums, and in which inductive limits

for all objects σ and π of $\mathrm{Mod}_L^{\mathrm{adm}}(A)$ and $\mathrm{Mod}_G^{\mathrm{adm}}(A)$ respectively. We construct a second exact sequence, in which parabolic induction is on the right.

By [Emerton 2010a, Proposition 4.1.5 and Proposition 4.1.7], parabolic induction induces an A-linear exact functor

$$\operatorname{Ind}_{P}^{G}: \operatorname{Mod}_{I}^{\operatorname{adm}}(A) \to \operatorname{Mod}_{G}^{\operatorname{adm}}(A).$$

By [Emerton 2010b, Corollary 3.6.7], taking *N*-coinvariants induces an *A*-linear right-exact functor (the so-called Jacquet functor)

$$(-)_N : \operatorname{Mod}_G^{\operatorname{adm}}(A) \to \operatorname{Mod}_L^{\operatorname{adm}}(A).$$

By Frobenius reciprocity and the universal property of coinvariants, there is a natural A-linear isomorphism

$$\operatorname{Hom}_{G}(\pi, \operatorname{Ind}_{P}^{G} \sigma) \cong \operatorname{Hom}_{L}(\pi_{N}, \sigma).$$
 (40)

for all objects π and σ of $\mathrm{Mod}_G^{\mathrm{adm}}(A)$ and $\mathrm{Mod}_L^{\mathrm{adm}}(A)$ respectively.

We deduce from [Kashiwara and Schapira 2006, Proposition 6.1.9] that these functors and the adjunction relation extend to the corresponding pro-categories. By [op. cit., Corollary 8.6.8], Ind_P^G is still exact so that $(-)_N$ still preserves projectives. Thus, denoting by $\operatorname{L}_{\bullet}(N,-)$ the left derived functors of $(-)_N$ in $\operatorname{Pro-Mod}_G^{\operatorname{adm}}(A)$, there is a Grothendieck spectral sequence of A-modules

$$\operatorname{Ext}^{i}_{\operatorname{Pro-}L}(\operatorname{L}_{j}(N,\pi),\sigma) \Rightarrow \operatorname{Ext}^{i+j}_{\operatorname{Pro-}G}(\pi,\operatorname{Ind}_{P}^{G}\sigma)$$

whose low degree terms form a natural exact sequence of A-modules

$$0 \to \operatorname{Ext}^1_{\operatorname{Pro-}L}(\pi_N, \sigma) \to \operatorname{Ext}^1_{\operatorname{Pro-}G}(\pi, \operatorname{Ind}_P^G \sigma) \to \operatorname{Hom}_{\operatorname{Pro-}L}(\operatorname{L}_1(N, \pi), \sigma)$$

$$\rightarrow \operatorname{Ext}^2_{\operatorname{Pro-}L}(\pi_N, \sigma) \rightarrow \operatorname{Ext}^2_{\operatorname{Pro-}G}(\pi, \operatorname{Ind}_P^G \sigma)$$
 (41)

for all objects π and σ of $\operatorname{Pro-Mod}_G^{\operatorname{adm}}(A)$ and $\operatorname{Pro-Mod}_L^{\operatorname{adm}}(A)$ respectively.

We let d denote the integer $\dim_F N$ and $\delta \in X^*(L)$ denote the algebraic character of the adjoint representation of L on $\det_F(\text{Lie }N)$. We define A-linear functors by setting

$$H_{\bullet}(N,-) := H^{[F:\mathbb{Q}_p]d-\bullet} Ord_P \otimes (\omega \circ \delta)$$
.

We deduce from [Emerton 2010b, Corollary 3.4.8 and Proposition 3.6.1] that we obtain a homological δ -functor

$$H_{\bullet}(N, -) : \operatorname{Mod}_{G}^{\operatorname{adm}}(A) \to \operatorname{Mod}_{L}^{\operatorname{adm}}(A)$$

and proceeding as in the proof of [Kashiwara and Schapira 2006, Corollary 8.6.8], we see that it extends to a homological δ -functor between the corresponding pro-categories.

By [Emerton 2010b, Proposition 3.6.2], there is an isomorphism of functors (hence the notation)

$$H_0(N,-)\cong (-)_N$$

which, by universality of derived functors, extends uniquely to a morphism of δ -functors

$$H_{\bullet}(N,-) \to L_{\bullet}(N,-)$$
 (42)

which is bijective in degree 0, and thus surjective in degree 1 (by a dimension-shifting argument). Using (38), we deduce from (41) a natural exact sequence of A-modules

$$0 \to \operatorname{Ext}_{I}^{1}(\pi_{N}, \sigma) \to \operatorname{Ext}_{G}^{1}(\pi, \operatorname{Ind}_{P}^{G} \sigma) \to \operatorname{Hom}_{L}(\operatorname{H}_{1}(N, \pi), \sigma)$$

$$\tag{43}$$

for all objects π and σ of $\mathrm{Mod}_G^{\mathrm{adm}}(A)$ and $\mathrm{Mod}_L^{\mathrm{adm}}(A)$ respectively.

Remark 4.2.1. (i) Nothing is known (to the author at least) regarding the nature of the morphism (42) in degree greater than 1.

(ii) Let H be a p-adic Lie group. Taking inductive limits induces an A-linear exact functor

$$\varinjlim : \operatorname{Ind-Mod}_H^{\operatorname{adm}}(A) \to \operatorname{Mod}_H^{\operatorname{l.adm}}(A)$$

which is essentially surjective, but not faithful nor full in general. Thus the situation here (i.e., deriving in Pro-Mod_H^{adm}(A)) is not exactly dual to that of [Emerton 2010b, §3.7] (i.e., deriving in Mod_H^{l,adm}(A)).

4.3. Adaptation of the computations. Let $I, J \subseteq \Delta$, σ be an L_I -representation and $n \in \mathbb{N}$. We let ${}^Iw_0 = w_{I,0}w_0$ (resp. ${}^{J\cap^Iw^{J-1}(I)}w_{J,0} = w_{J\cap^Iw^{J-1}(I),0}w_{J,0}$) denote the image of w_0 (resp. $w_{J,0}$) in IW (resp. ${}^{J\cap^Iw^{J-1}(I)}W_J$) and we define an auxiliary subset of Δ by setting $I' := {}^Iw_0^{-1}(I)$. We have $L_I = {}^Iw_0L_{I'}{}^Iw_0^{-1}$ and $P_I = {}^Iw_0P_{I'}{}^Iw_0^{-1}$, hence a natural G-equivariant isomorphism

$$\operatorname{Ind}_{P_I}^G \sigma \cong \operatorname{Ind}_{P_{I'}^-}^G \sigma^{I_{w_0}} \tag{44}$$

defined by $f \mapsto (g \mapsto f({}^{I}w_0g)).$

Lemma 4.3.1. The map ${}^IW^J \to {}^{I'}W^J$ defined by ${}^Iw^J \mapsto {}^Iw_0^{-1}{}^Iw^{J} {}^{I}\cap {}^Iw^{J-1}(I)$ $w_{J,0}$ is an order-reversing bijection.

Proof. First, note that $W_I = w_0 W_{I'} w_0$, so that left translation by w_0 induces a bijection

$$W_I \setminus W / W_J \xrightarrow{\sim} W_{I'} \setminus W / W_J$$
.

In particular, card ${}^{I}W^{J} = \operatorname{card}^{I'}W^{J}$. Thus, it is enough to prove that the order-reversing composite

$${}^IW^J \hookrightarrow W \xrightarrow{\sim} W$$

where the first arrow is defined by ${}^Iw^J \mapsto w_{I,0}{}^Iw^{JJ\cap {}^Iw^{J-1}(I)}w_{J,0}$ (it is injective since ${}^IW^J$ is a system of representatives of the double cosets $W_I \setminus W/W_J$, and order-preserving since the projection $W \to {}^IW^J$ is order-preserving) and the second arrow is the left multiplication by w_0 (it is an order-reversing bijection; see [Björner and Brenti 2005, Proposition 2.3.4(i)]), takes values in ${}^{I'}W^J$.

Now, let ${}^Iw^J \in {}^IW^J$. For all ${}^Iw \in {}^IW$ and $w_{I'} \in W_{I'}$ (using [op. cit., Proposition 2.3.2(ii)]),

$$\ell(w_{I'}^{I}w_{0}^{-1}^{I}w) = \ell(w_{0}(w_{0}w_{I'}w_{0})w_{I,0}^{I}w)$$

$$= \ell(w_{0}) - ((\ell(w_{I,0}) - \ell(w_{0}w_{I'}w_{0})) + \ell({}^{I}w))$$

$$= \ell(w_{I'}) + \ell(w_{0}w_{I,0}^{I}w).$$

Since ${}^Iw^{JJ\cap {}^Iw^{J-1}(I)}w_{J,0}\in {}^IW$, we deduce that ${}^Iw_0^{-1}Iw^{JJ\cap {}^Iw^{J-1}(I)}w_{J,0}\in {}^{I'}W$. Likewise, for all $w^J\in W^J$, we have $w_0w^Jw_{J,0}\in W^J$. Since

$$\begin{split} {}^{I}w_{0}^{-1}{}^{I}w^{JJ\cap {}^{I}w^{J-1}(I)}w_{J,0} &= w_{0}w_{I,0}{}^{I}w^{J}w_{J\cap {}^{I}w^{J-1}(I),0}w_{J,0} \\ &= w_{0}w_{I,0}w_{I\cap {}^{I}w^{J}(J),0}{}^{I}w^{J}w_{J,0} \\ &= w_{0}w_{I,0}^{I\cap {}^{I}w^{J}(J)I}w^{J}w_{J,0} \end{split}$$

and $w_{I,0}^{I\cap ^Iw^J(J)I}w^J\in W^J$, we deduce that $^Iw_0^{-1I}w^{JJ\cap ^Iw^{J-1}(I)}w_{J,0}\in W^J$. We conclude that

$${}^{I}w_{0}^{-1}{}^{I}w^{JJ\cap {}^{I}w^{J-1}(I)}w_{J,0}\in {}^{I'}W^{J}.$$

We see from Lemma 4.3.1 that the left translate by Iw_0 of the decomposition $G = \bigsqcup_{I'w^J \in {}^{I'}W^J} P_{I'}^{-I'}w^J P_J$ is the decomposition $G = \bigsqcup_{I_w^J \in {}^{I'}W^J} P_I^{-I}w^J P_J$ with the opposite closure relations. Proceeding as in Section 2.2, we can construct a natural filtration $\operatorname{Fil}_{P_J}^G(\operatorname{Ind}_{P_I}^G\sigma)$ by P_J -subrepresentations indexed by ${}^IW^J$ with the opposite Bruhat order, and there is a natural P_J -equivariant isomorphism

$$\operatorname{Gr}_{P_I}^{I_W^J}(\operatorname{Ind}_{P_I}^G \sigma) \cong \operatorname{c-ind}_{P_I}^{P_I^I w^J P_J} \sigma$$

for all ${}^{I}w^{J} \in {}^{I}W^{J}$. Furthermore, (44) identifies this filtration with $\operatorname{Fil}_{P_{J}}^{\bullet}(\operatorname{Ind}_{P_{J'}^{-}}^{G}\sigma^{lw_{0}})$, using Lemma 4.3.1 to identify the indexing posets, and induces a natural P_{J} -equivariant isomorphism

$$\operatorname{c-ind}_{P_I}^{P_I^I w^J P_J} \sigma \cong \operatorname{c-ind}_{P_{I'}^{-I'}}^{P_{I'}^{-I'} w^J P_J} \sigma^{I_{w_0}}$$

$$\tag{45}$$

for all ${}^Iw^J \in {}^IW^J$ with ${}^Iw^J = {}^Iw_0^{-1}Iw^{JJ} \cap {}^Iw^{J-1}(I)w_{J,0}$. Therefore, by Proposition 3.3.1, $\mathrm{Fil}_{P_J}^{\bullet}(\mathrm{Ind}_{P_I}^G\sigma)$ induces a filtration $\mathrm{Fil}_{P_J}^{\bullet}(\mathrm{H}_n(N_J,\mathrm{Ind}_{P_I}^G\sigma))$ by L_J -subrepresentations indexed by ${}^IW^J$ with the opposite Bruhat order and there is a natural L_J -equivariant isomorphism

$$\operatorname{Gr}_{P_I}^{lw^J}(\operatorname{H}_n(N_J,\operatorname{Ind}_{P_I}^G\sigma)) \cong \operatorname{H}_n(N_J,\operatorname{c-ind}_{P_I}^{P_Ilw^JP_J}\sigma)$$

for all ${}^{I}w^{J} \in {}^{I}W^{J}$.

Let ${}^I \! w^J \in {}^I \! W^J$ and set ${}^I \! w^J := {}^I \! w_0^{-1} \! w^{J \, J \, \cap^I \! w^{J-1}(I)} w_{J,0}$. We let $\tilde{\sigma}$ be an $L_{J \, \cap^I \! w^{J-1}(I)}$ -representation. Note that $J \, \cap^I \! w^{J-1}(I) = {}^{J \, \cap^I \! w^{J-1}(I)} w_{J,0} (J \, \cap^I \! w^{J-1}(I'))$. We have

$$L_{J\cap {}^{I}w^{J-1}(I)} = {}^{J\cap {}^{I}w^{J-1}(I)}w_{J,0}L_{J\cap {}^{I}w^{J-1}(I')}{}^{J\cap {}^{I}w^{J-1}(I)}w_{J,0}^{-1}$$

and

$$L_{J} \cap P_{J \cap {}^{I}w^{J-1}(I)} = {}^{J \cap {}^{I}w^{J-1}(I)}w_{J,0}L_{J} \cap P_{J \cap {}^{I}w^{J-1}(I')}^{-} {}^{J \cap {}^{I}w^{J-1}(I)}w_{J,0}^{-1},$$

hence a natural L_J -equivariant isomorphism

$$\operatorname{Ind}_{L_{J} \cap P_{J \cap I_{w}J - 1(I)}}^{L_{J}} \tilde{\sigma} \cong \operatorname{Ind}_{L_{J} \cap P_{J \cap I'_{w}J - 1(I')}}^{L_{J}} \tilde{\sigma}^{J \cap I'_{w}J - 1(I)} w_{J,0}$$

$$\tag{46}$$

defined by $f \mapsto (l \mapsto f(^{J \cap l_w^{J-1}(I)} w_{J,0} l))$. Proceeding as in the proof of Lemma 4.3.1, we obtain the following result.

Lemma 4.3.2. The map $^{J\cap^I w^{J-1}(I)}W_J \to ^{J\cap^{I'} w^{J-1}(I')}W_J$ defined by $w_J \mapsto ^{J\cap^I w^{J-1}(I)}w_{J,0}^{-1}w_J$ is an order-reversing bijection.

We deduce from Lemma 4.3.2 that the left translate by ${}^{I}w_0$ (resp. ${}^{J\cap {}^{I}w^{J-1}(I)}w_{J,0}$) of the decomposition

$$P_{I'}^{-I'}w^{J}P_{J} = \bigsqcup_{w'_{I} \in J \cap I'_{w^{J-1}(I')}W_{J}} P_{I'}^{-I'}w^{J}w'_{J}B \quad \text{(resp. } L_{J} = \bigsqcup_{w'_{J} \in J \cap I'_{w^{J-1}(I')}W_{J}} L_{J} \cap P_{J \cap I'_{w^{J-1}(I')}}^{-}w'_{J}B_{J})$$

is the decomposition

$$P_I{}^Iw^JP_J = \bigsqcup_{w_J \in {}^{J \cap I_wJ - 1}(I)} P_I{}^Iw^Jw_JB \quad \text{(resp. } L_J = \bigsqcup_{w_J \in {}^{J \cap I_wJ - 1}(I)} L_J \cap P_{J \cap I_wJ - 1}(I)w_JB_J)$$

with the opposite closure relations. Proceeding as in Section 2.2, we can construct a natural filtration $\operatorname{Fil}_{B}^{\bullet}(\operatorname{c-ind}_{P_{I}}^{P_{I}^{I}w^{J}P_{J}}\sigma)$ (resp. $\operatorname{Fil}_{B_{J}}^{\bullet}(\operatorname{Ind}_{L_{J}\cap P_{J\cap I_{w}J-1}(I)}^{L_{J}}\tilde{\sigma})$) by B- (resp. B_{J} -)subrepresentations indexed by ${}^{J\cap I_{w}J-1}(I)W_{J}$ with the opposite Bruhat order, and there is a natural B- (resp. B_{J} -)equivariant isomorphism

$$\operatorname{Gr}_{B}^{w_{J}}\left(\operatorname{c-ind}_{P_{I}}^{P_{I}^{I}w^{J}P_{J}}\sigma\right) \cong \operatorname{c-ind}_{P_{I}}^{P_{I}^{I}w^{J}w_{J}B}\sigma \quad \text{(resp. } \operatorname{Gr}_{B_{J}}^{w_{J}}\left(\operatorname{Ind}_{L_{J}\cap P_{J\cap I_{w}J-1}(I)}^{L_{J}}\tilde{\sigma}\right) \cong \operatorname{c-ind}_{L_{J}\cap P_{J\cap I_{w}J-1}(I)}^{L_{J}\cap P_{J\cap I_{w}J-1}(I)}\tilde{\sigma})$$

for all $w_J \in {}^{J \cap {}^I w^{J-1}(I)} W_J$. Further, (45) (resp. (46)) identifies this filtration with

$$\operatorname{Fil}_{B}^{\bullet}\left(\operatorname{c-ind}_{P_{I'}^{-l'}w^{J}P_{J}}^{P_{I'}^{-l'}w^{J}P_{J}}\sigma^{lw_{0}}\right) \quad \left(\operatorname{resp.} \ \operatorname{Fil}_{B_{J}}^{\bullet}\left(\operatorname{Ind}_{L_{J}\cap P_{I\cap I'_{W^{J}-1}(I')}}^{L_{J}}\tilde{\sigma}^{\int_{I\cap I_{W^{J}-1}(I')}^{I'}w_{J,0}}\right)\right),$$

using Lemma 4.3.2 to identify the indexing posets, and induces a natural B- (resp. B_J -)equivariant isomorphism

$$\operatorname{c-ind}_{P_{I}}^{P_{I}^{I}w^{J}w_{J}B}\sigma\cong\operatorname{c-ind}_{P_{I'}^{-1}}^{P_{I'}^{-1}w^{J}w'_{J}B}\sigma^{I_{w_{0}}}$$

$$(\operatorname{resp. }\operatorname{c-ind}_{L_{J}\cap P_{J\cap I_{w}J-1(I)}}^{L_{J}\cap P_{J\cap I_{w}J-1(I)}}\tilde{\sigma}\cong\operatorname{c-ind}_{L_{J}\cap P_{J\cap I'_{w}J-1(I')}}^{L_{J}\cap P_{J\cap I'_{w}J-1(I')}}\tilde{\sigma}^{J\cap I_{w}J-1(I)}w_{J,0})$$

for all $w_J \in {}^{J \cap {}^I w^{J-1}(I)} W_J$ with $w_J' = {}^{J \cap {}^I w^{J-1}(I)} w_{J,0}^{-1} w_J$. Therefore, by Proposition 3.3.2, $\operatorname{Fil}_B^{\bullet}(\operatorname{c-ind}_{P_I}^{P_I I_w J_P_J} \sigma)$ induces a filtration $\operatorname{Fil}_B^{\bullet}(\operatorname{H}_n(N_J,\operatorname{c-ind}_{P_I}^{P_I I_w J_P_J} \sigma))$ by B_J -subrepresentations indexed by ${}^{J \cap {}^I w^{J-1}(I)} W_J$ with the opposite Bruhat order and there is a natural B_J -equivariant isomorphism

$$\operatorname{Gr}_{B}^{w_{J}}(\operatorname{H}_{n}(N_{J},\operatorname{c-ind}_{P_{I}}^{P_{I}^{I}w^{J}P_{J}}\sigma))\cong \operatorname{H}_{n}(N_{J},\operatorname{c-ind}_{P_{I}}^{P_{I}^{I}w^{J}w_{J}B}\sigma)$$

for all $w_J \in {}^{J \cap {}^I w^{J-1}(I)} W_J$.

Theorem 4.3.3. Let σ be a locally admissible L_I -representation, ${}^Iw^J \in {}^IW^J$ and $n \in \mathbb{N}$. For all $w_J \in {}^{J \cap {}^Iw^{J-1}(I)}W_J$, there is a natural $B_{J,{}^{J \cap {}^Iw^{J-1}(I)}w_{J,0}^{-1}w_J}$ -equivariant isomorphism

$$\operatorname{Gr}_{B}^{w_{J}}(\operatorname{H}_{n}(N_{J},\operatorname{c-ind}_{P_{I}}^{P_{I}^{I}w^{J}P_{J}}\sigma))$$

$$\cong \operatorname{Gr}_{B_J}^{w_J} \left(\operatorname{Ind}_{L_J \cap P_{I \cap I_w J - 1(I)}}^{L_J} \left(\operatorname{H}_{n - [F:\mathbb{Q}_p] d_{I_w J}} \left(L_I \cap N_{I \cap I_w J(J)}, \sigma \right)^{I_w J} \otimes (\omega \circ \delta_{I_w J}) \right) \right)$$

which is even $L_J \cap P_{J \cap J \cap l_w J^{-1}(I)} w_{J,0}^{-1} equivariant$ when $w_J = {}^{J \cap l_w J^{-1}(I)} w_{J,0}$.

Proof. We set ${}^I\!w^J:={}^I\!w^{-1}_0{}^I\!w^{JJ\cap{}^I\!w^{J-1}(I)}w_{J,0}$ and we define an $L_{J\cap{}^I\!w^{J-1}(I)}$ -representation by setting

$$\tilde{\sigma} := \left(\left(\mathbf{H}^{[F:\mathbb{Q}_p](d_J - d_{I'_w J}) - n} \mathrm{Ord}_{L_{I'} \cap P_{I' \cap I'_w J_{(J)}}} \sigma^{I_{w_0}} \right)^{I'_w J} \otimes \left(\omega \circ (\delta_J - \delta_{I'_w J}) \right)^{J \cap I_w J - 1_{(I)}} w_{J,0}^{-1},$$

where d_J denotes the integer $\dim_F N_J$ and $\delta_J \in X^*(L_J)$ denotes the algebraic character of the adjoint representation of L_J on $\det_F(\operatorname{Lie} N_J)$. We prove that there is a natural $L_{J \cap I_W J^{-1}(I)}$ -equivariant isomorphism

$$\tilde{\sigma} \cong \mathcal{H}_{n-[F:\mathbb{Q}_p]d_{I_wJ}} \left(L_I \cap N_{I \cap {}^I w^J(J)}, \sigma \right)^{I_wJ} \otimes (\omega \circ \delta_{I_wJ}). \tag{47}$$

We have $L_{I\cap I_w^J(J)} = {}^Iw_0L_{I'\cap I'_w^J(J)}{}^Iw_0^{-1}$ and $L_I\cap P_{I\cap I_w^J(J)} = {}^Iw_0L_{I'}\cap P_{I'\cap I'_w^J(J)}{}^Iw_0^{-1}$, hence natural $L_{I'\cap I'_w^J(J)}$ -equivariant isomorphisms

$$\operatorname{H}^{\bullet}\operatorname{Ord}_{L_{I'}\cap P_{I'\cap I'w^{J}(J)}}\sigma^{{}^{I}w_{0}}\cong \left(\operatorname{H}^{\bullet}\operatorname{Ord}_{L_{I}\cap P_{I\cap Iw^{J}(J)}}\sigma\right)^{{}^{I}w_{0}}.$$

Using Lemma 2.1.1(iii), we have (with notations analogous to d_J and δ_J)

$$H^{\bullet}\mathrm{Ord}_{L_I \cap P_{I \cap I_w J_{(J)}}} = H_{[F:\mathbb{Q}_p](d_{I \cap I_w J_{(J)}} - d_I) - \bullet}(L_I \cap N_{I \cap I_w J_{(J)}}, -) \otimes (\omega^{-1} \circ (\delta_{I \cap I_w J_{(J)}} - \delta_I)).$$

Thus in order to prove (47), it remains to check that

$$d_J = (d_{I \cap {}^I w^J(I)} - d_I) + d_{Iw^J} + d_{I'w^J}, \quad \delta_J = {}^I w^{J-1} (\delta_{I \cap {}^I w^J(I)} - \delta_I) + \delta_{Iw^J} + {}^{J \cap {}^I w^{J-1}(I)} w_{J,0} (\delta_{I'w^J}).$$

We do these computations on the corresponding Lie algebras: d_J and δ_J correspond to $\Phi^+ \setminus \Phi_J^+$, $(d_{I \cap ^I w^J(J)} - d_I)$ and $^I w^{J-1} (\delta_{I \cap ^I w^J(J)} - \delta_I)$ correspond to $(\Phi^+ \setminus \Phi_J^+) \cap ^I w^{J-1} (\Phi_I^+)$, $d_{I_w^J}$ and $\delta_{I_w^J}$ correspond to $(\Phi^+ \setminus \Phi_J^+) \cap ^I w^{J-1} (-\Phi^+)$, and $d_{I_w^J}$ and $^{J \cap ^I w^{J-1}(I)} w_{J,0} (\delta_{I_w^J})$ correspond to $(\Phi^+ \setminus \Phi_J^+) \cap ^I w^{J-1} (\Phi^+ \setminus \Phi_I^+)$ (noting that $^I w_0 (-\Phi^+) = (-\Phi_I^+) \sqcup (\Phi^+ \setminus \Phi_I^+)$ and $(\Phi^+ \setminus \Phi_J^+) \cap ^I w^{J-1} (-\Phi_I^+) = \varnothing$). Thus, the two equalities above follow from the partition

$$\Phi^{+} \setminus \Phi_{I}^{+} = \left((\Phi^{+} \setminus \Phi_{I}^{+}) \cap {}^{I} w^{J-1} (\Phi_{I}^{+}) \right) \sqcup \left((\Phi^{+} \setminus \Phi_{I}^{+}) \cap {}^{I} w^{J-1} (-\Phi^{+}) \right) \sqcup \left((\Phi^{+} \setminus \Phi_{I}^{+}) \cap {}^{I} w^{J-1} (\Phi^{+} \setminus \Phi_{I}^{+}) \right),$$

which is obtained from the partition $\Phi = \Phi_I^+ \sqcup (-\Phi^+) \sqcup (\Phi^+ \backslash \Phi_I^+)$ by applying ${}^I w^{J-1}$ and taking the intersection with $\Phi^+ \backslash \Phi_I^+$.

Let $w_J \in {}^{J \cap {}^l w^{J-1}(I)} W_J$ and set $w_J' := {}^{J \cap {}^l w^{J-1}(I)} w_{J,0}^{-1} w_J$. By construction, (45) induces a natural B_J -equivariant isomorphism

$$\operatorname{Gr}_{B}^{w_{J}}(\operatorname{H}_{n}(N_{J},\operatorname{c-ind}_{P_{I}}^{P_{I}^{I}w^{J}P_{J}}\sigma)) \cong \operatorname{Gr}_{B}^{w'_{J}}(\operatorname{H}^{[F:\mathbb{Q}_{p}]d_{J}-n}\operatorname{Ord}_{P_{J}}(\operatorname{c-ind}_{P'_{I'}}^{P'_{I'}w^{J}P_{J}}\sigma^{I_{w_{0}}}) \otimes (\omega \circ \delta_{J})).$$

By Theorem 3.3.3, there is a natural $B_{J,w'_{I}}$ -equivariant isomorphism

$$\mathrm{Gr}_{B}^{w'_J} \big(\mathrm{H}^{[F:\mathbb{Q}_p]d_J - n} \mathrm{Ord}_{P_J} \big(\mathrm{c\text{-}ind}_{P_{I'}^{-}}^{P_{I'}^{-} l'w^J P_J} \, \sigma^{l_{w_0}} \big) \otimes (\omega \circ \delta_J) \big) \\ \cong \mathrm{Gr}_{B_J}^{w'_J} \big(\mathrm{Ind}_{L_J \cap P_{I \cap I'w^{J-1}(I')}}^{L_J} \, \tilde{\sigma}^{J \cap l_w^{-} J - 1(I)}_{w^{J,0}} \big) \\$$

which is even $L_J \cap P_{J \cap I'w^{J-1}(I')}$ -equivariant when $w'_J = 1$. By construction, (46) and (47) induce a natural B_J -equivariant isomorphism

$$\operatorname{Gr}_{B_{J}}^{w_{J}'}\left(\operatorname{Ind}_{L_{J}\cap P_{J\cap I_{w}J-1(I')}^{L_{J}}}^{L_{J}}\tilde{\sigma}^{J\cap I_{w}J-1(I)}w_{J,0}\right) \\
\cong \operatorname{Gr}_{B_{J}}^{w_{J}}\left(\operatorname{Ind}_{L_{J}\cap P_{J\cap I_{w}J-1(I)}}^{L_{J}}\left(\operatorname{H}_{n-[F:\mathbb{Q}_{p}]d_{I_{w}J}}\left(L_{I}\cap N_{I\cap I_{w}J(J)},\sigma\right)^{I_{w}J}\otimes\left(\omega\circ\delta_{I_{w}J}\right)\right)\right).$$

Composing these three isomorphisms yields the result.

Theorem 4.3.3 with $w_J = {}^{J \cap I_W J^{-1}(I)} w_{J,0}$ yields a natural L_J -equivariant morphism analogous to (30),

$$\operatorname{Ind}_{L_{J}\cap P_{J\cap I_{w}J-1(I)}}^{L_{J}}\left(\operatorname{H}_{n-[F:\mathbb{Q}_{p}]d_{I_{w}J}}\left(L_{I}\cap N_{I\cap I_{w}J(J)},\sigma\right)^{I_{w}J}\otimes\left(\omega\circ\delta_{I_{w}J}\right)\right)\to\operatorname{H}_{n}\left(N_{J},\operatorname{c-ind}_{P_{I}}^{P_{I}^{I}w^{J}P_{J}}\sigma\right), \quad (48)$$

and Conjecture 3.3.4 is equivalent to (48) being an isomorphism. We also have analogues of Propositions 3.3.5 and 3.3.7. In the case I = J, we obtain the following analogue of Corollary 3.3.8.

Corollary 4.3.4. Let P = LN be a standard parabolic subgroup and σ be a locally admissible L-representation.

- (i) For all $n \in \mathbb{N}$ such that $0 < n < [F : \mathbb{Q}_p]$, we have $H_n(N, \operatorname{Ind}_P^G \pi) = 0$.
- (ii) If $\sigma_{L\cap s_{\alpha}Ns_{\alpha}^{-1}}=0$ for all $\alpha\in\Delta^1\setminus(\Delta^1_L\cup\Delta^{\perp,1}_L)$, then there is a natural L-equivariant isomorphism

$$\mathrm{H}_{[F:\mathbb{Q}_p]}\left(N,\operatorname{Ind}_P^G\sigma
ight)\cong\bigoplus_{lpha\in\Delta_L^{\perp,1}}\sigma^lpha\otimes(\omega\circlpha)\,.$$

Remark 4.3.5. The results hold true with P^- , N^- and ω^{-1} instead of P, N and ω respectively.

5. Application to extensions

The aim of this section is to compute the extensions between parabolically induced representations of G. In Section 5.1, we review some cuspidality and genericity properties and we prove some preliminary results on extensions which will be used in the case where G is split and Z is connected. Then, the main results are proved in Section 5.2. Finally, some of these results are lifted to characteristic 0 in Section 5.3.

5.1. *Preliminaries.* We fix a standard parabolic subgroup P = LN. We first define some cuspidality properties and discuss the relations between them.

Definition 5.1.1. We say that an admissible smooth representation σ of L over k is:

- supersingular if $\overline{\mathbb{F}}_p \otimes_k \sigma$ is supersingular (in the sense of [Abe et al. 2017a]),
- supercuspidal if it is irreducible and not a subquotient of $\operatorname{Ind}_Q^L \tau$ for any proper parabolic subgroup $Q \subset L$ with Levi quotient L_Q and any irreducible admissible smooth representation τ of L_Q over k,
- right (resp. left) cuspidal if $\operatorname{Ord}_Q \sigma = 0$ (resp. $\sigma_{N_Q} = 0$) for any proper parabolic subgroup $Q \subset L$ with unipotent radical N_Q .

Remark 5.1.2. In [Abe et al. 2017b, Definition 6.3], left and right cuspidality are defined for smooth representations using the left and right adjoint functors of Ind_Q^G , namely L_Q^L and R_Q^L . Since $\operatorname{L}_Q^L = (-)_{N_Q}$ and the restriction of R_Q^L to admissible representations is Ord_{Q^-} [op. cit., Corollary 4.13]), these definitions coincide for admissible representations.

Lemma 5.1.3. Let σ be an irreducible admissible smooth representation of L over k. The following conditions are equivalent.

- (i) σ is supercuspidal.
- (ii) σ is left and right cuspidal.

(iii) $\overline{\mathbb{F}}_p \otimes_k \sigma$ is a (finite) direct sum of supersingular representations.

In particular, σ is supersingular if and only if it is absolutely irreducible and supercuspidal.

Proof. Over $\overline{\mathbb{F}}_p$, the equivalence between (i) and (ii) is [op. cit., Corollary 6.9], and the equivalence between "supercuspidal" and "supersingular" is [Abe et al. 2017a, Theorem 5]. By [Emerton 2010b, Lemma 4.1.2], $\overline{\mathbb{F}}_p \otimes_k \sigma$ is a finite direct sum of irreducible admissible smooth representations of L over $\overline{\mathbb{F}}_p$. Since Ind_Q^L , $(-)_{N_Q}$ and Ord_Q commute with $\overline{\mathbb{F}}_p \otimes_k -$, we deduce the equivalences over k.

We now study some genericity property for smooth representations of L over k with central character. We assume that $\Delta_L^{\perp,1} \neq \varnothing$.

Lemma 5.1.4. Let σ be a smooth representation of L over k with central character $\zeta: Z_L \to k^{\times}$ and $\alpha \in \Delta_L^{\perp,1}$. If $\zeta \circ \alpha^{\vee} = \omega^{-1}$, then $\sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha) \cong \sigma$.

Proof. For convenience, we recall the construction of the representation $\sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha)$. Let $G_{\alpha} \subseteq G$ be the standard Levi subgroup corresponding to α . We fix a representative $n_{\alpha} \in \mathcal{N}$ of s_{α} . For every $\beta \in \Delta_L$ and for all integers i, j > 0, $i\alpha + j\beta \notin \Phi$ (since $\alpha \perp \beta$), thus U_{α} and U_{β} commute for every $\beta \in \Delta_L$ [Borel and Tits 1965, Proposition 2.5], or more directly using the Baker–Campbell–Hausdorff formula. We deduce that G_{α} and L normalise each other (since G_{α} and L are generated by Z and respectively $U_{\pm \alpha}$ and $(U_{\beta})_{\beta \in \pm \Delta_L}$). In particular, n_{α} normalises L (since $n_{\alpha} \in G_{\alpha}$) and the n_{α} -conjugate σ^{α} does not depend on the choice of n_{α} in $n_{\alpha}Z$ up to isomorphism (since $Z \subseteq L$). Furthermore, L normalises U_{α} and α extends (uniquely) to an algebraic character of L (since $\alpha \in \Delta^1$).

We let $I_{\alpha} \subseteq L$ be the kernel of $\alpha: L \to F^{\times}$. Note that $L = SI_{\alpha}$. We may and will assume that n_{α} lies in the subgroup of G_{α} generated by $U_{\pm \alpha}$ [Abe et al. 2017a, §II.4] so that n_{α} commutes with I_{α} . Thus, the action of I_{α} on $\sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha)$ and σ is the same.

Now, assume $\zeta \circ \alpha^{\vee} = \omega^{-1}$. For any $\lambda \in X_*(S)$ we have $\lambda - s_{\alpha}(\lambda) = \langle \alpha, \lambda \rangle \alpha^{\vee}$, so that

$$\lambda - s_{\alpha}(\lambda) \in X_*(S \cap \mathbf{Z}_L)$$
 and $\zeta \circ (\lambda - s_{\alpha}(\lambda)) = (\zeta \circ \alpha^{\vee})^{\langle \alpha, \lambda \rangle} = \omega^{-\langle \alpha, \lambda \rangle} = (\omega^{-1} \circ \alpha) \circ \lambda$.

We deduce that for any $s \in S$, $s(n_{\alpha}sn_{\alpha}^{-1})^{-1} \in S \cap Z_L$ and $\zeta(s(n_{\alpha}sn_{\alpha}^{-1})^{-1}) = (\omega^{-1} \circ \alpha)(s)$. Thus, the action of S on $\sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha)$ and σ is the same.

The following result yields a converse to Lemma 5.1.4 when G is split and Z is connected [Breuil and Herzig 2015, Proposition 2.1.1].

Lemma 5.1.5. Let σ be a smooth representation of L over k with central character $\zeta: Z_L \to k^{\times}$ and $\alpha \in \Delta_L^{\perp,1}$. Assume that there exists $\lambda \in X_*(\mathbf{Z}_L)$ such that $\langle \alpha, \lambda \rangle = 1$ and $\langle \beta, \lambda \rangle = 0$ for all $\beta \in \Delta_L^{\perp,1} \setminus \{\alpha\}$. If $\zeta \circ \alpha^{\vee} \neq \omega^{-1}$, then $s_{\alpha}(\zeta)(\omega^{-1} \circ \alpha) \neq \zeta$ and $s_{\alpha}(\zeta)(\omega^{-1} \circ \alpha) \neq s_{\beta}(\zeta)(\omega^{-1} \circ \beta)$ for all $\beta \in \Delta_L^{\perp,1} \setminus \{\alpha\}$.

Proof. We have $(s_{\alpha}(\zeta)(\omega^{-1} \circ \alpha)) \circ \lambda = (\zeta \circ \lambda)((\zeta \circ \alpha^{\vee})\omega)^{-1}$ and $(s_{\beta}(\zeta)(\omega^{-1} \circ \beta)) \circ \lambda = (\zeta \circ \lambda)$ for all $\beta \in \Delta_L^{\perp,1} \setminus \{\alpha\}$. Thus, if $s_{\alpha}(\zeta)(\omega^{-1} \circ \alpha) = \zeta$ or $s_{\alpha}(\zeta)(\omega^{-1} \circ \alpha) = s_{\beta}(\zeta)(\omega^{-1} \circ \beta)$ for some $\beta \in \Delta_L^{\perp,1} \setminus \{\alpha\}$, then precomposing each side of the equality with λ yields the equality $(\zeta \circ \alpha^{\vee})\omega = 1$.

We now give some preliminary results on extensions. Let H be a p-adic Lie group. For locally admissible smooth representations π , π' of L over A, we let $\operatorname{Ext}^{\bullet}_{H}(\pi',\pi)$ denote the A-modules of extensions computed in $\operatorname{Mod}^{\operatorname{l.adm}}_{H}(A)$ à la Yoneda or using an injective resolution of π . If π , π' are admissible, then in degree 1 it is equivalent to compute $\operatorname{Ext}^{\bullet}_{H}(\pi',\pi)$ in $\operatorname{Mod}^{\operatorname{adm}}_{H}(A)$ à la Yoneda, but this is not known in higher degree [Emerton 2010b, Remark 3.7.8], except when $H = \operatorname{GL}_{2}(\mathbb{Q}_{p})$ [Paškūnas 2013, Corollary 5.17]. Let $\widetilde{Z} \subseteq H$ be a central closed subgroup and $\zeta: \widetilde{Z} \to A^{\times}$ be a smooth character. We write $\operatorname{Mod}^{\operatorname{l.adm}}_{H,\zeta}(A)$ for the full subcategory of $\operatorname{Mod}^{\operatorname{l.adm}}_{H}(A)$ whose objects are the representations on which \widetilde{Z} acts via ζ . If \widetilde{Z} acts on π , π' via ζ , then we let $\operatorname{Ext}^{\bullet}_{H,\zeta}(\pi',\pi)$ denote the A-modules of extensions computed in $\operatorname{Mod}^{\operatorname{l.adm}}_{H,\zeta}(A)$ à la Yoneda, or equivalently using an injective resolution of π .

We now assume that G is split and we write T for the maximal split torus S = Z. Using Notation 2.3.3, we have $d_w = \ell(w)$ for all $w \in W$, so that in particular $\Delta^1 = \Delta$. Let $L' \subseteq G$ be a standard Levi subgroup such that $\Delta_L \perp \Delta_{L'}$. Note that LL' is the standard Levi subgroup corresponding to $\Delta_L \sqcup \Delta_{L'}$. Let σ be a locally admissible smooth representation of L over k with central character $\zeta: Z_L \to k^\times$. The following construction was communicated to me by N. Abe.

First, we assume that G^{der} is simply connected and we let $\widetilde{\mathbf{Z}} \subseteq \mathbf{Z}$ be a closed subgroup. Recall that this is equivalent to the existence of fundamental weights $(\mu_{\alpha})_{\alpha \in \Delta}$ [Breuil and Herzig 2015, Proposition 2.1.1]. We set $\chi := \zeta \circ \sum_{\alpha \in \Delta_{L'}} (\alpha^{\vee} \circ \mu_{\alpha})$. Thus $\chi \circ \alpha^{\vee} = 1$ for all $\alpha \in \Delta_L$ and $\chi \circ \alpha^{\vee} = \zeta \circ \alpha^{\vee}$ for all $\alpha \in \Delta_{L'}$, so that χ extends uniquely to L and $\sigma_0 := \sigma \otimes \chi^{-1}$ extends uniquely to a locally admissible smooth representation of LL' over k [Abe 2013, Lemma 3.2]. We let $\chi' : T \to k^{\times}$ be a smooth character such that $\chi'_{|Z_{L'}} = \chi_{|Z_{L'}}$, so that χ' extends uniquely to L, and we set $\sigma' := \sigma_0 \otimes \chi'$. There is a commutative diagram of k-vector spaces

$$\operatorname{Ext}_{T,\chi_{\mid Z_{L'}}}^{\bullet}(\chi',\chi) \longrightarrow \operatorname{Ext}_{L',\chi_{\mid Z_{L'}}}^{\bullet}(\operatorname{Ind}_{B_{L'}^{-}}^{L'}\chi',\operatorname{Ind}_{B_{L'}^{-}}^{L'}\chi)$$

$$\downarrow \qquad \qquad \downarrow \qquad \qquad \downarrow$$

$$\operatorname{Ext}_{L,\chi_{\mid \widetilde{Z}}}^{\bullet}(\chi',\chi) \longrightarrow \operatorname{Ext}_{LL',\chi_{\mid \widetilde{Z}}}^{\bullet}(\operatorname{Ind}_{LB_{L'}^{-}}^{LL'}\chi',\operatorname{Ind}_{LB_{L'}^{-}}^{LL'}\chi)$$

$$\downarrow \qquad \qquad \downarrow \qquad \qquad \downarrow$$

$$\operatorname{Ext}_{L,\zeta_{\mid \widetilde{Z}}}^{\bullet}(\sigma',\sigma) \longrightarrow \operatorname{Ext}_{LL',\zeta_{\mid \widetilde{Z}}}^{\bullet}(\operatorname{Ind}_{LB_{L'}^{-}}^{LL'}\sigma',\operatorname{Ind}_{LB_{L'}^{-}}^{LL'}\sigma)$$

$$(49)$$

where the horizontal arrows are induced by the functors $\operatorname{Ind}_{B_{L'}}^{L'}$ and $\operatorname{Ind}_{LB_{L'}}^{LL'}$, the upper vertical arrows are induced by extending representations to L and LL', and the lower vertical arrows are induced by tensoring representations with σ_0 . Furthermore, the lower horizontal arrow of (49) composed with the k-linear morphism induced by the functor $\operatorname{Ind}_{B_{-L'}}^G$:

$$\operatorname{Ext}^{\bullet}_{LL',\zeta_{|\widetilde{Z}|}}(\operatorname{Ind}_{LB_{I'}}^{LL'}\sigma',\operatorname{Ind}_{LB_{I'}}^{LL'}\sigma) \to \operatorname{Ext}^{\bullet}_{G,\zeta_{|\widetilde{Z}|}}(\operatorname{Ind}_{P^{-}}^{G}\sigma',\operatorname{Ind}_{P^{-}}^{G}\sigma), \tag{50}$$

is the k-linear morphism induced by the functor $\operatorname{Ind}_{P^-}^G$:

$$\operatorname{Ext}_{L,\zeta_{|\widetilde{Z}}}^{\bullet}(\sigma',\sigma) \to \operatorname{Ext}_{G,\zeta_{|\widetilde{Z}}}^{\bullet}(\operatorname{Ind}_{P^{-}}^{G}\sigma',\operatorname{Ind}_{P^{-}}^{G}\sigma). \tag{51}$$

- **Lemma 5.1.6.** (i) In degree 1, there is a k-linear injection from the cokernel of the upper horizontal arrow of (49) into the cokernel of (51).
- (ii) Assume **Z** connected. In all degrees, there is a k-linear injection from the kernel of the upper horizontal arrow of (49) into the kernel of (51).

Proof. We prove (i). The map in question is induced by the composite right-hand side vertical arrow of (49) composed with (50). Let \mathcal{E} be an extension of $\operatorname{Ind}_{B_{L'}^-}^{L'}\chi'$ by $\operatorname{Ind}_{B_{L'}^-}^{L'}\chi$ with central character $\chi_{|Z_{L'}}$ (so that \mathcal{E} extends to LL'). Then $\operatorname{Ind}_{P-L'}^G(\sigma_0 \otimes \mathcal{E})$ is an extension of $\operatorname{Ind}_{P-}^G\sigma'$ by $\operatorname{Ind}_{P-}^G\sigma$ on which \widetilde{Z} acts via ζ . There are L-equivariant isomorphisms

$$\operatorname{Ord}_{P}\left(\operatorname{Ind}_{P^{-}L'}^{G}\left(\sigma_{0}\otimes\mathcal{E}\right)\right)\cong\operatorname{Ord}_{LB_{L'}}\left(\sigma_{0}\otimes\mathcal{E}\right)\cong\sigma_{0}\otimes\operatorname{Ord}_{B_{L'}}\mathcal{E}.$$

The first one results from [Emerton 2010a, Proposition 4.3.4] and the second one from the fact that $U_{L'}$ acts trivially on σ_0 (note that $\operatorname{Ord}_{B_{L'}} \mathcal{E}$ extends to L). If the class of \mathcal{E} is not in the image of the upper horizontal arrow of (49), then there is a T-equivariant isomorphism $\operatorname{Ord}_{B_{L'}} \mathcal{E} \cong \chi$, hence an L-equivariant isomorphism $\operatorname{Ord}_{P-L'}(\sigma_0 \otimes \mathcal{E}) \cong \sigma$; thus the class of $\operatorname{Ind}_{P-L'}^G(\sigma_0 \otimes \mathcal{E})$ is not in the image of (51).

We prove (ii). The map in question is induced by the left-hand side composite vertical arrow of (49). Thus, it is enough to prove that the latter is injective. We assume **Z** connected. Recall that this is equivalent to the existence of fundamental coweights $(\lambda_{\alpha})_{\alpha \in \Delta}$ [Breuil and Herzig 2015, Proposition 2.1.1]. We let $T' \subseteq T$ be the closed subgroup generated by the images of $(\lambda_{\alpha})_{\alpha \in \Delta_{L'}}$, so that $T' \subseteq \mathbf{Z}_L$ and the product induces an isomorphism $T' \times \mathbf{Z}_{L'} \xrightarrow{\sim} T$. There is a commutative diagram of k-vector spaces

$$\operatorname{Ext}_{T,\chi_{\mid Z_{L'}}}^{\bullet}(\chi',\chi) \stackrel{\sim}{\longrightarrow} \operatorname{Ext}_{T'}^{\bullet}(\chi'_{\mid T'},\chi_{\mid T'})$$

$$\downarrow \qquad \qquad \qquad \parallel$$

$$\operatorname{Ext}_{L,\chi_{\mid \widetilde{Z}}}^{\bullet}(\chi',\chi) \stackrel{\sim}{\longrightarrow} \operatorname{Ext}_{T'}^{\bullet}(\chi'_{\mid T'},\chi_{\mid T'})$$

$$\downarrow \qquad \qquad \downarrow$$

$$\operatorname{Ext}_{L,\zeta_{\mid \widetilde{Z}}}^{\bullet}(\sigma',\sigma) \stackrel{\sim}{\longrightarrow} \operatorname{Ext}_{T'}^{\bullet}(\sigma'_{\mid T'},\sigma_{\mid T'})$$

where the horizontal arrows are induced by restricting representations to T' (the upper one is bijective with inverse induced by tensoring representations with $\chi_{|Z_{L'}}$, and the middle and lower ones are well-defined since a locally admissible smooth representation of L over k is locally Z_L -finite [Emerton 2010a, Lemma 2.3.4], the left-hand side vertical arrows are the same as in (49) and the lower right-hand side vertical arrow is induced by tensoring representations with $\sigma_{0|T'}$ (it is injective since T' acts on σ_0 via $\zeta \chi^{-1}$).

Now, we do not assume G^{der} simply connected. Instead we take a z-extension of G, i.e., an exact sequence of linear algebraic F-groups

$$1 \to \widetilde{\mathbf{Z}} \to \widetilde{\mathbf{G}} \to \mathbf{G} \to 1$$

such that \widetilde{G} is reductive with simply connected derived subgroup and \widetilde{Z} is a central torus [Colliot-Thélène 2008, §3.1]. The projection $\widetilde{G} \to G$ identifies the corresponding root systems. We let $\widetilde{P} \subseteq \widetilde{G}$ be the standard parabolic subgroups corresponding to P and $\widetilde{L} \subseteq \widetilde{P}$ be the standard Levi subgroup corresponding to L. Note that \widetilde{L} is a z-extensions of L. We let σ' be a locally admissible smooth representation of L over k with central character ζ . By inflation, we obtain locally admissible smooth representations $\widetilde{\sigma}$ and $\widetilde{\sigma}'$ of \widetilde{L} over k. There is a commutative diagram of k-vector spaces

$$\operatorname{Ext}_{L}^{\bullet}(\sigma',\sigma) \longrightarrow \operatorname{Ext}_{G}^{\bullet}\left(\operatorname{Ind}_{P^{-}}^{G}\sigma',\operatorname{Ind}_{P^{-}}^{G}\sigma\right)$$

$$\downarrow^{\downarrow} \qquad \qquad \downarrow^{\downarrow}$$

$$\operatorname{Ext}_{\widetilde{L},\zeta_{|\widetilde{Z}}}^{\bullet}(\tilde{\sigma}',\tilde{\sigma}) \longrightarrow \operatorname{Ext}_{\widetilde{G},\zeta_{|\widetilde{Z}}}^{\bullet}\left(\operatorname{Ind}_{\widetilde{P}^{-}}^{\widetilde{G}}\tilde{\sigma}',\operatorname{Ind}_{\widetilde{P}^{-}}^{\widetilde{G}}\tilde{\sigma}\right)$$
(52)

where the horizontal arrows are induced by the functors $\operatorname{Ind}_{P^-}^G$ and $\operatorname{Ind}_{\widetilde{P}^-}^{\widetilde{G}}$ and the vertical arrows are induced by inflating representations to \widetilde{L} and \widetilde{G} (they are well defined and bijective since $\zeta_{|\widetilde{Z}}$ is trivial).

Proposition 5.1.7. Assume $F = \mathbb{Q}_p$ and G split. Let σ be a locally admissible smooth representation of L over k with central character $\zeta: Z_L \to k^{\times}$.

(i) Assume $\Delta_L^{\perp} \neq \emptyset$ and let $\alpha \in \Delta_L^{\perp}$. If $\zeta \circ \alpha^{\vee} \neq \omega^{-1}$, then the k-linear injection

$$\operatorname{Ext}^1_L\left(\sigma^\alpha\otimes(\omega^{-1}\circ\alpha),\sigma\right)\hookrightarrow\operatorname{Ext}^1_G\left(\operatorname{Ind}_{P^-}^G\sigma^\alpha\otimes(\omega^{-1}\circ\alpha),\operatorname{Ind}_{P^-}^G\sigma\right)$$

induced by the functor $\operatorname{Ind}_{P^-}^G$ is not surjective.

(ii) If p = 2, then the functor $\operatorname{Ind}_{P^-}^G$ induces a k-linear injection

$$\operatorname{Ext}^1_L(\sigma,\sigma) \hookrightarrow \operatorname{Ext}^1_G\left(\operatorname{Ind}_{P^-}^G\sigma,\operatorname{Ind}_{P^-}^G\sigma\right)$$

whose cokernel is of dimension at least $\operatorname{card}\{\alpha \in \Delta_L^{\perp} \mid \zeta \circ \alpha^{\vee} = 1\}.$

Remark 5.1.8. We expect the results to hold true for a nonsplit reductive group with $\Delta_L^{\perp,1}$ instead of Δ_L^{\perp} .

Proof. By taking a z-extension of G and using (52), we can and do assume that G^{der} is simply connected and prove analogous results for the morphism (51).

Assume $\Delta_L^{\perp} \neq \varnothing$ and let $\alpha \in \Delta_L^{\perp}$. We use Lemma 5.1.6(i) with L' defined by $\Delta_{L'} = \{\alpha\}$, $\chi = \zeta \circ \alpha^{\vee} \circ \mu_{\alpha}$, and $\chi' = s_{\alpha}(\chi)(\omega^{-1} \circ \alpha)$, so that $\sigma' = \sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha)$ (since $\sigma_0^{\alpha} = \sigma_0$ by Lemma 5.1.4 with 1 instead of ω). If $\zeta \circ \alpha^{\vee} \neq \omega^{-1}$, then the upper horizontal arrow of (49) in degree 1 is not surjective by the mod p analogue of [Hauseux 2017, Lemme 3.1.4] (since $\chi \circ \alpha^{\vee} = \zeta \circ \alpha^{\vee} = 1$); thus (51) in degree 1 is not surjective, hence (i).

We use Lemma 5.1.6(i) with L' defined by $\Delta_{L'} = \{\alpha \in \Delta_L^{\perp} \mid \zeta \circ \alpha^{\vee} = 1\}$, $\chi = \zeta \circ \sum_{\alpha \in \Delta_{L'}} (\alpha^{\vee} \circ \mu_{\alpha})$, and $\chi' = \chi$, so that $\sigma' = \sigma$. If p = 2, then the cokernel of the upper horizontal arrow of (49) in degree 1 is of dimension at least card $\Delta_{L'}$ [Hauseux 2017, Théorème 3.2.4(ii) and Remarque 3.2.5(ii)] (since $\chi \circ \alpha^{\vee} = \zeta \circ \alpha^{\vee} = 1$ so that $s_{\alpha}(\chi) = \chi$ by Lemma 5.1.4 for all $\alpha \in \Delta_{L'}$), noting that all the extensions constructed there have a central character [op. cit., Lemme 3.1.5]; thus the cokernel of (51) in degree 1 is of dimension at least card $\Delta_{L'}$, hence (ii).

Proposition 5.1.9. Assume $F = \mathbb{Q}_p$, G split and Z connected. Let σ be a locally admissible smooth representation of L over k with central character $\zeta: Z_L \to k^{\times}$. If $p \neq 2$, then the functor $\operatorname{Ind}_{P^-}^G$ induces a k-linear morphism

$$\operatorname{Ext}_{L}^{2}\left(\sigma,\sigma\right) \to \operatorname{Ext}_{G}^{2}\left(\operatorname{Ind}_{P^{-}}^{G}\sigma,\operatorname{Ind}_{P^{-}}^{G}\sigma\right)$$

whose kernel is of dimension at least card{ $\alpha \in \Delta_L^{\perp} \mid \zeta \circ \alpha^{\vee} = \omega^{-1}$ }.

Proof. By taking a z-extension of G (noting that the centre of \widetilde{G} is also connected because \widetilde{Z} is connected) and using (52), we can and do assume that G^{der} is simply connected and prove an analogous result for the morphism (51).

We use Lemma 5.1.6(ii) with L' defined by $\Delta_{L'} = \{\alpha \in \Delta_L^{\perp} \mid \zeta \circ \alpha^{\vee} = \omega^{-1}\}$ and $\chi' = \chi$, so that $\sigma' = \sigma$. If $p \neq 2$, then we see in the proof of [Hauseux 2017, Théorème 3.2.4(i)] that the kernel of the upper horizontal arrow of (49) in degree 2 is of dimension at least card $\Delta_{L'}$ (since $\chi \circ \alpha^{\vee} = \zeta \circ \alpha^{\vee} = \omega^{-1}$ for all $\alpha \in \Delta_{L'}$); thus the kernel of (49) in degree 2 is also of dimension at least card $\Delta_{L'}$, hence the result. \square

5.2. Extensions between parabolically induced representations. We begin with a result when there is no inclusion between the two parabolic subgroups, assuming a special case of Conjecture 3.3.4 (see also Remark 3.3.6).

Proposition 5.2.1. Let P = LN, P' = L'N' be standard parabolic subgroups and σ , σ' be admissible smooth representations of L, L' respectively over k. Assume Conjecture 3.3.4 is true for A = k, n = 1 and ${}^{I}w^{J} = 1$. If $P' \nsubseteq P$, $P \nsubseteq P'$, and σ , σ' are right, left cuspidal respectively, then

$$\operatorname{Ext}_G^1\left(\operatorname{Ind}_{P'^-}^G\sigma',\operatorname{Ind}_{P^-}^G\sigma\right)=0.$$

Proof. We put $I := \Delta_L$ and $J := \Delta_{L'}$. Using (31), (39) with $\pi = \operatorname{Ind}_{P_I}^G \sigma$ and P_J , L_J , σ' instead of P, L, σ respectively yields an exact sequence of k-vector spaces

$$0 \to \operatorname{Ext}_{L_{J}}^{1}\left(\sigma', \operatorname{Ind}_{L_{J} \cap P_{I}^{-}}^{L_{J}}\left(\operatorname{Ord}_{L_{I} \cap P_{J}} \sigma\right)\right) \to \operatorname{Ext}_{G}^{1}\left(\operatorname{Ind}_{P_{I}^{-}}^{G} \sigma', \operatorname{Ind}_{P_{I}^{-}}^{G} \sigma\right) \to \operatorname{Hom}_{L_{J}}\left(\sigma', \operatorname{H}^{1}\operatorname{Ord}_{P_{J}}\left(\operatorname{Ind}_{P_{I}^{-}}^{G} \sigma\right)\right).$$

$$(53)$$

Assume $I \not\subseteq J$ and σ right cuspidal. Then $\operatorname{Ord}_{L_I \cap P_J} \sigma = 0$ (since $L_I \cap P_J$ is a proper parabolic subgroup of L_I) and there is a natural L_J -equivariant isomorphism

$$\operatorname{Ind}_{L_I \cap P_I^-}^{L_J} \left(\operatorname{H}^1 \operatorname{Ord}_{L_I \cap P_J} \sigma \right) \xrightarrow{\sim} \operatorname{H}^1 \operatorname{Ord}_{P_J} \left(\operatorname{Ind}_{P_I^-}^G \sigma \right). \tag{54}$$

Indeed, by assumption (30) is a natural L_I -equivariant isomorphism

$$\operatorname{Ind}_{L_{I} \cap P_{I}^{-}}^{L_{J}} \left(\operatorname{H}^{1} \operatorname{Ord}_{L_{I} \cap P_{J}} \sigma \right) \xrightarrow{\sim} \operatorname{H}^{1} \operatorname{Ord}_{P_{J}} \left(\operatorname{c-ind}_{P_{I}^{-}}^{P_{I}^{-} P_{J}} \sigma \right)$$

and by Proposition 3.3.5(i) with n=1, we have $H^1\mathrm{Ord}_{P_J}(\mathrm{c\text{-}ind}_{P_I-}^{P_I^-l_w^JP_J}\sigma)=0$ for all ${}^Iw^J\in {}^IW^J$ such that ${}^Iw^J\neq 1$ (since either $d_{{}^Iw^J}=1$ and $\mathrm{Ord}_{L_I\cap P_{I\cap I_w^J(J)}}\sigma=0$, or $d_{{}^Iw^J}>1$ and $1-[F:\mathbb{Q}_p]d_{{}^Iw^J}<0$). Thus, we deduce (54) from Proposition 3.3.1.

Assume $J \not\subseteq I$ and σ' left cuspidal. Then $\sigma'_{L_J \cap N_I^-} = 0$ (since $L_J \cap P_I^-$ is a proper parabolic subgroup of L_J) and using (40) with $\pi = \sigma'$ and L_J , $L_J \cap P_I^-$, $L_{J \cap I}$, $L_J \cap N_I^-$, $H^1 \text{Ord}_{L_I \cap P_J} \sigma$ instead of G, P, L,

N, σ respectively, we obtain

$$\operatorname{Hom}_{L_J}(\sigma',\operatorname{Ind}_{L_I\cap P_I^-}^{L_J}(\operatorname{H}^1\operatorname{Ord}_{L_I\cap P_J}\sigma))=0.$$

We conclude using (53).

Now, we prove unconditional results whenever there is an inclusion between the two parabolic subgroups. We treat the cases $F = \mathbb{Q}_p$ and $F \neq \mathbb{Q}_p$ separately.

Theorem 5.2.2. Assume $F = \mathbb{Q}_p$. Let P = LN, P' = L'N' be standard parabolic subgroups and σ , σ' be admissible smooth representations of L, L' respectively over k.

(i) If P' = P, σ , σ' are supercuspidal and $\sigma' \ncong \sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha)$ for all $\alpha \in \Delta_L^{\perp,1}$, then the functor $\operatorname{Ind}_{P^-}^G$ induces a k-linear isomorphism

$$\operatorname{Ext}_L^1(\sigma',\sigma) \xrightarrow{\sim} \operatorname{Ext}_G^1(\operatorname{Ind}_{P^-}^G \sigma', \operatorname{Ind}_{P^-}^G \sigma).$$

(ii) If $P' \subsetneq P$ and σ is right cuspidal, then the functor $\operatorname{Ind}_{P^-}^G$ induces a k-linear isomorphism

$$\operatorname{Ext}^1_L\big(\operatorname{Ind}^L_{L\cap P'^-}\sigma',\sigma\big)\xrightarrow{\sim}\operatorname{Ext}^1_G\big(\operatorname{Ind}^G_{P'^-}\sigma',\operatorname{Ind}^G_{P^-}\sigma\big).$$

(iii) If $P \subsetneq P'$ and σ' is left cuspidal, then the functor $\operatorname{Ind}_{P'^-}^G$ induces a k-linear isomorphism

$$\operatorname{Ext}^1_{L'}(\sigma',\operatorname{Ind}^{L'}_{L'\cap P^-}\sigma) \xrightarrow{\sim} \operatorname{Ext}^1_G(\operatorname{Ind}^G_{P'^-}\sigma',\operatorname{Ind}^G_{P^-}\sigma).$$

Remark 5.2.3. Assume P' = P and σ , σ' irreducible. In general, we do not know the dimension of the cokernel of the k-linear injection $\operatorname{Ext}^1_L(\sigma',\sigma) \hookrightarrow \operatorname{Ext}^1_G(\operatorname{Ind}_{P^-}^G\sigma',\operatorname{Ind}_{P^-}^G\sigma)$ induced by $\operatorname{Ind}_{P^-}^G$, but we prove that it is at most $\operatorname{card}\{\alpha \in \Delta_L^{\perp,1} \mid \sigma' \cong \sigma^\alpha \otimes (\omega^{-1} \circ \alpha)\}$ whenever σ is right cuspidal or σ' is left cuspidal (see the proof). If σ , σ' are supersingular, then letting $\zeta: Z_L \to k^\times$ denote the central character of σ [Emerton 2010b, Lemma 4.1.7], we expect this dimension to be equal to

$$\operatorname{card}\left\{\alpha\in\Delta_L^{\perp,1}\;\middle|\;\sigma'\cong\sigma^\alpha\otimes\left(\omega^{-1}\circ\alpha\right)\;\operatorname{and}\;\zeta\circ\alpha^\vee\neq\omega^{-1}\right\}$$

except when p=2 and in some exceptional cases [Hauseux 2017, Remarque 3.2.5] when G is split and P=B. We prove this when G is split and Z is connected (see Theorem 5.2.7 below), in which case the cardinal above is equal to 1 if $\sigma' \cong \sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha) \ncong \sigma$ for some $\alpha \in \Delta_L^{\perp,1}$ and 0 otherwise by Lemma 5.1.5. When G is split but Z is not connected, one could prove that the cardinal above is a lower bound using Proposition 5.1.7(i) and some generalisation of [Hauseux 2017, §2.2] for $P \neq B$.

Proof. We prove slightly more general results.

Assume $P' \subseteq P$ and σ satisfies the condition in Corollary 3.3.8(ii), e.g., σ is right cuspidal. Using [Emerton 2010a, Proposition 4.3.4] and Corollary 3.3.8(ii), (39) with $\pi = \operatorname{Ind}_{P^-}^G \sigma$ and $\operatorname{Ind}_{L \cap P'^-}^L \sigma'$ instead of σ yields an exact sequence of k-vector spaces

$$0 \to \operatorname{Ext}_L^1\big(\operatorname{Ind}_{L \cap P'^-}^L \sigma', \sigma\big) \to \operatorname{Ext}_G^1\big(\operatorname{Ind}_{P'^-}^G \sigma', \operatorname{Ind}_{P^-}^G \sigma\big) \to \bigoplus_{\alpha \in \Delta_L^{\perp, 1}} \operatorname{Hom}_L\big(\operatorname{Ind}_{L \cap P'^-}^L \sigma', \sigma^\alpha \otimes (\omega^{-1} \circ \alpha)\big). \tag{55}$$

If P' = P and σ , σ' are irreducible, then $\sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha)$ is also irreducible for all $\alpha \in \Delta_L^{\perp,1}$, and thus the last term of (55) has dimension equal to $\operatorname{card}\{\alpha \in \Delta_L^{\perp,1} \mid \sigma' \cong \sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha)\}$, hence (i). If $P' \subsetneq P$ and σ is right cuspidal, then $L \cap P'$ is a proper parabolic subgroup of L and

$$\operatorname{Ord}_{L\cap P'}\left(\sigma^{\alpha}\otimes\left(\omega^{-1}\circ\alpha\right)\right)\cong\left(\operatorname{Ord}_{L\cap P'}\sigma\right)^{\alpha}\otimes\left(\omega^{-1}\circ\alpha\right)=0$$

for all $\alpha \in \Delta_L^{\perp,1}$; thus the last term of (55) is zero by [Emerton 2010a, Theorem 4.4.6], hence (ii).

Assume $P \subseteq P'$ and σ' satisfies the condition in Corollary 4.3.4(ii) for $P'^- = L'N'^-$, e.g., σ' is left cuspidal. Using [Vignéras 2016, Theorem 5.3, 3] and Corollary 4.3.4(ii) for $P'^- = L'N'^-$ (see Remark 4.3.5), (43) with $\pi = \operatorname{Ind}_{P'^-}^G \sigma'$ and P'^- , L', N'^- , $\operatorname{Ind}_{L'\cap P^-}^{L'} \sigma$ instead of P, L, N, σ respectively yields an exact sequence of k-vector spaces

$$0 \to \operatorname{Ext}^{1}_{L'} \left(\sigma', \operatorname{Ind}^{L'}_{L' \cap P^{-}} \sigma \right) \to \operatorname{Ext}^{1}_{G} \left(\operatorname{Ind}^{G}_{P'^{-}} \sigma', \operatorname{Ind}^{G}_{P^{-}} \sigma \right) \to \bigoplus_{\alpha \in \Delta^{\perp, 1}_{L'}} \operatorname{Hom}_{L'} \left(\sigma'^{\alpha} \otimes (\omega^{-1} \circ \alpha), \operatorname{Ind}^{L'}_{L' \cap P^{-}} \sigma \right). \tag{56}$$

If P' = P and σ , σ' are irreducible, then $\sigma'^{\alpha} \otimes (\omega^{-1} \circ \alpha)$ is also irreducible for all $\alpha \in \Delta_{L}^{\perp,1}$, and thus the last term of (56) has dimension equal to $\operatorname{card}\{\alpha \in \Delta_{L'}^{\perp,1} \mid \sigma' \cong \sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha)\}$, hence (i). If $P \subsetneq P'$ and σ' is left cuspidal, then $L' \cap P^-$ is a proper parabolic subgroup of L' and

$$(\sigma'^{\alpha} \otimes (\omega^{-1} \circ \alpha))_{L' \cap N^{-}} \cong (\sigma'_{L' \cap N^{-}})^{\alpha} \otimes (\omega^{-1} \circ \alpha) = 0$$

for all $\alpha \in \Delta_L^{\perp,1}$; thus the last term of (56) is zero using (40) with $\pi = \sigma'^{\alpha} \otimes (\omega^{-1} \circ \alpha)$ and $L', L' \cap P^-, L' \cap L, L' \cap N^-$ instead of G, P, L, N respectively, hence (iii).

Theorem 5.2.4. Assume $F \neq \mathbb{Q}_p$. Let P = LN be a standard parabolic subgroup. The functor $\operatorname{Ind}_{P^-}^G$ induces an A-linear isomorphism

$$\operatorname{Ext}_L^1(\sigma',\sigma) \xrightarrow{\sim} \operatorname{Ext}_G^1\left(\operatorname{Ind}_{P^-}^G\sigma',\operatorname{Ind}_{P^-}^G\sigma\right)$$

for all locally admissible smooth representations σ , σ' of L over A.

Proof. Let σ , σ' be locally admissible smooth representations of L over A. Using [Emerton 2010a, Proposition 4.3.4] and Corollary 3.3.8(i), (39) with $\pi = \operatorname{Ind}_{P^-}^G \sigma$ and σ' instead of σ yields the isomorphism in the statement.

Corollary 5.2.5. Assume $F \neq \mathbb{Q}_p$. Let P = LN, P' = L'N' be standard parabolic subgroups and σ , σ' be admissible smooth representations of L, L' respectively over k.

(i) If $P' \subseteq P$, then the functor $\operatorname{Ind}_{P^-}^G$ induces a k-linear isomorphism

$$\operatorname{Ext}^1_L(\operatorname{Ind}^L_{L\cap P'^-}\sigma',\sigma)\xrightarrow{\sim}\operatorname{Ext}^1_G(\operatorname{Ind}^G_{P'^-}\sigma',\operatorname{Ind}^G_{P^-}\sigma).$$

(ii) If $P \subseteq P'$, then the functor $\operatorname{Ind}_{P'^-}^G$ induces a k-linear isomorphism

$$\operatorname{Ext}^1_{L'}(\sigma',\operatorname{Ind}^{L'}_{L'\cap P^-}\sigma)\xrightarrow{\sim} \operatorname{Ext}^1_G(\operatorname{Ind}^G_{P'^-}\sigma',\operatorname{Ind}^G_{P^-}\sigma).$$

Remark 5.2.6. Theorem 5.2.2(i) and Theorem 5.2.4 are encompassed in a more general (but conditional to a conjecture of Emerton) result. Let P = LN be a standard parabolic subgroup, σ , σ' be locally

admissible smooth representations of L over A and $n \in \mathbb{N}$. The functor $\operatorname{Ind}_{P^-}^G$ induces an A-linear morphism

$$\operatorname{Ext}_{L}^{n}(\sigma',\sigma) \to \operatorname{Ext}_{G}^{n}(\operatorname{Ind}_{P^{-}}^{G}\sigma',\operatorname{Ind}_{P^{-}}^{G}\sigma)$$
(57)

and there is a spectral sequence of A-modules [Emerton 2010b, (3.7.4)]

$$\operatorname{Ext}_{L}^{i}(\sigma', \mathsf{R}^{j}\operatorname{Ord}_{P}(\operatorname{Ind}_{P^{-}}^{G}\sigma)) \Rightarrow \operatorname{Ext}_{G}^{i+j}(\operatorname{Ind}_{P^{-}}^{G}\sigma', \operatorname{Ind}_{P^{-}}^{G}\sigma)$$
(58)

where $R^{\bullet}Ord_P$ denotes the right derived functors of $Ord_P : Mod_G^{l.adm}(A) \to Mod_L^{l.adm}(A)$. Now assume that [op. cit., Conjecture 3.7.2] is true, i.e., $R^{\bullet}Ord_P \xrightarrow{\sim} H^{\bullet}Ord_P$. Using Corollary 3.3.8, one can deduce from (58) that:

- if $n < [F : \mathbb{Q}_p]$, then (57) is an isomorphism;
- if $n = [F : \mathbb{Q}_p]$, then (57) is injective and if furthermore σ, σ' are supercuspidal, then the dimension of its cokernel is at most card $\{\alpha \in \Delta_L^{\perp, 1} \mid \sigma' \cong \sigma^\alpha \otimes (\omega^{-1} \circ \alpha)\}$.

One can also generalise Proposition 5.2.1 and Theorem 5.2.2 (ii) and (iii) in all degrees $n \leq [F : \mathbb{Q}_p]$.

Finally, we complete Theorem 5.2.2(i) when G is split and Z is connected.

Theorem 5.2.7. Assume $F = \mathbb{Q}_p$, G split and Z connected. Let P = LN be a standard parabolic subgroup and σ , σ' be supersingular representations of L over k.

(i) If $\sigma' \cong \sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha) \ncong \sigma$ for some $\alpha \in \Delta_L^{\perp}$, then $\operatorname{Ext}_L^1(\sigma', \sigma) = 0$ and

$$\dim_k \operatorname{Ext}_G^1(\operatorname{Ind}_{P^-}^G \sigma', \operatorname{Ind}_{P^-}^G \sigma) = 1.$$

(ii) If either $\sigma' \cong \sigma$ and $p \neq 2$, or $\sigma' \ncong \sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha)$ for any $\alpha \in \Delta_{L}^{\perp}$, then the functor $\operatorname{Ind}_{P^{-}}^{G}$ induces a k-linear isomorphism

$$\operatorname{Ext}_L^1(\sigma',\sigma) \xrightarrow{\sim} \operatorname{Ext}_G^1(\operatorname{Ind}_{P^-}^G \sigma',\operatorname{Ind}_{P^-}^G \sigma).$$

(iii) If p = 2, then the functor $\operatorname{Ind}_{P^-}^G$ induces a k-linear injection

$$\operatorname{Ext}^1_L(\sigma',\sigma) \hookrightarrow \operatorname{Ext}^1_G(\operatorname{Ind}_{P^-}^G\sigma',\operatorname{Ind}_{P^-}^G\sigma)$$

whose cokernel is of dimension $\operatorname{card} \{ \alpha \in \Delta_L^{\perp} \mid \sigma' \cong \sigma^{\alpha} \}.$

Proof. Since σ is absolutely irreducible, it has a central character $\zeta: Z_L \to k^{\times}$ [Emerton 2010b, Lemma 4.1.7].

We first assume that $\sigma' \cong \sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha) \ncong \sigma$ for some $\alpha \in \Delta_{L}^{\perp}$. We have $\zeta \circ \alpha^{\vee} \neq \omega^{-1}$ by Lemma 5.1.4, so that σ and σ' have distinct central characters by Lemma 5.1.5; thus $\operatorname{Ext}_{L}^{1}(\sigma', \sigma) = 0$ [Paškūnas 2010, Proposition 8.1]. Furthermore, $\sigma' \ncong \sigma^{\beta} \otimes (\omega^{-1} \circ \beta)$ for any $\beta \in \Delta_{L}^{\perp} \setminus \{\alpha\}$ (since the central characters are distinct by Lemma 5.1.5). Using (55) with P' = P and L' = L, we deduce that

$$\dim_k \operatorname{Ext}_G^1 \left(\operatorname{Ind}_{P^-}^G \sigma', \operatorname{Ind}_{P^-}^G \sigma \right) \leq 1.$$

Since the left-hand side is nonzero by Proposition 5.1.7(i), this proves (i).

We now prove (ii). If $\sigma' \ncong \sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha)$ for any $\alpha \in \Delta_L^{\perp}$, then the result follows from Theorem 5.2.2(i). Assume that $\sigma' \cong \sigma$ and $p \neq 2$. The terms of low degree of (58) form an exact sequence of k-vector spaces

$$0 \to \operatorname{Ext}_{L}^{1}(\sigma, \sigma) \to \operatorname{Ext}_{G}^{1}\left(\operatorname{Ind}_{P^{-}}^{G}\sigma, \operatorname{Ind}_{P^{-}}^{G}\sigma\right) \to \operatorname{Hom}_{L}\left(\sigma, R^{1}\operatorname{Ord}_{P}\left(\operatorname{Ind}_{P^{-}}^{G}\sigma\right)\right)$$
$$\to \operatorname{Ext}_{L}^{2}\left(\sigma, \sigma\right) \to \operatorname{Ext}_{G}^{2}\left(\operatorname{Ind}_{P^{-}}^{G}\sigma, \operatorname{Ind}_{P^{-}}^{G}\sigma\right). \tag{59}$$

Since there is an injection of functors $R^1Ord_P \hookrightarrow H^1Ord_P$ [Emerton 2010b, Remark 3.7.3], we deduce from Corollary 3.3.8(ii) and Lemma 5.1.5 that

$$\dim_k \operatorname{Hom}_L(\sigma, R^1 \operatorname{Ord}_P(\operatorname{Ind}_{P^-}^G \sigma)) \leq \operatorname{card}\{\alpha \in \Delta_L^{\perp} \mid \zeta \circ \alpha^{\vee} = \omega^{-1}\}.$$

Thus, we deduce from Proposition 5.1.9 that the third arrow of (59) is zero, hence the result.

Finally, we assume p=2 and we prove (iii). By Proposition 5.1.7(ii) we have a lower bound and using (55) with P'=P and L'=L we obtain an upper bound. Using Lemmas 5.1.4 and 5.1.5 together with the fact that $\omega=1$, we see that both are equal to $\operatorname{card}\{\alpha\in\Delta_L^\perp\mid\sigma'\cong\sigma^\alpha\}$.

Remark 5.2.8. Theorem 5.2.7(i) can also be generalised in the context of Remark 5.2.6. Let P = LN be a standard parabolic subgroup and σ , σ' be supersingular representations of L over k such that $\sigma' \cong \sigma^{\alpha} \otimes (\omega^{-1} \circ \alpha) \ncong \sigma$ for some $\alpha \in \Delta_L^{\perp}$. Assume G split, Z connected and [Emerton 2010b, Conjecture 3.7.2] is true. Using Corollary 3.3.8 and Lemma 5.1.5, one can deduce from (58) that $\operatorname{Ext}_L^{[F:\mathbb{Q}_p]}(\sigma',\sigma)=0$ and

$$\dim_k \operatorname{Ext}_G^{[F:\mathbb{Q}_p]} \left(\operatorname{Ind}_{P^-}^G \sigma', \operatorname{Ind}_{P^-}^G \sigma \right) = 1.$$

Corollary 5.2.9. Assume **G** split and **Z** connected. If Conjecture 3.3.4 is true for A = k, n = 1 and ${}^{I}w^{J} = 1$, then [Hauseux 2016b, Conjecture 3.17] is true.

Proof. Even though [Hauseux 2016b, Conjecture 3.17] is formulated under the hypotheses G split, Z connected and G^{der} simply connected, we do not need the last one to prove it: (i) is Proposition 5.2.1, which is conditional to Conjecture 3.3.4 for A=k, n=1 and ${}^Iw^J=1$; (ii) is Theorem 5.2.7(i); (iii) and (iv) are Corollary 5.2.5(i) and (ii) respectively when $F\neq \mathbb{Q}_p$, Theorem 5.2.2(ii) and (iii) respectively when $F=\mathbb{Q}_p$ and $P'\neq P$, Theorem 5.2.7(ii) when $F=\mathbb{Q}_p$, P'=P and $p\neq 2$, and Theorem 5.2.7(iii) when $F=\mathbb{Q}_p$, P'=P and P=0 and P=0 (noting that if P=0, then P=0 and P=0 is irreducible if and only if P=0 for all P=0 for all P=0 and P=0 (noting that if P=0, then P=0 and P=0 is irreducible if and only if P=0 for all P=0 f

5.3. Results for unitary continuous p-adic representations. Let H be a p-adic Lie group. A continuous representation of H over E is an E-Banach space Π endowed with an E-linear action of H such that the map $H \times \Pi \to \Pi$ is continuous. It is admissible if the continuous dual $\Pi^* := \operatorname{Hom}_H^{\operatorname{cont}}(\Pi, E)$ is of finite type over the Iwasawa algebra $E \otimes_{\mathcal{O}} \mathcal{O}[\![H_0]\!]$ for some (equivalently any) compact open subgroup $H_0 \subseteq H$ [Schneider and Teitelbaum 2002]. It is unitary if there exists an H-stable bounded open \mathcal{O} -lattice $\Pi^0 \subseteq \Pi$. We write $\operatorname{Ban}_H^{\operatorname{adm}, u}(E)$ for the category of admissible unitary continuous representations of H over E and H-equivariant E-linear continuous morphisms. It is an E-abelian category.

We fix a uniformiser ϖ of \mathcal{O} . Following [Emerton 2010a, §2.4], we let $\operatorname{Mod}_H^{\varpi-\operatorname{adm}}(\mathcal{O})^{\operatorname{fl}}$ be the category of ϖ -torsion-free ϖ -adically complete and separated \mathcal{O} -modules Π^0 such that $\Pi^0/\varpi\Pi^0$ is admissible as

a smooth representation of H over k and H-equivariant \mathcal{O} -linear morphisms. It is an \mathcal{O} -abelian category and the localised category $E \otimes_{\mathcal{O}} \operatorname{Mod}_H^{\varpi-\operatorname{adm}}(\mathcal{O})^{\operatorname{fl}}$ is equivalent to $\operatorname{Ban}_H^{\operatorname{adm},\operatorname{u}}(E)$.

The E-vector spaces $\operatorname{Ext}^1_H(\Pi',\Pi)$ of Yoneda extensions between admissible unitary continuous representations Π , Π' of H over E are computed in $\operatorname{Ban}^{\operatorname{adm},\operatorname{u}}_H(E)$. For all $n \geq 1$, the $\mathcal{O}/\varpi^n\mathcal{O}$ -modules $\operatorname{Ext}^1_H(\pi',\pi)$ of Yoneda extensions between admissible smooth representations π , π' of H over $\mathcal{O}/\varpi^n\mathcal{O}$ are computed in $\operatorname{Mod}^{\operatorname{adm}}_H(\mathcal{O}/\varpi^n\mathcal{O})$. The following result is a slight generalisation of [Hauseux 2016a, Proposition B.2].

Proposition 5.3.1. Let H be a p-adic Lie group, Π , Π' be admissible unitary continuous representations of H and π , π' be the reductions mod ϖ of H-stable bounded open \mathcal{O} -lattices Π^0 , Π'^0 of Π , Π' respectively. Assume that $\dim_k \operatorname{Hom}_H(\pi',\pi) < \infty$. There is an E-linear isomorphism

$$\operatorname{Ext}^1_H(\Pi',\Pi) \cong E \otimes_{\mathcal{O}} \varprojlim_{n \geq 1} \left(\operatorname{Ext}^1_H(\Pi'^0/\varpi^n \Pi'^0,\Pi^0/\varpi^n \Pi^0) \right).$$

Furthermore, $\dim_E \operatorname{Ext}^1_H(\Pi', \Pi) \leq \dim_k \operatorname{Ext}^1_H(\pi', \pi)$.

Proof. In the proof of [Hauseux 2016a, Proposition B.2], the hypothesis that π' is of finite length is only used to prove that $\operatorname{Hom}_H(\Pi'^0/\varpi^n\Pi'^0,\Pi^0/\varpi^n\Pi^0)$ is of finite type over $\mathcal{O}/\varpi^n\mathcal{O}$ for all $n \geq 1$. But this can be proved by induction using that $\dim_k \operatorname{Hom}_H(\pi',\pi) < \infty$.

Let P = LN be a parabolic subgroup. We recall that the continuous parabolic induction functor is defined for any continuous representation Σ of L over E by

$$\operatorname{Ind}_{P^{-}}^{G}\Sigma := \left\{ f: G \to \Sigma \text{ continuous } \middle| \ f(pg) = p \cdot f(g) \ \forall p \in P^{-}, \forall g \in G \right\}.$$

We obtain an E-linear exact functor $\operatorname{Ind}_{P^-}^G : \operatorname{Ban}_L^{\operatorname{adm}, \operatorname{u}}(E) \to \operatorname{Ban}_G^{\operatorname{adm}, \operatorname{u}}(E)$ [Emerton 2010a, §4.1]. Furthermore, there is a natural G-equivariant E-linear continuous isomorphism [op. cit., Lemma 4.1.3]

$$\operatorname{Ind}_{P^{-}}^{G} \Sigma \cong E \otimes_{\mathcal{O}} \varprojlim_{n>1} \left(\operatorname{Ind}_{P^{-}}^{G} (\Sigma^{0} / \varpi^{n} \Sigma^{0}) \right).$$

We extend the definition of the ordinary parts functor to any admissible unitary continuous representation Π of G over E by setting

$$\operatorname{Ord}_{P} \Pi := E \otimes_{\mathcal{O}} \varprojlim_{n \geq 1} \left(\operatorname{Ord}_{P} (\Pi^{0} / \varpi^{n} \Pi^{0}) \right)$$

for some (equivalently any) G-stable bounded open \mathcal{O} -lattice $\Pi^0 \subseteq \Pi$. We obtain an E-linear left-exact functor $\operatorname{Ord}_P : \operatorname{Ban}_G^{\operatorname{adm},\operatorname{u}}(E) \to \operatorname{Ban}_L^{\operatorname{adm},\operatorname{u}}(E)$ which is a left quasi-inverse and the right adjoint of $\operatorname{Ind}_{P^-}^G$ [Emerton 2010a, Theorem 3.4.8, Corollary 4.3.5 and Theorem 4.4.6].

Definition 5.3.2. We say that an admissible unitary continuous representation Σ of L over E is *right cuspidal* if $Ord_O \Sigma = 0$ for any proper parabolic subgroup $Q \subset L$.

Remark 5.3.3. We also extend the Jacquet functor to continuous representations of G over E by taking the Hausdorff completion of the N-coinvariants. We obtain the left adjoint of Ind_P^G by Frobenius reciprocity and the universal property of coinvariants. However, we do not know whether it preserves admissibility. For unitary representations, it does not behave well with respect to reduction mod ϖ^n $(n \ge 1)$. Nevertheless,

we say that an admissible unitary continuous representation Σ of L over E is *left cuspidal* if $\Sigma_{N_Q} = 0$ for any proper parabolic subgroup $Q \subset L$ with unipotent radical N_Q .

We now turn to extensions computations. Our main tool is the following result, which gives a weak *p*-adic analogue of the exact sequence (39).

Proposition 5.3.4. Let P = LN be a standard parabolic subgroup, Σ , Σ' be admissible unitary continuous representations of L respectively over E and σ , σ' be the reductions mod ϖ of L-stable bounded open \mathcal{O} -lattices Σ^0 , Σ'^0 of Σ , Σ' respectively. Assume that $\dim_k \operatorname{Hom}_L(\sigma', \sigma) < \infty$. There is a natural exact sequence of E-vector spaces

$$0 \to \operatorname{Ext}^1_L(\Sigma', \Sigma) \to \operatorname{Ext}^1_G\big(\operatorname{Ind}_{P^-}^G\Sigma', \operatorname{Ind}_{P^-}^G\Sigma\big) \to \operatorname{Hom}_L\big(\Sigma', E \otimes_{\mathcal{O}} \varprojlim_{n \geq 1} \big(\operatorname{H}^1\operatorname{Ord}_P\big(\operatorname{Ind}_{P^-}^G(\Sigma^0/\varpi^n\Sigma^0)\big)\big)\big).$$

Proof. For all $n \ge 1$, (39) with $A = \mathcal{O}/\varpi^n \mathcal{O}$, $\pi = \operatorname{Ind}_{P^-}^G(\Sigma^0/\varpi^n \Sigma^0)$ and using [Emerton 2010a, Proposition 4.3.4] $\sigma = \Sigma'^0/\varpi^n \Sigma'^0$ yields an exact sequence of $\mathcal{O}/\varpi^n \mathcal{O}$ -modules

$$0 \to \operatorname{Ext}_{L}^{1}(\Sigma'^{0}/\varpi^{n}\Sigma'^{0}, \Sigma^{0}/\varpi^{n}\Sigma^{0}) \to \operatorname{Ext}_{G}^{1}\left(\operatorname{Ind}_{P^{-}}^{G}(\Sigma'^{0}/\varpi^{n}\Sigma'^{0}), \operatorname{Ind}_{P^{-}}^{G}(\Sigma^{0}/\varpi^{n}\Sigma^{0})\right) \\ \to \operatorname{Hom}_{L}\left(\Sigma'^{0}/\varpi^{n}\Sigma'^{0}, \operatorname{H}^{1}\operatorname{Ord}_{P}\left(\operatorname{Ind}_{P^{-}}^{G}(\Sigma^{0}/\varpi^{n}\Sigma^{0})\right)\right). \tag{60}$$

The composite $\mathrm{H}^1\mathrm{Ord}_P \circ \mathrm{Ind}_{P^-}^G : \mathrm{Mod}_L^{\mathrm{adm}}(\mathcal{O}/\varpi^n\mathcal{O}) \to \mathrm{Mod}_L^{\mathrm{adm}}(\mathcal{O}/\varpi^n\mathcal{O})$ is left-exact for all $n \ge 1$ by [Emerton 2010a, Proposition 4.3.4; Emerton 2010b, Corollary 3.4.8]. Thus

$$\underbrace{\lim_{n>1}}_{n>1} \left(H^1 \operatorname{Ord}_P (\operatorname{Ind}_{P^-}^G (\Sigma^0 / \varpi^n \Sigma^0)) \right)$$

is a ϖ -adically admissible representation of L over \mathcal{O} by [Emerton 2010a, Corollary 3.4.5]. Furthermore, it is ϖ -torsion-free and the projective limit topology coincide with the ϖ -adic topology [op. cit., Proposition 3.4.3(1) and (3)]. Thus

$$E \otimes_{\mathcal{O}} \varprojlim_{n \geq 1} (\mathsf{H}^1 \mathsf{Ord}_P \big(\mathsf{Ind}_{P^-}^G (\Sigma^0 / \varpi^n \Sigma^0) \big) \big)$$

is an admissible unitary continuous representation of L over E. Taking the projective limit over $n \ge 1$ of (60) and inverting ϖ and using Proposition 5.3.1 yields the desired exact sequence.

Remark 5.3.5. In order to obtain an analogue of (39) for any admissible unitary continuous representations Σ , Π of L, G respectively over E, one has to prove that the ϖ -torsion of $\varprojlim_{n\geq 1} (H^1\mathrm{Ord}_P(\Pi^0/\varpi^n\Pi^0))$ is of bounded exponent (i.e., annihilated by a power of ϖ) for some (equivalently any) G-stable bounded open \mathcal{O} -lattice $\Pi^0 \subseteq \Pi$.

We now use Proposition 5.3.4 to compute extensions between parabolically induced representations.

Theorem 5.3.6. Assume $F = \mathbb{Q}_p$. Let P = LN, P' = L'N' be standard parabolic subgroups, Σ , Σ' be admissible unitary continuous representations of L, L' over E and σ , σ' be the reductions mod ϖ of L, L'-stable bounded open \mathcal{O} -lattices of Σ , Σ' . Assume that $\dim_k \operatorname{Hom}_G(\operatorname{Ind}_{P'^-}^G \sigma', \operatorname{Ind}_{P^-}^G \sigma) < \infty$ and Σ is right cuspidal.

(i) If P' = P, Σ , Σ' are topologically irreducible and $\Sigma' \ncong \Sigma^{\alpha} \otimes (\varepsilon^{-1} \circ \alpha)$ for all $\alpha \in \Delta_L^{\perp,1}$, then the functor $\operatorname{Ind}_{P^-}^G$ induces an E-linear isomorphism

$$\operatorname{Ext}_L^1(\Sigma', \Sigma) \xrightarrow{\sim} \operatorname{Ext}_G^1(\operatorname{Ind}_{P^-}^G \Sigma', \operatorname{Ind}_{P^-}^G \Sigma).$$

(ii) If $P' \subseteq P$, then the functor $\operatorname{Ind}_{P^-}^G$ induces an E-linear isomorphism

$$\operatorname{Ext}^1_L\big(\operatorname{Ind}^L_{L\cap P'^-}\Sigma',\,\Sigma\big)\xrightarrow{\sim}\operatorname{Ext}^1_G\big(\operatorname{Ind}^G_{P'^-}\Sigma',\,\operatorname{Ind}^G_{P^-}\Sigma\big).$$

Remark 5.3.7. Assume P' = P and Σ , Σ' topologically irreducible. We do not know the dimension of the cokernel of the E-linear injection $\operatorname{Ext}_L^1(\Sigma', \Sigma) \hookrightarrow \operatorname{Ext}_G^1(\operatorname{Ind}_{P^-}^G \Sigma', \operatorname{Ind}_{P^-}^G \Sigma)$ induced by $\operatorname{Ind}_{P^-}^G$, but we prove that it is at most $\operatorname{card}\{\alpha \in \Delta_L^{\perp,1} \mid \Sigma' \cong \Sigma^\alpha \otimes (\varepsilon^{-1} \circ \alpha)\}$ (see the proof). If Σ , Σ' are absolutely topologically irreducible and supercuspidal, then letting $\zeta: Z_L \to \mathcal{O}^\times \subset E^\times$ be the central character of Σ [Dospinescu and Schraen 2013, Theorem 1.1(2)], we expect this dimension to be equal to

card
$$\{\alpha \in \Delta_L^{\perp, 1} \mid \Sigma' \cong \Sigma^{\alpha} \otimes (\varepsilon^{-1} \circ \alpha) \text{ and } \zeta \circ \alpha^{\vee} \neq \varepsilon^{-1} \}$$
.

Proof. Let $\Sigma^0 \subseteq \Sigma$ be an L-stable bounded open \mathcal{O} -lattice. For all $n \ge 1$, we deduce from Propositions 3.3.1 and 3.3.5(i) that there is a natural L-equivariant $\mathcal{O}/\varpi^n\mathcal{O}$ -linear isomorphism

$$H^{1}\operatorname{Ord}_{P}\left(\operatorname{Ind}_{P^{-}}^{G}(\Sigma^{0}/\varpi^{n}\Sigma^{0})\right) \cong \bigoplus_{\alpha \in \Delta^{1}\backslash \Delta_{L}} H^{1}\operatorname{Ord}_{P}\left(\operatorname{c-ind}_{P^{-}}^{P^{-}s_{\alpha}P}(\Sigma^{0}/\varpi^{n}\Sigma^{0})\right). \tag{61}$$

Furthermore, if $\alpha \in \Delta_L^{\perp,1}$, then there is a natural L-equivariant $\mathcal{O}/\varpi^n\mathcal{O}$ -linear isomorphism

$$\mathrm{H}^{1}\mathrm{Ord}_{P}\left(\mathrm{c\text{-}ind}_{P^{-}}^{P^{-}s_{\alpha}P}(\Sigma^{0}/\varpi^{n}\Sigma^{0})\right) \cong (\Sigma^{0}/\varpi^{n}\Sigma^{0})^{\alpha} \otimes (\omega^{-1} \circ \alpha)$$

hence a natural L-equivariant E-linear continuous isomorphism

$$E \otimes_{\mathcal{O}} \varprojlim_{n \geq 1} (\mathsf{H}^1 \mathsf{Ord}_P (\mathsf{c\text{-}ind}_{P^-}^{P^- s_\alpha P} (\Sigma^0 / \varpi^n \Sigma^0))) \cong \Sigma^\alpha \otimes (\varepsilon^{-1} \circ \alpha),$$

whereas if $\alpha \notin \Delta_L^{\perp}$, there is natural filtration of $H^1 \operatorname{Ord}_P(\operatorname{c-ind}_{P^-}^{P^-s_{\alpha}P}(\Sigma^0/\varpi^n\Sigma^0))$ by B_L -subrepresentations such that each term of the associated graded representation is isomorphic as an $\mathcal{O}/\varpi^n\mathcal{O}$ -modules to

$$C_{\rm c}^{\rm sm}(U_L', {\rm Ord}_{L\cap s_{\rm or} Ps_{\rm or}^{-1}}(\Sigma^0/\varpi^n\Sigma^0))$$

for some closed subgroup $U'_L \subseteq U_L$, and since $\operatorname{Ord}_{L \cap s_{\alpha}Ps_{\alpha}^{-1}} \Sigma = 0$ we deduce that

$$E \otimes_{\mathcal{O}} \varprojlim_{n \geq 1} (\mathsf{H}^1 \mathsf{Ord}_P \big(\mathsf{c\text{-}ind}_{P^-}^{P^- s_\alpha P} (\Sigma^0 / \varpi^n \Sigma^0) \big) \big) = 0.$$

Thus, taking the projective limit of (61) over $n \ge 1$ and inverting ϖ yields a natural L-equivariant E-linear continuous isomorphism

$$E \otimes_{\mathcal{O}} \varprojlim_{n \geq 1} \left(\operatorname{H}^{1} \operatorname{Ord}_{P} \left(\operatorname{Ind}_{P^{-}}^{G} (\Sigma^{0} / \varpi^{n} \Sigma^{0}) \right) \right) \cong \bigoplus_{\alpha \in \Delta_{L}^{\perp, 1}} \Sigma^{\alpha} \otimes (\varepsilon^{-1} \circ \alpha).$$
 (62)

Now Proposition 5.3.4 with $\operatorname{Ind}_{L\cap P'^-}^L\Sigma'$ instead of Σ' yields, using (62), an exact sequence of *E*-vector spaces

$$0 \to \operatorname{Ext}_{L}^{1}\left(\operatorname{Ind}_{L \cap P'^{-}}^{L} \Sigma', \Sigma\right) \to \operatorname{Ext}_{G}^{1}\left(\operatorname{Ind}_{P'^{-}}^{G} \Sigma', \operatorname{Ind}_{P^{-}}^{G} \Sigma\right)$$

$$\to \bigoplus_{\alpha \in \Delta_{L}^{\perp, 1}} \operatorname{Hom}_{L}\left(\operatorname{Ind}_{L \cap P'^{-}}^{L} \Sigma', \Sigma^{\alpha} \otimes (\varepsilon^{-1} \circ \alpha)\right). \quad (63)$$

If P' = P and Σ , Σ' are topologically irreducible, then $\Sigma^{\alpha} \otimes (\varepsilon^{-1} \circ \alpha)$ is also topologically irreducible for all $\alpha \in \Delta_L^{\perp,1}$, and thus the last term of (63) has dimension equal to $\operatorname{card}\{\alpha \in \Delta_L^{\perp,1} \mid \Sigma' \cong \Sigma^{\alpha} \otimes (\varepsilon^{-1} \circ \alpha)\}$, hence (i). If $P' \subsetneq P$, then $L \cap P'$ is a proper parabolic subgroup of L so that

$$\operatorname{Ord}_{L\cap P'}(\Sigma^{\alpha}\otimes(\varepsilon^{-1}\circ\alpha))=(\operatorname{Ord}_{L\cap P'}\Sigma)^{\alpha}\otimes(\varepsilon^{-1}\circ\alpha)=0$$

for all $\alpha \in \Delta_L^{\perp,1}$; thus the last term of (63) is zero, hence (ii).

Remark 5.3.8. Theorem 5.2.2(iii) cannot be directly lifted to characteristic 0 because we do not have a weak p-adic analogue of the exact sequence (43) (since it uses the Jacquet functor, see Remark 5.3.3). However, assuming Conjecture 3.3.4 true for $A = \mathcal{O}/\varpi^r \mathcal{O}$ ($r \ge 1$), n = 1, $I \subseteq J$ and $Iw^J = s_\alpha$ ($\alpha \in \Delta^1 \setminus J$), one can recover this case: with notation and assumptions as in Theorem 5.3.6, if $P \subseteq P'$ and Σ' is left cuspidal, then the functor $\operatorname{Ind}_{P'^-}^G$ induces an E-linear isomorphism

$$\operatorname{Ext}^1_{L'}(\Sigma',\operatorname{Ind}^L_{L'\cap P^-}\Sigma) \xrightarrow{\sim} \operatorname{Ext}^1_G(\operatorname{Ind}^G_{P'^-}\Sigma',\operatorname{Ind}^G_{P^-}\Sigma).$$

Theorem 5.3.9. Assume $F \neq \mathbb{Q}_p$. Let P = LN be a standard parabolic subgroup, Σ , Σ' be admissible unitary continuous representations of L over E and σ , σ' be the reductions mod ϖ of L-stable bounded open \mathcal{O} -lattices of Σ , Σ' respectively. Assume that $\dim_k \operatorname{Hom}_L(\sigma', \sigma) < \infty$. Then, the functor $\operatorname{Ind}_{P^-}^G$ induces an E-linear isomorphism

$$\operatorname{Ext}_L^1(\Sigma', \Sigma) \xrightarrow{\sim} \operatorname{Ext}_G^1(\operatorname{Ind}_{P^-}^G \Sigma', \operatorname{Ind}_{P^-}^G \Sigma).$$

Proof. Let $\Sigma^0 \subseteq \Sigma$ be an L-stable bounded open \mathcal{O} -lattice. Then $H^1\mathrm{Ord}_P(\mathrm{Ind}_{P^-}^G(\Sigma^0/\varpi^n\Sigma^0)) = 0$ for all $n \ge 1$, by Corollary 3.3.8(i). Thus, the result follows from Proposition 5.3.4.

We end with a remark on the case where there is no inclusion between the two parabolic subgroups.

Remark 5.3.10. Let P = LN, P' = L'N' be standard parabolic subgroups, Σ , Σ' be admissible unitary continuous representations of L, L' respectively over E and σ , σ' be the reductions mod ϖ of L, L'-stable bounded open \mathcal{O} -lattices of Σ , Σ' respectively. Assume Conjecture 3.3.4 is true for $A = \mathcal{O}/\varpi^r\mathcal{O}$ $(r \geq 1)$, n = 1 and ${}^Iw^J = 1$. Assume further $\dim_k \operatorname{Hom}_G(\operatorname{Ind}_{P'^-}^G\sigma', \operatorname{Ind}_{P^-}^G\sigma) < \infty$ and the ϖ -torsion of $\varprojlim_{n \geq 1}(\operatorname{H}^1\operatorname{Ord}_{L\cap P}(\Sigma^0/\varpi^n\Sigma^0))$ is of bounded exponent (see Remark 5.3.5). Then, one can prove the following p-adic analogue of Proposition 5.2.1: if $P' \not\subseteq P$, $P \not\subseteq P'$, and Σ , Σ' are right, left cuspidal respectively, then

$$\operatorname{Ext}_G^1\left(\operatorname{Ind}_{P'^-}^G\Sigma',\operatorname{Ind}_{P^-}^G\Sigma\right)=0.$$

Acknowledgements

I am grateful to C. Breuil, whose ideas motivated the present results, and for several comments on a preliminary version of this paper. I thank N. Abe for his help with the proof of Proposition 5.1.7, as well as G. Henniart, V. Paškunas and M.-F. Vignéras for answering my questions. I also thank M.-F. Vignéras for her encouragement. Finally, I want to thank the anonymous referee for a very careful reading of the manuscript.

References

[Abdellatif 2014] R. Abdellatif, "Classification des représentations modulo p de SL(2, F)", Bull. Soc. Math. France 142:3 (2014), 537–589. MR Zbl

[Abe 2013] N. Abe, "On a classification of irreducible admissible modulo *p* representations of a *p*-adic split reductive group", *Compos. Math.* **149**:12 (2013), 2139–2168. MR Zbl

[Abe et al. 2017a] N. Abe, G. Henniart, F. Herzig, and M.-F. Vignéras, "A classification of irreducible admissible mod *p* representations of *p*-adic reductive groups", *J. Amer. Math. Soc.* **30**:2 (2017), 495–559. MR Zbl

[Abe et al. 2017b] N. Abe, G. Henniart, and M.-F. Vignéras, "Modulo *p* representations of reductive *p*-adic groups: functorial properties", preprint, 2017. To appear in *Trans. Amer. Math. Soc.* arXiv

[Barthel and Livné 1994] L. Barthel and R. Livné, "Irreducible modular representations of GL₂ of a local field", *Duke Math. J.* 75:2 (1994), 261–292. MR Zbl

[Bernšteĭn and Zelevinskiĭ 1976] I. N. Bernšteĭn and A. V. Zelevinskiĭ, "Representations of the group GL(n, F), where F is a local non-Archimedean field", *Uspehi Mat. Nauk* **31**:3(189) (1976), 5–70. In Russian; translated in *Russian Math. Surveys* **31**:3 (1976), 1–68. MR

[Björner and Brenti 2005] A. Björner and F. Brenti, *Combinatorics of Coxeter groups*, Graduate Texts in Mathematics 231, Springer, New York, 2005. MR Zbl

[Borel and Tits 1965] A. Borel and J. Tits, "Groupes réductifs", *Inst. Hautes Études Sci. Publ. Math.* 27 (1965), 55–150. MR Zbl

[Borel and Tits 1972] A. Borel and J. Tits, "Compléments à l'article: "Groupes réductifs", *Inst. Hautes Études Sci. Publ. Math.* **41** (1972), 253–276. MR Zbl

[Breuil 2003] C. Breuil, "Sur quelques représentations modulaires et p-adiques de $GL_2(\mathbb{Q}_p)$, I", Compositio Math. 138:2 (2003), 165–188. MR Zbl

[Breuil and Herzig 2015] C. Breuil and F. Herzig, "Ordinary representations of $G(\mathbb{Q}_p)$ and fundamental algebraic representations", *Duke Math. J.* **164**:7 (2015), 1271–1352. MR Zbl

[Cheng 2013] C. Cheng, "Mod p representations of $SL_2(\mathbb{Q}_p)$ ", J. Number Theory 133:4 (2013), 1312–1330. MR

[Colliot-Thélène 2008] J.-L. Colliot-Thélène, "Résolutions flasques des groupes linéaires connexes", *J. Reine Angew. Math.* **618** (2008), 77–133. MR Zbl

[Colmez 2010] P. Colmez, "Représentations de $GL_2(\mathbb{Q}_p)$ et (ϕ, Γ) -modules", pp. 281–509 Astérisque **330**, 2010. MR Zbl

[Demazure and Gabriel 1970] M. Demazure and P. Gabriel, *Groupes algébriques, Tome I: Géométrie algébrique, généralités, groupes commutatifs*, Masson, Paris, 1970. MR Zbl

[Digne and Michel 1991] F. Digne and J. Michel, *Representations of finite groups of Lie type*, London Mathematical Society Student Texts **21**, Cambridge University Press, 1991. MR Zbl

[Dospinescu and Schraen 2013] G. Dospinescu and B. Schraen, "Endomorphism algebras of admissible *p*-adic representations of *p*-adic Lie groups", *Represent. Theory* **17** (2013), 237–246. MR Zbl

[Emerton 2006] M. Emerton, "Jacquet modules of locally analytic representations of *p*-adic reductive groups, I: Construction and first properties", *Ann. Sci. École Norm. Sup.* (4) **39**:5 (2006), 775–839. MR Zbl

[Emerton 2010a] M. Emerton, "Ordinary parts of admissible representations of *p*-adic reductive groups, I: Definition and first properties", pp. 355–402 Astérisque **331**, 2010. MR Zbl

[Emerton 2010b] M. Emerton, "Ordinary parts of admissible representations of *p*-adic reductive groups, II: Derived functors", pp. 403–459 Astérisque **331**, 2010. MR Zbl

[Hauseux 2016a] J. Hauseux, "Extensions entre séries principales p-adiques et modulo p de G(F)", J. Inst. Math. Jussieu 15:2 (2016), 225–270. MR Zbl

[Hauseux 2016b] J. Hauseux, "Sur une conjecture de Breuil-Herzig", J. reine angew. Math. (online publication October 2016).

[Hauseux 2017] J. Hauseux, "Compléments sur les extensions entre séries principales p-adiques et modulo p de G(F)", Bull. Soc. Math. France 145:1 (2017), 161–192. MR Zbl

[Hauseux et al. 2016] J. Hauseux, T. Schmidt, and C. Sorensen, "Deformation rings and parabolic induction", preprint, 2016. To appear in *J. Théor. Nombres Bordeaux*. arXiv

[Henniart and Vignéras 2015] G. Henniart and M.-F. Vignéras, "A Satake isomorphism for representations modulo *p* of reductive groups over local fields", *J. Reine Angew. Math.* **701** (2015), 33–75. MR Zbl

[Herzig 2011] F. Herzig, "The classification of irreducible admissible mod p representations of a p-adic GL_n ", Invent. Math. **186**:2 (2011), 373–434. MR Zbl

[Hu 2017] Y. Hu, "An application of a theorem of Emerton to mod p representations of GL_2 ", J. Lond. Math. Soc. (2) 96:3 (2017), 545–564. MR Zbl

[Kashiwara and Schapira 2006] M. Kashiwara and P. Schapira, *Categories and sheaves*, Grundlehren Math. Wissenschaften **332**, Springer, Berlin, 2006. MR Zbl

[Kozioł 2016] K. Kozioł, "A classification of the irreducible mod p representations of U(1, 1)($\mathbb{Q}_{p^2}/\mathbb{Q}_p$)", Ann. Inst. Fourier (Grenoble) **66**:4 (2016), 1545–1582. MR Zbl

[Oort 1964] F. Oort, "Yoneda extensions in abelian categories", Math. Ann. 153 (1964), 227–235. MR Zbl

[Paškūnas 2010] V. Paškūnas, "Extensions for supersingular representations of $GL_2(\mathbb{Q}_p)$ ", pp. 317–353 Astérisque **331**, 2010. MR Zbl

[Paškūnas 2013] V. Paškūnas, "The image of Colmez's Montreal functor", *Publ. Math. Inst. Hautes Études Sci.* **118** (2013), 1–191. MR Zbl

[Schneider and Teitelbaum 2002] P. Schneider and J. Teitelbaum, "Banach space representations and Iwasawa theory", *Israel J. Math.* **127** (2002), 359–380. MR Zbl

[Vignéras 2016] M.-F. Vignéras, "The right adjoint of the parabolic induction", pp. 405–425 in *Arbeitstagung Bonn 2013* (Bonn, 2013), edited by W. Ballmann et al., Progr. Math. **319**, Springer, 2016. MR Zbl

Communicated by Marie-France Vignéras

Received 2016-09-07 Revised 2017-04-23 Accepted 2017-05-25

julien.hauseux@math.univ-lille1.fr Département de Mathématiques, Université de Lille, Villeneuve d'Ascq, France



Algebra & Number Theory

msp.org/ant

EDITORS

MANAGING EDITOR

Bjorn Poonen

Massachusetts Institute of Technology

Cambridge, USA

EDITORIAL BOARD CHAIR

David Eisenbud

University of California

Berkeley, USA

BOARD OF EDITORS

Richard E. Borcherds	University of California, Berkeley, USA	Martin Olsson	University of California, Berkeley, USA
Antoine Chambert-Loir	Université Paris-Diderot, France	Raman Parimala	Emory University, USA
J-L. Colliot-Thélène	CNRS, Université Paris-Sud, France	Jonathan Pila	University of Oxford, UK
Brian D. Conrad	Stanford University, USA	Anand Pillay	University of Notre Dame, USA
Samit Dasgupta	University of California, Santa Cruz, USA	Michael Rapoport	Universität Bonn, Germany
Hélène Esnault	Freie Universität Berlin, Germany	Victor Reiner	University of Minnesota, USA
Gavril Farkas	Humboldt Universität zu Berlin, Germany	Peter Sarnak	Princeton University, USA
Hubert Flenner	Ruhr-Universität, Germany	Joseph H. Silverman	Brown University, USA
Sergey Fomin	University of Michigan, USA	Michael Singer	North Carolina State University, USA
Edward Frenkel	University of California, Berkeley, USA	Christopher Skinner	Princeton University, USA
Andrew Granville	Université de Montréal, Canada	Vasudevan Srinivas	Tata Inst. of Fund. Research, India
Joseph Gubeladze	San Francisco State University, USA	J. Toby Stafford	University of Michigan, USA
Roger Heath-Brown	Oxford University, UK	Pham Huu Tiep	University of Arizona, USA
Craig Huneke	University of Virginia, USA	Ravi Vakil	Stanford University, USA
Kiran S. Kedlaya	Univ. of California, San Diego, USA	Michel van den Bergh	Hasselt University, Belgium
János Kollár	Princeton University, USA	Marie-France Vignéras	Université Paris VII, France
Philippe Michel	École Polytechnique Fédérale de Lausanne	Kei-Ichi Watanabe	Nihon University, Japan
Susan Montgomery	University of Southern California, USA	Shou-Wu Zhang	Princeton University, USA
Shigefumi Mori	RIMS, Kyoto University, Japan		

PRODUCTION

production@msp.org Silvio Levy, Scientific Editor

See inside back cover or msp.org/ant for submission instructions.

The subscription price for 2018 is US \$340/year for the electronic version, and \$535/year (+\$55, if shipping outside the US) for print and electronic. Subscriptions, requests for back issues and changes of subscriber address should be sent to MSP.

Algebra & Number Theory (ISSN 1944-7833 electronic, 1937-0652 printed) at Mathematical Sciences Publishers, 798 Evans Hall #3840, c/o University of California, Berkeley, CA 94720-3840 is published continuously online. Periodical rate postage paid at Berkeley, CA 94704, and additional mailing offices.

ANT peer review and production are managed by EditFLow® from MSP.

PUBLISHED BY

mathematical sciences publishers nonprofit scientific publishing

http://msp.org/
© 2018 Mathematical Sciences Publishers

Algebra & Number Theory

Volume 12 No. 4 2018

A generic slice of the moduli space of line arrangements KENNETH ASCHER and PATRICIO GALLARDO	751
Parabolic induction and extensions JULIEN HAUSEUX	779
Akizuki–Witt maps and Kaletha's global rigid inner forms OLIVIER TAÏBI	833
(φ, Γ)-modules de de Rham et fonctions L p -adiques Joaquín Rodrigues Jacinto	885
Invariant theory of $\bigwedge^3(9)$ and genus-2 curves ERIC M. RAINS and STEVEN V SAM	935
Sums of two cubes as twisted perfect powers, revisited MICHAEL A. BENNETT, CARMEN BRUNI and NUNO FREITAS	959